Ultrathin Organic Thin-Film Transistors : Investigating Hybrid Gate Dielectrics and Stable Semiconductor Monolayers

Von der Fakultät Chemie der Universität Stuttgart zur Erlangung der Würde eines Doktors der Naturwissenschaften (Dr. rer. nat.) genehmigte Abhandlung

vorgelegt von

Rachana Acharya

aus Pune, Indien

Hauptberichter : Prof. Dr. Dr. h.c. Guido Schmitz Mitberichter : Prof. Dr. rer. nat. habil. Sabine Ludwigs Prüfungsvorsitzender: Prof. Dr. Peer Fischer Tag der mündlichen Prüfung: 22.12.2021

Institut für Materialwissenschaft (IMW) der Universität Stuttgart

2021

Dedicated to Late Dr. G. N. Acharya; Scientist, Role model, Grandfather, Friend.

Erklärung über die Eigenständigkeit der Dissertation

Ich versichere, dass ich die vorliegende Arbeit mit dem Titel

"Ultrathin organic thin-film transistors : Investigating Hybrid Gate Dielectrics and Stable Semiconductor Monolayers"

Selbstständig verfasst und keine anderen als die angegebenen Quellen und Hilfsmittel benutzt habe. Aus fremden Quellen entnommene Passagen und Gedanken sind als solche kenntlich gemacht.

Declaration of authorship

I hereby certify that the dissertation entitled

"Ultrathin organic thin-film transistors : Investigating Hybrid Gate Dielectrics and Stable Semiconductor Monolayers"

is entirely my own work except where otherwise indicated. Passages and ideas from other sources have been clearly indicated.

Rachana Acharya, Stuttgart, 2022

Abstract

Organic electronics is an emerging field of research which includes the investigation of novel materials such as organic semiconductors and the development of devices such as organic thin-film transistors (TFTs). The unique properties of organic semiconductors, such as the ability to process them at relatively low temperatures, enable the development of potential applications of organic TFTs in flexible and wearable electronics such as rollable and foldable displays, conformable sensors and electronic skin. In order to facilitate the portable and lightweight nature of flexible electronics by powering them with small batteries or solar cells, a low operating voltage and an overall low power consumption are some of the main requirements of organic TFTs. High-capacitance gate dielectrics, such as hybrid gate dielectrics with an ultrathin metal oxide and an organic self-assembled monolayer (SAM), are an essential choice towards fulfilling these operation requirements. Organic TFTs are fabricated by depositing different materials as thin films by a variety of processes, and the individual film properties of the different components influence the overall electrical characteristics of organic TFTs. The main contribution of this thesis is to establish a correlation between the material properties of the individual components and the electrical properties of the organic TFTs, and moreover, suitably modify the fabrication process to achieve better electrical characteristics in organic TFTs.

In this thesis, hybrid gate dielectrics consisting of an ultrathin aluminum oxide (AlO_x) film and a phosphonic acid SAM are investigated. The AlO_x films are fabricated by exposing the surface of the underlying aluminum gate electrode to an oxygen plasma, and the SAMs are processed from solution. Phosphonic acid molecules with an alkyl or a fluoroalkyl chain with different chain lengths have been chosen to form the SAMs. Two small-molecule organic semiconductors are selected as the active material in the organic TFTs: dinaphtho[2,3-b:2',3'-f]thieno[3,2-b]thiophene (DNTT) and 2,7-diphenyl[1]benzothieno[3,2-b][1]benzothiophene (DPh-BTBT).

The significance of both components of the hybrid gate dielectric in simultaneously achieving a leakage-current density of 10^{-7} A/cm² as well as an operating voltage of 3 V has been established. The thickness of the AlO_x films was measured by Transmission Electron Microscopy and the films were characterized by Electron Energy Loss Spectroscopy and Atom Probe Tomography to determine the thin-film composition. Depending on the parameters of the oxygen-plasma process, AlO_x films with a thickness ranging between 4 nm and 7 nm were fabricated, and consequently, organic TFTs with

a gate-dielectric capacitance between 1 μ F/cm² and 1.6 μ F/cm². In particular, chargecarrier mobilities ranging from 1.8 to 2.3 cm²/Vs were obtained for a number of favorable combinations of the plasma power and plasma duration that produce AlO_x films with a small surface roughness and thus promote the formation of high-quality SAMs and well-ordered DNTT films on these gate dielectrics. The influence of the thickness of the SAM by employing different chain-length phosphonic acid molecules on different TFT characteristics such as the threshold voltage, gate-leakage current, charge-carrier mobility and the subthreshold swing has been examined in DNTT and DPh-BTBT TFTs. By employing the medium-chain-length phosphonic acid molecules, an optimum chargecarrier mobility of 2 cm²/Vs for the DNTT TFTs, and a turn-on voltage of 0 V for the DPh-BTBT TFTs was achieved. The growth and morphology of the organic semiconductor DNTT on different gate-dielectric surfaces was observed by Scanning Electron Microscopy and Atomic Force Microscopy and was correlated with surface properties of

The stability of ultrathin films with a thickness of one-two molecular monolayers of the organic semiconductor DNTT was investigated, and spontaneous morphological changes occurring in the films were observed and correlated with the stability of organic TFTs based on these films. The structural reconfiguration of the ultrathin DNTT films and the degradation of the charge-carrier mobility of organic TFTs based on these films were prevented to a certain extent by cryogenic cooling and in-situ encapsulation.

the SAMs and the electrical characteristics of TFTs based on those films.

The hybrid gate dielectric with its two components, the organic semiconductor and the gate dielectric-semiconductor interface are the focal points in this thesis. The material and surface properties of the individual components of the gate dielectric have been correlated with the film properties of the organic semiconductor and further with the electrical characteristics of organic TFTs.

Zusammenfassung

Organische Elektronik ist ein aufkommendes Forschungsgebiet, das die Untersuchung neuartiger Materialien wie organischer Halbleiter und die Entwicklung von Bauelementen wie organischen Dünnschichttransistoren (TFTs) umfasst. Die einzigartigen Eigenschaften organischer Halbleiter, wie beispielsweise die Fähigkeit, sie bei relativ niedrigen Temperaturen zu verarbeiten, ermöglichen die Entwicklung potenzieller Anwendungen organischer TFTs in flexibler und tragbarer Elektronik wie rollbare und faltbare Displays, anpassungsfähige Sensoren und elektronische Haut. Um die tragbare und leichte Natur flexibler Elektronik zu ermöglichen, indem sie mit kleinen Batterien oder Solarzellen betrieben wird, sind niedrige Betriebsspannung und allgemein geringer Stromverbrauch einige der Hauptanforderungen an organische TFTs. Gate-Dielektrika mit hoher Kapazität, wie z. B. hybride Gate-Dielektrika, bestehend aus einem ultradünnen Metalloxid und einer organischen selbstorganisierten Monolage (SAM), sind eine wesentliche Voraussetzung für die Erfüllung dieser Betriebsanforderungen. Organische TFTs werden durch die Abscheidung verschiedener Materialien in Form von dünnen Schichten durch eine Vielzahl von Verfahren hergestellt, und die individuellen Materialund Schichteigenschaften der verschiedenen Komponenten beeinflussen die elektrischen Gesamteigenschaften der organischen TFTs.

Der Hauptbeitrag dieser Arbeit besteht darin, eine Korrelation zwischen den Materialeigenschaften der einzelnen Komponenten und den elektrischen Eigenschaften der organischen TFTs herzustellen und darüber hinaus den Herstellungsprozess geeignet zu modifizieren, um bessere elektrische Eigenschaften in organischen TFTs zu erreichen. In der vorliegenden Arbeit werden hybride Gate-Dielektrika untersucht, die aus einer ultradünnen Schicht Aluminiumoxid (AlO_x) und einer Phosphonsäure-SAM bestehen. Die AlO_x -Schicht wird hergestellt, indem die Oberfläche der darunterliegenden Aluminium-Gate-Elektrode einem Sauerstoffplasma ausgesetzt wird, und die SAMs werden aus einer Lösung verarbeitet. Für die SAM wurden Phosphonsäuremoleküle mit einer Alkylkette oder einer Fluoralkylkette mit unterschiedlichen Kettenlängen gewählt. Als aktives Material für die organischen TFTs wurden zwei organische Halbleiter ausgewählt: Dinaphtho[2,3-b:2',3'-f]thieno[3,2-b]thiophene (DNTT) und 2,7-Diphenyl[1]benzothieno[3,2-b][1]benzothiophene (DPh-BTBT).

Die Rolle der beiden Bestandteile des hybriden Gate-Dielektrikums für das gleichzeitige Erreichen einer Leckstromdichte von 10^{-7} A/cm² sowie einer Betriebsspannung

von 3 V wurde festgestellt. Die AlO_x-Schichten wurden mittels Transmissionselektronenmikroskopie, Elektronenenergieverlustspektroskopie und Atomsondentomographie charakterisiert, um die Dicke und Zusammensetzung der Dünnschicht zu bes-Je nach den Parametern des Sauerstoff-Plasma-Prozesses wurden AlO_xtimmen. Schichten mit einer Dicke zwischen 4 nm und 7 nm hergestellt und folglich organische TFTs mit einer Gate-Dielektrizitätskapazität zwischen 1 μ F/cm² und 1.6 μ F/cm². Insbesondere wurden Ladungsträgermobilitäten im Bereich von 1,8 bis 2,3 cm²/Vs für eine Reihe günstiger Kombinationen von Plasmaleistung und Plasmadauer erreicht, die AlO_x-Filme mit geringer Oberflächenrauhigkeit erzeugen und so die Bildung hochwertiger SAMs fördern und wohlgeordnete DNTT-Filme auf diesen Gatedielektrika. Insbesondere wurden Ladungsträgermobilitäten im Bereich von 1,8 bis 2,3 cm²/Vs für eine Reihe günstiger Kombinationen von Plasmaleistung und Plasmadauer erreicht, die AlO_x-Filme mit geringer Oberflächenrauhigkeit erzeugen und so die Bildung hochwertiger SAMs und wohlgeordneter DNTT-Filme auf diesen Gatedielektrika fördern. Die Auswirkung der Dicke der SAM durch die Verwendung von Phosphonsäuremolekülen unterschiedlicher Kettenlänge auf verschiedene TFT-Eigenschaften wie Schwellenspannung, Gate-Leckstrom, Ladungsträgermobilität und Schwingung unterhalb der Schwelle wurde in DNTT- und DPh-BTBT-TFTs untersucht. Durch die Verwendung der mittelkettigen Phosphonsäuremoleküle wurde eine optimale Ladungsträgermobilität von 2 cm²/Vs für die DNTT-TFTs und eine Einschaltspannung von 0 V für die DPh-BTBT-TFTs erreicht. Das Wachstum und die Morphologie des organischen Halbleiters DNTT auf verschiedenen Gate-Dielektrikum-Oberflächen wurden mittels Rasterelektronenmikroskopie und Rasterkraftmikroskopie analysiert und mit den Oberflächeneigenschaften der SAMs und den elektrischen Eigenschaften der auf diesen Filmen basierenden TFTs korreliert.

Die Stabilität von ultradünnen DNTT-Filmen, mit einer Dicke von 1-2 molekularen Monolagen, wurde untersucht und spontane Veränderungen der Morphologie in den DNTT-Filmen mit der Stabilität von organischen TFTs korreliert. Die Stabilität von ultradünnen Filmen mit einer Dicke von 1-2 molekularen Monolagen des organischen Halbleiters DNTT wurde untersucht und spontan auftretende morphologische Veränderungen in den Filmen wurden beobachtet und mit der Stabilität von organischen TFTs basierend auf diesen Filmen korreliert. Weiterhin wurde untersucht, inwieweit die strukturelle Neuanordnung der ultradünnen DNTT-Filme und die Abnahme der Ladungsträgermobilität der organischen TFTs, die auf diesen ultradünnen DNTT-Filmen basieren, durch kryogene Abkühlung und In-situ-Verkapselung verhindert werden können.

Das hybride Gate-Dielektrikum mit seinen beiden Komponenten, dem organischen Halbleiter und der Gate-Dielektrikum-Halbleiter-Grenzschicht, steht im Mittelpunkt dieser Arbeit. Insbesondere wurden die Material- und Oberflächeneigenschaften der einzelnen Komponenten des Gatedielektrikums mit den Filmeigenschaften des organischen Halbleiters und weiter mit den elektrischen Eigenschaften organischer TFTs korreliert.

Contents

Ał	ostra	ct (En	glish/German)	vii	
Li	List of Figures xvii				
Li	st of	Tables	5	xxiii	
Ał	obrev	viation	IS	xxv	
Sy	mbo	ls	х	xvii	
1	Intr	oducti	ion	1	
2	The	ory		7	
	2.1	Organ	ic semiconductors	. 7	
	2.2	Charg	e transport in organic semiconductors	. 10	
	2.3	Organ	ic thin-film transistors	. 16	
		2.3.1	Current-voltage characteristics	. 19	
		2.3.2	Operation Regimes	. 21	
		2.3.3	TFT Operation Parameters	. 23	
	2.4	Dielec	tric properties	. 25	
		2.4.1	Capacitance Measurement	. 28	
	2.5	Growt	h and morphology of organic semiconductor films	. 31	
		2.5.1	Thermodynamic considerations of vapor nucleation and growth .	. 32	
		2.5.2	Kinetics of nucleation and growth	. 34	
		2.5.3	Growth modes	. 36	
		2.5.4	Factors influencing growth and morphology of organic films	. 37	
		2.5.5	Thickness of organic semiconductor films	. 38	
	2.6	Thin-f	ilm characterization techniques	. 39	
		2.6.1	Contact angle goniometry	. 39	
		2.6.2	Atomic force microscopy	. 39	
		2.6.3	Scanning Electron Microscopy	. 41	
		2.6.4	Transmission Electron Microscopy	. 42	
		2.6.5	X-ray Diffraction	. 42	
3	Hyb	orid ga	te dielectrics in organic thin-film transistors	43	
	3.1	Self-as	sembled monolayers	. 47	
	3.2	Hybrid	d gate dielectrics for low-voltage organic TFTs	. 51	

	3.3	Importance of individual components of a hybrid gate dielectric 53
		3.3.1 The oxide component of the hybrid gate dielectric
		3.3.2 The SAM component of the hybrid gate dielectric
	3.4	Experimental
	3.5	Summary and outlook
4	Opt	imizing plasma oxidation for aluminum oxide films 59
	4.1	Thickness of plasma-grown AlO_x films $\ldots \ldots \ldots$
	4.2	Dielectric properties of plasma-grown AlO_x films
	4.3	Electrical properties of organic TFTs with plasma-grown AlO_x films as
		gate dielectrics
	4.4	Surface properties
	4.5	Correlation of surface properties of the dielectric with electrical properties
	1.0	of the IF1s \dots
	4.0	Experimental
	4.7	Summary and Outlook
5	Inve	estigating the thickness and composition of plasma-grown aluminum
	oxic	le films in hybrid dielectrics 81
	5.1	Thickness of plasma-grown AlO_x films $\ldots \ldots \ldots \ldots \ldots \ldots 83$
	5.2	Composition of plasma-grown AlO_x films $\ldots \ldots \ldots$
	5.3	Qualitative measurement by TEM-EELS
		5.3.1 Experimental
		5.3.2 Results
	F 4	5.3.3 Outlook
	5.4	Quantitative measurement by Atom Probe
		$[10 \text{ mography} \dots \dots$
		$5.4.1 \text{Experimental} \dots \dots \dots \dots \dots \dots \dots \dots \dots $
		$5.4.2 \text{Results} \dots \dots \dots \dots \dots \dots \dots \dots \dots $
	55	Summary and Outlook
	5.5	
6	Infl	uence of self-assembled monolayer thickness in hybrid gate dielectrics:
		yl and fluoroalkyl phosphonic acids
	6.1	Background
	6.2	Dielectric properties of hybrid gate dielectrics with self-assembled mono- layers
	6.3	Electrical properties of organic TFTs with self-assembled monolayer hy-
		brid gate dielectrics
	6.4	Surface properties
		6.4.1 Surface roughness and Surface energy
		6.4.2 Morphology of SAMs as determined by the chain length of the individual molecules
	6.5	Growth and morphology of organic semiconductors on self-assembled mono-
	-	layers
		6.5.1 Growth of the organic semiconductor DNTT on different surfaces . 123
		6.5.2 Morphology of organic semiconductors DNTT and DPh-BTBT on
		phosphonic acid SAMs of different chain lengths $\ldots \ldots \ldots \ldots 131$

	6.6	Experimental	134	
	6.7	Summary and Outlook	138	
7	 6.7 Stal ultr 7.1 7.2 7.3 7.4 	Summary and Outlook	138 f 139 143 143 144 146 149 150 150 152 153 154 157 158	
	7.5	7.4.2 TFT fabrication and characterization	$\begin{array}{c} 158 \\ 159 \end{array}$	
8	Cor	nclusion	161	
Α	Ado	ditional characterization of DNTT films	167	
В	Opt	timizing plasma oxidation for aluminum oxide films	171	
С	Opt tion	timizing the substrate temperature during semiconductor deposi- n for different substrates	175	
D	Opt SAI	timizing combinations of the organic semiconductor/phosphonic aci M	d 181	
Bi	Bibliography 185			
A	cknov	wledgements	215	

List of Figures

2.1	Carbon chemistry for organic semiconductors	8
2.2	High-performance small-molecule organic semiconductors for p-channel	10
2.2	organic transistors	10
2.3	Schematic description of charge-transport models	13
2.4	Schematic representation of the different molecular orientations of the	
	organic semiconductor molecule pentacene	15
2.5	Schematic representation of organic TFT architectures	17
2.6	Schematic representation of an organic TFT	17
2.7	Energy levels for p-channel and n-channel organic TFTs	18
2.8	Current-voltage characteristics of a p-channel organic transistor with dif-	
	ferent operation regions	19
2.9	Extraction of the charge-carrier mobility and the threshold voltage from	
	measured transistor characteristics	23
2.10	Comparison of the frequency-dependent areal capacitance of high- k and	
	low- k dielectric films $\ldots \ldots \ldots$	28
2.11	Schematic representation and an example of a frequency-dependent ca-	
	pacitance measurement	29
2.12	Schematic representation of the orientation of organic semiconductor molecul	es
	on different substrates	32
2.13	Atomistic processes during film growth by vapor deposition	34
2.14	Growth modes for thin-film growth	36
2.15	Schematic representation of an Atomic Force Microscopy (AFM) mea-	
	surement and the R_{RMS} surface roughness parameter	40
2.16	Schematic representations of (a) Scanning Electron Microscopy (SEM)	
	and (b) Transmission Electron Microscopy (TEM) techniques	41
2.17	Schematic representation of a grazing incidence X-ray diffraction mea-	
	surement	42
91	Schematic representation of malecular accompling into calf accompled mana	
0.1	Schematic representation of molecules assembling into sen-assembled mono-	17
าก	Commonly used melecules for self assembled monoleums (CAMs) in rate	41
3.2	dielectrics in organic TETs	18
? ?	Assembly machanisms of phosphonic acid malaculas on an aluminum orida	40
ე.ე	surface	10
34	Schematic representation and comparison of thick and thin ovide di	49
0.4	electrics with hybrid gate dielectrics composed of a thin oxide layer and	
	a self-assembled monolaver	52
	a son assembled menorager	04

3.5	Transfer characteristics and gate currents of DNTT TFTs fabricated with either (a) a SAM dielectric (without oxygen plasma), (b) a bare-AlO _x dielectric (without SAM) and (c) a hybrid AlO_x/SAM gate dielectric. The molecule chosen to form the SAM is $HC_{14}PA$	54
3.6	Transfer characteristics and gate currents of DPh-BTBT TFTs fabricated with either (a) a SAM dielectric (without oxygen plasma), (b) a bare-AlO _x dielectric (without SAM) and (c) a hybrid AlO_x/SAM gate dielectric. The melacula characteristic form the SAM is EC $_{\rm PA}$	55
3.7	Schematic cross-section of the organic TFTs fabricated in this thesis chap- ter along with the chemical structures of the organic semiconductors and the different phosphonic acid molecules chosen to form SAMs in the hy- brid gate dialogating	57
	brid gate dielectrics	97
4.1	Importance of the plasma process to fabricate an ultrathin AlO_x film for gate dielectric in organic transistors	61
4.2	Thickness variation of plasma-grown AlO_x films	62
4.3	Capacitance of plasma-grown $\mathrm{AlO}_{\mathrm{x}}$ and hybrid $\mathrm{AlO}_{\mathrm{x}}/\mathrm{SAM}$ dielectrics $~$	64
4.4	Calculation of the relative permittivity of plasma-grown AlO_x	66
4.5	Current density through capacitors with plasma-grown $\rm AlO_x$ dielectric films	67
4.6	Electrical properties of TFTs with a bare- AlO_x gate dielectric for different plasma parameter combinations	69
4.7	Electrical properties of TFTs with a hybrid AlO_x/SAM gate dielectric for different plasma parameter combinations	69
4.8	Charge-carrier mobilities for organic TFTs with plasma-grown AlO_x gate	70
4.9	dielectrics as a function of the plasma process parameters \dots \dots \dots Surface energy of bare-AlO _x and hybrid AlO _x /SAM dielectrics as a func-	70
4.10	tion of plasma process parameters $\dots \dots \dots$	71
	tion of plasma process parameters	72
4.114.12	Morphology of DNTT films on plasma-grown AlO_x dielectric films Correlation of surface properties of AlO_x/SAM hybrid gate dielectrics and	74
4.13	charge-carrier mobilities of organic TFTs	75
4.14	hybrid gate dielectrics $\dots \dots \dots$	76
	process parameters	78
5.1	Thickness of plasma-grown AlO_x films as measured by transmission electron microscopy (TEM)	83
5.2	Schematic representation of the operating principle of Electron Energy Loss Spectroscopy (EELS)	87
53	High-loss region of the EELS spectrum and the corresponding spatially	01
5.5	resolved elemental analysis	89
0.4	resolved elemental analysis	90

5.5	X-ray spectrum obtained in conjunction with the EELS measurement and	
	the corresponding elemental information	. 91
5.6	EEL spectra corresponding to plasma-grown AlO_x films averaged and smoothened using principle component analysis	02
57	Ouslitative analysis by comparing measured and reference FFL spectra	. 32
5.7	Schematic representation of the seture for an atom probe tomography	. 95
5.0	schematic representation of the setup for an atom probe tomography	
	the basic measurement principle	05
5.0	Schematic representation and SEM images depicting the fabrication of	. 90
0.9	samples for atom probe tomography using a focused ion beam	. 98
5.10	SEM images depicting the specimen sharpening process using a focused	
	ion beam	. 99
5.11	Mass spectrum depicting the different ionized fragments identified during	
	the atom probe measurement	. 100
5.12	Reconstructed view of $Al/AlO_x/Al$ films post an atom probe measure-	
	ment and the corresponding compostion analysis	. 101
5.13	Reconstructed view of $Al/AlO_x/Cr$ films post an atom probe measure-	
	ment and the corresponding compostion analysis	. 102
6.1	Alkyl and fluoroalkyl phosphonic acid molecules of different chain lengths	
	selected to form SAMs in hybrid gate dielectrics	. 107
6.2	Gate-dielectric capacitance of AlO _x /SAM hybrid gate dielectrics as a	
	function of the phosphonic acid molecule chain length	. 111
6.3	Electrical characteristics of DNTT and DPh-BTBT TFTs with AlO _x /SAM	
	hybrid gate dielectrics for phosphonic acid molecules of different chain	
	lengths used to form the SAM	. 113
6.4	Threshold voltage of organic TFTs with AlO _x /SAM hybrid gate dielectrics	
	as a function of the phosphonic acid molecule chain length	. 114
6.5	Gate-leakage current of organic TFTs with AlO _x /SAM hybrid gate di-	
	electrics as a function of the phosphonic acid molecule chain length	. 115
6.6	Charge-carrier mobility of organic TFTs with AlO _x /SAM hybrid gate	
	dielectrics as a function of the phosphonic acid molecule chain length	. 116
6.7	Subthreshold swing of organic TFTs with AlO _x /SAM hybrid gate di-	
	electrics as a function of the phosphonic acid molecule chain length	. 117
6.8	Current on-off ratio of organic TFTs with AlO _x /SAM hybrid gate di-	
	electrics as a function of the phosphonic acid molecule chain length	. 118
6.9	Surface roughness of AlO_x/SAM hybrid dielectrics as a function of the	
	phosphonic acid molecule chain length	. 119
6.10	Surface energy of AlO_x/SAM hybrid dielectrics as a function of the phos-	
	phonic acid molecule chain length	. 120
6.11	Schematic representation of the molecular ordering and packing density	
	of SAMs dependent on the chain length of the individual molecules	. 121
6.12	Growth process of the semiconductor DNTT on a bare- AlO_x surface im-	
	aged by scanning electron microscopy	. 124
6.13	Growth process of the semiconductor DNTT on an $AlO_x/HC_{14}PA$ surface	
	imaged by scanning electron microscopy	. 126
6.14	Growth process of the semiconductor DNTT on an $AlO_x/HC_{18}PA$ surface	
	imaged by scanning electron microscopy	. 126

6.15	Growth process of the semiconductor DNTT on an $AlO_x/FC_{14}PA$ surface imaged by scanning electron microscopy	. 127
6.16	Semiconductor morphology of DNTT with nominally 2.5 nm, 5 nm and	. 121
6 17	25 nm films on three different alkyl phosphonic acid SAMs	. 131
0.17	nm films on three different fluoroalkyl phosphonic acid SAMs	. 132
6.18	Schematic cross-section of the organic TFTs fabricated in this thesis chap-	-
	ter along with the chemical structures of the organic semiconductors and	
	the different phosphonic acid molecules chosen to form SAMs in the hy-	12/
6.19	Schematic representation of the experimental setup for observing the	. 194
0.20	growth process of the organic semiconductor on different gate dielectric	
	surfaces	. 137
7.1	Thin-film morphology of ultrathin DNTT films with different thicknesses	
	to determine the minimum thickness required for fabricating ultrathin	
79	organic TFTs	. 143
1.2	observed over 12 hours	. 144
7.3	Evolution of the morphology of ultrathin DNTT films on different dieletric	
	surfaces observed over 12 hours	. 145
7.4	Influence of the nominal thickness of vacuum-deposited DNTT films on the evolution of the electrical characteristics of TFTs based on these films	3 147
7.5	Morphology of a thick DNTT film observed over 12 hours	. 148
7.6	Effect of cryogenic cooling on the spontaneously occurring morphological	
	changes in ultrathin DNTT films	. 149
7.7	Effect of in-situ encapsulation of ultrathin DNTT films with PTFE and its influence on the evolution of TFT characteristics	151
7.8	Effect of in-situ encapsulation of ultrathin DNTT films with C_{60} and its	. 101
	influence on the evolution of TFT characteristics	. 152
7.9	Effect of in-situ encapsulation of ultrathin DNTT films with TiOPc and its influence on the evolution of TET characteristics	154
7.10	Summary of the influence of in-situ encapsulation of ultrathin DNTT	. 104
	TFTs and the evolution of the charge-carrier mobility	. 155
A 1	Thin-film morphology and orientation of DNTT molecules on a hybrid	
11.1	gate dielectric surface as imaged by AFM	. 167
A.2	Specular X-ray diffractogram depicting the molecular arrangement of a	
A 9	DNTT film deposited on a hybrid gate dielectric surface	. 168
A.3	DNTT films measured immediately after TFT fabrication	. 169
B.1	Electrical (transfer) characteristics of DNTT TFTs with plasma-grown	
F -	AlO_x gate dielectrics for different plasma process parameters $\ldots \ldots$. 172
В.2	Electrical (output) characteristics of DNTT TFTs with plasma-grown	172
	mox base diologines for different plasma process parameters	. ттэ

B.3	Thin-film morphologies of DNTT films deposited on to plasma-grown AlO_x and AlO_x/SAM hybrid gate dielectric surfaces for different plasma process parameters and their correlation with the charge-carrier mobilities of DNTT TFTs based on those films $\ldots \ldots \ldots$
C.1	Electrical characteristics of DPh-BTBT TFTs on different substrates de- pendent on the substrate temperature during semiconductor deposition . 176
C.2	Orientation of DPh-BTBT molecules and thin-film morphology of DPh- BTBT films deposited on a hybrid gate dielectric on different substrates
	dependent on the substrate temperature during semiconductor deposition 178
C.3	Orientation of DPh-BTBT molecules and thin-film morphology of DPh- BTBT films deposited on an AlO_x dielectric on different substrates de-
	pendent on the substrate temperature during semiconductor deposition . 179
D.1	Electrical characteristics of DNTT TFTs with different gate dielectrics 182
D.2	Comparison of DNTT and DPh-BTBT TFTs with AlO_x/SAM hybrid gate dielectrics with fluoroalkyl phosphonic acid SAMs to determine the
	optimum semiconductor-SAM combination
D.3	Orientation of DPh-BTBT molecules on different hybrid AlO_x/SAM di-
	electrics on PEN substrates
D.4	Electrical characteristics of DPh-BTBT TFTs with an AlO_x/SAM hybrid
	gate dielectric with different fluoroalkyl phosphonic acid SAMs 184

List of Tables

4.1	Thickness of AlO_x films (with error margins) prepared by different com-	
	TEM	63
		00
6.1	Summary of literature reports investigating the influence of chain length	
	in phosphonic acid molecules in hybrid gate dielectrics	109
6.2	Correlation between surface properties of the gate dielectric and organic	
	semiconductor growth and electrical properties of organic TFTs	128

Abbreviations

- AlO_x aluminum oxide, unknown stoichiometry
- $\mathbf{a}\textbf{-}\mathbf{Si:}\mathbf{H}$ hydrogenated-amorphous silicon
- **AFM** atomic force microscopy
- AMLCD active-matrix liquid-crystal display
- $\mathbf{C_{60}} \ \mathrm{Buckminster} \ \mathrm{fullerene}$
- DNTT dinaphtho[2,3-b:2',3'-f]thieno[3,2-b]thiophene
- DPh-BTBT 2,7-diphenyl[1]benzothieno[3,2-b][1]benzothiophene
- **EELS** electron energy loss spectroscopy
- ${\bf FIB}\,$ focused ion beam
- HHCF height-height correlation function
- HOMO highest occupied molecular orbital
- LUMO lowest unoccupied molecular orbital
- **MOSFET** metal-oxide semiconductor field-effect transistor
- **ODPA** octadecylphosphonic acid
- **ODTS** octadecyltrichlorosilane
- ${\bf OFET}\,$ organic field-effect transistor

- **OLED** organic light-emitting diode
- $\mathbf{OTS} \ \text{octadecylsilane}$
- **PEN** poly(ethylene)2,6-naphthalate
- ${\bf PTFE} \hspace{0.1 cm} {\rm polytetrafluoroethylene}$
- $\mathbf{R_{RMS}}$ root-mean-square surface roughness
- ${\bf SAM}$ self-assembled monolayer
- ${\bf SEM} \ {\rm scanning} \ {\rm electron} \ {\rm microscopy}$
- **TEM** transmission electron microscopy
- ${\bf TFT}\,$ thin-film transistor
- ${\bf TiOPc}\,$ titanyl phthalocyanine
- ${\bf ToF-SIMS}$ time-of-flight secondary ion mass spectrometry

Symbols

μ	charge-carrier mobility	m^2/Vs
μ_0	intrinsic charge-carrier mobility	m^2/Vs
σ	electrical conductivity	S/m
θ	liquid contact angle	0
ε_0	absolute permittivity	F/m
L	channel length	m
W	channel width	m
k	dielectric constant (or relative permittivity)	
\mathbf{C}	capacitance	\mathbf{F}
$\mathbf{C}_{\mathrm{diel}}$	area-normalized gate-dielectric capacitance	F/m^2
$\mathbf{E}_{\mathbf{F}}$	Fermi energy level	eV
I_{G}	gate current	А
I_D	drain current	А
$k_{\rm B}$	Boltzmann constant	$\mathrm{m}^2\mathrm{kg}~\mathrm{s}^{\text{-}2}\mathrm{K}^{\text{-}1}$
$\mathrm{T}_{\mathrm{sub}}$	substrate temperature	$^{\circ}\mathrm{C}$
V_{th}	threshold voltage	V
V_{DS}	drain-source voltage	V
V_{GS}	gate-source voltage	V

Chapter 1

Introduction

The widespread use of digital electronics and the undeniable reliance on them are an inseparable part of human life in today's hyper-connected, fast-paced world. Coupled with the powers of the internet, there is a perpetual demand for frequent breakthroughs which lead to a larger bandwidth, newer technological advancements and which branch out into alternate, diverse applications. In such an environment, there is a continuous effort to explore and develop unique innovations that will improve the quality of life in previously unthinkable ways.

The discovery of semiconducting properties in materials and fabrication of electrical devices employing them has been one of the most important developments of the 20th century. The transistor is the fundamental building block of modern electronic systems and is used to amplify and switch electrical signals using a semiconductor as the active material. The electronics industry has come a long way to develop the modern gadgets and advanced technology available today, following a long history of decades of intense research and development. The first ideas of the transistor date back to Julius E. Lilienfeld [1] (1930) and to Oskar Heil [2] (1935), who both patented devices which resemble modern-day field-effect transistors. They also noted that the active material in such a device, key to its functionality, would have to be a semiconductor, but were unfortunately unable to fabricate a working device, most likely due to the lack of high-quality semiconductors. A major breakthrough was achieved with the first successful and reproducible demonstrations of the transistor by John Bardeen and Walter Brattain [3] (1947) and William Shockley [4] (1948) at Bell Labs, for which they later jointly received the 1956 Nobel Prize in Physics for "their research on semiconductors and their discovery

of the transistor effect". The transistors fabricated were, respectively, a point-contact transistor and a bipolar-junction transistor, which are both almost obsolete in modernday electronics. The next most impactful development occurred with the invention of the metal-oxide-semiconductor field-effect transistor (MOSFET) by Martin Mohamed Atalla [5](1960) and Dawon Kahng [6] (1963). The characteristic feature in the MOS-FET was the surface passivation of the crystalline silicon wafer with an insulating layer of controlled, thermally-grown silicon dioxide [7], which gave MOSFETs several advantages such as a higher degree of scalability, lower power consumption and a higher integration density as compared to bipolar-junction transistors. Due to all these benefits, the MOSFET quickly superseded all its predecessors in terms of its performance and made it possible to build high-density integrated circuits. The MOSFET was the first truly compact transistor that was miniaturized and mass produced and became the most common type of transistor to be used in computers, electronics, and communications, having a profound impact on the development of digital electronics. Although other semiconductor materials such as germanium and gallium arsenide have better electrical properties than silicon, the ability to form high-quality semiconductor-insulator interfaces makes silicon the champion semiconductor material in modern electronics. However, the material properties and processing conditions required to fabricate silicon-based MOSFETs make them unsuitable for applications envisioned outside of the current conventional single-crystalline silicon electronics domain. Alongside, another major breakthrough included the development of the thin-film transistor (TFT) which was first developed by Paul Weimer in 1962 using cadmium selenide (CdSe) as the semiconductor [8]. In this device configuration, all the materials are deposited as thin films onto an independent surface, thus broadening the choice of the substrate beyond silicon wafers to those substrate materials compatible with the fabrication process. This invention was followed by pioneering work by T. Peter Brody and coworkers in the development of indium arsenide (InAs) [9] and CdSe TFTs, which were later used to demonstrate the most common application of the TFT, the active-matrix liquid-crystal display (AMLCD) [10]. In 1979, P.G. le Comber, W.E. Spear and A. Ghaith developed the first functional hydrogenatedamorphous silicon (a-Si:H) TFT with a silicon nitride (SiN) gate dielectric [11], after which it was soon realized that a-Si:H TFTs are better suited for large area AMLCDs, which went on to be mass produced as a common display technology. Apart from amorphous silicon, polycrystalline silicon was also extensively researched for fabricating TFTs for display applications. Low-temperature polycrystalline silicon (LTPS) is crystallized

at a relatively lower temperature (~ 450 °C) compared to the traditional temperature of 900 °C and above for single-crystalline silicon. LTPS TFTs are commonly used to drive displays based on organic light-emitting diodes (OLEDs) and are beneficial for large glass displays which are likely to undergo deformation at elevated temperatures. When the semiconductor is deposited as a thin film for TFT applications, it lacks the high degree of crystallinity of the active material in MOSFETs, and thus has a relatively lower performance. However, the independent choice of the substrate, separate from the semiconductor material, makes TFTs the ideal choice for large-area applications with low-frequency requirements, such as in displays.

Apart from silicon, other semiconductor technologies are also being explored and investigated, e.g. oxide semiconductors (e.g. ZnO, In_2O_3) [12–14], transition metal dichalcogenides (e.g. WS₂, MoS₂) [15–17], carbon nanotubes [18–20], nanowires (e.g. Si, GaAs, InAs/InP, In_2O_3) [21–26], and of course, central to this thesis, organic materials.

Although the synthesis of conductive organic materials (carbon-hydrogen based materials) can be dated back to the late 19th century [27, 28], the first instances of observing electrical conductivity in organic materials only occurred in the 1950s and 1960s [29,30]. The first real breakthrough was achieved in 1977 when Alan J. Heeger, Alan MacDiarmid, Hideki Shirakawa and coworkers discovered high-conductivity polymers by synthesizing oxidized iodine-doped polyacetylene [31]. They were jointly awarded the 2000 Nobel Prize in Chemistry which catapulted progress in the field of organic electronics.

Since inorganic semiconductors like silicon are the most widely used materials in modern electronics today, the relatively newer organic semiconductors are often viewed in comparison to them, to recognize the unique benefits provided by the latter. Inorganic semiconductors are relatively very brittle and require advanced processing conditions in order to maintain the high quality of semiconductors necessary for high performance. The temperature at which different semiconductors are processed during fabrication of TFTs and other devices is the most important and distinguishing factor to be considered. All popular semiconductors prevalent today are processed at relatively high temperatures, which limits the choice of substrates or surfaces on which devices and circuits can be fabricated, to only those which can withstand such high temperatures and limits the functionality range of the substrates available. In contrast, organic-semiconductor materials can be processed at much lower temperatures, which are much closer to room temperature. This enables the possibility of using a wide variety of alternative and new substrates, which are not compatible with high-temperature processing, but can provide unique advantages when used to fabricate organic-semiconductor devices. For example, plastic foils can be used for mechanical flexibility in bendable or rollable applications [32–34], glass substrates can provide optical transparency [35,36], and electronic devices and circuits can potentially be fabricated on paper [37–40], textiles and fabrics [41,42] and even human skin [43,44]. Thus, a widened choice of possible substrates opens up a new dimension of possible applications with different advantages such as large-area fabrication, roll-to-roll processing along with high-throughput fabrication.

Considering the different mechanism of charge conduction in organic materials as compared to their inorganic counterparts, the electrical conductance of organic semiconductors is usually much smaller, which limits their use for high-performance circuit design. Moreover, the weaker bonding mechanisms in organic materials make them less stable against electronic noise and thermal fluctuation. There is also a higher variation between individual devices which results in a lower integration density. All these factors make it impossible for organic semiconductors to replace silicon in MOSFETs, and that is an unrealistic goal for the future development of organic semiconductors in electronic devices. Instead, the fundamental idea is the targeted development of niche applications benefiting from the unique advantage of organic semiconductors, i.e. the availability of alternative substrates to fabricate devices and circuits on.

Another major distinguishing factor with the development of organic semiconductors for electronic devices is the wide variety of materials which are chemically synthesized and investigated, instead of a single material system. This gives the possibility of selectively synthesizing and tailoring the properties of semiconductor materials depending on the target application along with other materials or processes used in device fabrication.

The emerging field of organic electronics can be categorized based on the target applications of organic materials: 1) organic light-emitting diodes (OLEDs) [45–47], 2) organic solar cells [48–50] and 3) organic field-effect transistors (OFETs). OLEDs have been extensively researched and have already been commercialized to be used in digital displays in all major electronic gadgets like television screens, computer monitors, smartphones, etc. Organic solar cells are promising, competitive devices currently facing certain hurdles towards being manufactured commercially and research is ongoing in improving their efficiencies, stability and overall performance.

In contrast, organic field-effect transistors are still much further behind in terms of research and development towards commercial products. However, a few small-scale applications have been rolled out into the market. Although significant work was done in the 1980s for field-effect measurements in organic films, their performance was still quite poor, quantified usually in terms of the charge-carrier mobility. It was only in the 1990s, after significant efforts, that the performance of organic transistors was improved and charge-carrier mobilities became comparable to those of a-Si:H TFTs ($\sim 0.1-1 \text{ cm}^2/\text{Vs}$). Today, organic transistors have consistently reached significantly larger charge-carrier mobilities, in the range of 1-10 cm²/Vs, and are promising candidates for a variety of potential applications. However, there are still significant obstacles on that path in terms of device stability, reliability, reproducibility and uniformity. There is also a major gap in the fundamental understanding of the operation principles of organic transistors. Additionally, a large portion of the research efforts are also consumed in finding the optimum combination of materials, process parameters and device configurations from the wide selection variety available.

One of the primary target applications for OFETs is in the active-matrix OLED display [51], which currently uses TFTs based on LTPS and indium gallium zinc oxide (IGZO) to drive the backplane. Apart from that, other potential applications include conformable sensors [52, 53], foldable displays (e.g. e-books or e-paper) [54], soft or smart bionic and robotic devices [55–57], wearable or implantable medical devices [58], etc. Most of these target applications, if not all, are designed with a portable or wearable nature in mind and will be powered by small batteries or solar cells to ensure their lightweight nature. For this very reason, the organic transistors driving these devices have a key requirement to operate at low supply voltages. The operating voltage of a transistor is largely dictated by the capacitance of the gate dielectric, making the gate dielectric a crucial component in their development. The importance of the gate dielectric is also extended to the interface with the organic semiconductor and can have a huge impact on the overall functioning of the device. Properties of the organic-semiconductor component of the organic transistor are a subject of major focus, since charge transport in the device takes place in this region. A wide range of material and film properties are instrumental towards effective charge transport in the device, and establishing a connection between the material properties of the organic semiconductor and electrical properties of the organic transistor is of vital importance. The aspect of film thickness for these two components of the organic transistor, viz. the gate dielectric and the organic semiconductor, is a focal point in this work.

This thesis focuses on investigating ultrathin, high-capacitance, hybrid gate dielectrics and the stability of ultrathin semiconductor layers for low-voltage

organic thin-film transistors. The two components of the hybrid gate dielectric, i.e. the inorganic gate oxide and the organic self-assembled monolayer, and different aspects pertaining to them, are studied. Extending to the gate dielectric-semiconductor interface, there is a focus on the growth and morphology of organic semiconductors and the stability of ultrathin semiconductor layers at this interface.

This thesis is organized as follows. Chapter 2 provides a theoretical framework for organic semiconductors, charge transport in organic thin-film transistors and their basic operation. In addition, an introduction to dielectric properties as well as nucleation and growth properties of organic semiconductors are discussed. A basic description of all the characterization techniques used in the thesis is also detailed. Chapter 3 introduces the concept of hybrid gate dielectrics along with the significance and operating role of each component of the hybrid gate dielectric. Chapter 4 investigates the process parameters used to prepare the gate oxide, i.e. aluminum oxide (AlO_x) , the first component of the hybrid gate dielectric and investigates the optimum parameter combination required to fabricate an ultrathin and dense AlO_x film as the gate oxide required for high-performance hybrid dielectrics. The relationship between surface properties of the AlO_x film and the electrical properties of the organic transistor is also established. Chapter 5 describes an in-depth characterization of the AlO_x component of the gate dielectric using advanced techniques with an atomic-scale resolution. The AlO_x films investigated in Chapter 5 are prepared using the optimum process parameters as determined in Chapter 4 and are characterized for the thickness as well as the composition of the AlO_x films. Chapter 6 investigates the second component of the hybrid gate dielectric, i.e. the self-assembled monolayer (SAM), particularly the effect of changing the thickness of the SAM using two different types of molecules used to form the SAM, and its effects on the performance of organic thin-film transistors. Chapter 6 also examines the nucleation and growth of organic-semiconductor films on gate-dielectric surfaces and the influence of different dielectric surface properties. Thereafter, Chapter 7 investigates ultrathin semiconductor layers, time-dependent morphological changes occurring in them and their stabilization using either cryogenic cooling or in-situ encapsulation towards the fabrication of ultrathin organic TFTs. Chapter 8 provides a summary of the work in this thesis and provides an outlook for the future development of organic transistors.

Chapter 2

Theory

2.1 Organic semiconductors

The unique chemistry of elemental carbon facilitates the formation of a large variety of hydrocarbon compounds, the study of which constitutes the field of organic chemistry. The family of aromatic compounds, identified by their characteristic conjugated systems, i.e. alternating single and double bonds are particularly important for the electronic properties of organic molecules. The ground state configuration of electrons in a carbon atom can be represented as: $1s^2 2s^2 2p^2$, with the four valence electrons located in the doubly-occupied $2\mathrm{s}^2$ and the singly-occupied $2\mathrm{p}_x$ and $2\mathrm{p}_y$ atomic orbitals. However, carbon atoms also have the ability to form hybridized electronic configurations, in which atomic orbitals in the ground state can be imagined to be combining together and modifying themselves to form hybrid orbitals. For example, in an sp^2 hybridized configuration found in carbon atoms in conjugated systems, one 2s and two 2p ground state orbitals combine to form three sp^2 hybrid orbitals, as depicted in Fig.2.1(a). The four valence electrons of the carbon atom in an sp^2 hybridized configuration are now distributed in three singly-occupied sp^2 hybrid orbitals and one unhybridized $2p_z$ orbital. The three sp^2 hybrid orbitals are located trigonally in a single plane, while the $2p_z$ orbital is located perpendicular to this plane, as shown schematically in Fig.2.1(b). In a typical conjugated system where carbon atoms are in the sp^2 hybridized state, each carbon atom forms two single bonds and a double bond with its neighboring atoms. The single bond is formed by the overlap of sp² orbitals, also known as a σ (sigma) bond, and is localized to the overlap region of the sp^2 orbitals. A double bond consists of a



Figure 2.1: (a) Electronic configuration for a carbon atom in the ground state and after sp² hybridization (b) Trigonal geometry of atomic orbitals in an sp² hybridized carbon atom (c) σ - π bonding in a benzene molecule and delocalization of π electrons, visualized as a π electron cloud (d) Resonance structures of benzene (e) Energy level diagram depicting the energy-level splitting during bonding in a benzene molecule and the formation of HOMO-LUMO energy levels

 σ bond by the overlapping sp² orbitals and a π (pi) bond formed due to the overlap of $2p_z$ orbitals.

The molecule benzene is the simplest aromatic compound and the main building block for small molecules used in organic electronics. The six carbon atoms in a benzene molecule form twelve σ bonds with the overlap of sp² orbitals and three π bonds by the overlap of $2p_z$ orbitals. Since the $2p_z$ orbitals are located perpendicular to the plane of the atoms, the overlap between the two lobes occurs laterally and there is zero electron density at the shared nodal plane, passing through the two bonded nuclei. As a result, compared to a σ bond, the π bond overlap is relatively weaker and the electrons in the π bonds (or π electrons) are largely delocalized among all six $2p_z$ orbitals [59]. Thus, the π electrons in a benzene molecule can be visualized as an electron cloud delocalized in the planes above and below the plane of the atoms, as shown schematically in Fig.2.1(c). The molecule benzene can be represented as either of the resonance structures shown in Fig.2.1(d) to signify delocalization of π electrons within the molecule. This delocalization of π electrons is essential to the charge transport properties of aromatic compounds. During the formation of the σ bond, the strong overlap between the sp² hybrid orbitals causes a large splitting of the sp² energy level into a bonding σ and an antibonding σ^* molecular orbital. In comparison, the energy level splitting during the formation of the π bond is much smaller because of the weak overlap of $2p_z$ orbitals, and the resulting π and π^* molecular orbitals have a much smaller energy difference. In benzene, the $2p_z$ orbital overlap leads to energetic splitting into three bonding π orbitals which are
occupied by electrons and three antibonding π^* orbitals which are vacant. The π molecular orbital with the highest energy is also called as the Highest Occupied Molecular Orbital (HOMO) and is analogous to the valence band in inorganic solids. Similarly, the π^* molecular orbital with the lowest energy is called the Lowest Unoccupied Molecular Orbital (LUMO), analogous to the conduction band. By extension, the energy difference between the HOMO and LUMO energy levels is analogous to the band gap in inorganic solids. The formation of molecular orbitals due to the energetic splitting of atomic orbitals upon bonding and the subsequent HOMO-LUMO energy levels can be seen schematically in Fig.2.1(e). The HOMO-LUMO energy difference in benzene is approximately 6.8 eV [60], making it an insulator. In the family of acenes, as the aromatic chain length increases with more benzene rings fused together, the number of $2\mathbf{p}_z$ orbitals which overlap with each other also increases. This leads to an increased energetic splitting between the bonding and further between the antibonding molecular orbitals. As a result, a larger number of π and π^* molecular orbitals reduces the HOMO-LUMO energy difference and leads to an increased delocalization of π electrons [60]. Pentacene, which was initially one of the most popular and prominent organic semiconductors, has five benzene rings fused together and a HOMO-LUMO energy difference of approximately 2.2 eV [61-63].

The two primary classes of organic semiconductors are small molecules and polymers [64]. Small molecules, such as accenes, typically have a low molar mass which makes it easy to purify them and they are most commonly deposited by gas-phase processes such as thermal sublimation. They can also be synthesized as single crystals to study intrinsic charge transport processes [65–68]. Alternatively, their properties can be tailored with functional groups such as side chains to make them soluble for solution-processed deposition techniques. Some of the commonly used, high-performance small-molecule organic semiconductor molecules used to fabricate p-channel (hole-conducting) organic transistors are shown in Fig.2.2. The second class of organic semiconductors are polymers, which typically have a very large molecular weight and are commonly processed exclusively from solution [69–71]. The two organic semiconductors employed in this thesis are:

DNTT : dinaphtho[2,3-b:2',3'-f]thieno[3,2-b]thiophene
 (HOMO-LUMO energy difference = 3 eV) [72]

DPh-BTBT: 2,7-diphenyl[1]benzothieno[3,2-b][1]benzothiophene
 (HOMO-LUMO energy difference = 3.2 eV) [73]



Figure 2.2: Commonly used high-performance small-molecule organic semiconductors for p-channel organic transistors (a) pentacene (b) TIPS-pentacene (c) DPVAnt (di(phenylvinyl)anthracene) (d) DNTT (dinaphtho-[2,3-b:2',3'-f]thieno[3,2b]thiophene) and its derivatives (shown in blue) (e) C_{10} -DNTT (f) DPh-DNTT (g) BTBT (benzothieno[3,2-b][1]benzothiophene) and its derivatives (shown in red) (h) C_{8} -BTBT (i) DPh-BTBT

2.2 Charge transport in organic semiconductors

The delocalized nature of π orbitals in a conjugated carbon system allows for charge carriers to be transported along the molecule, i.e. *intra*molecular charge transport. However, since the dimensions of organic transistors are much larger than individual organic semiconductor molecules, *inter*molecular charge transport needs to be established as well. Essentially, efficient charge transport in the conducting channel of the organic transistor depends on the degree of π molecular orbital overlap between neighboring molecules over the length of the channel [74]. The various factors contributing to charge transport, the different attempts at modeling charge-transport behavior and strategies to enhance it based on this knowledge are further explored in this section.

In order to describe different aspects of charge transport in organic semiconductors throughout this thesis, it is important to define certain fundamental parameters associated with it. Upon applying an external electric field E, charge carriers are accelerated to a drift velocity $v_{\rm D}$, and the parameter used to describe the efficiency with which electronic charges are transported through the solid is defined as the charge-carrier mobility μ :

$$\mu = \frac{\partial v_{\rm D}}{\partial E}$$

The electrical-current density j as a result of the transport of charge carriers is determined by the charge-carrier density n and the drift velocity $v_{\rm D}$:

$$j = e \cdot n \cdot v_{\rm D}$$
$$= e \cdot n \cdot \mu \cdot E$$

where e is the elementary electronic charge. The electrical conductivity σ of the system, which is the ratio of the current density to the applied electric field is defined as:

$$\sigma=\mu\cdot n\cdot e$$

The conducting channel in an organic transistor is made up of several organic semiconductor molecules organized in a more or less ordered fashion, ideally with a suitable processing technique to encourage the overlap of π molecular orbitals of neighboring molecules and subsequently, efficient charge transport. The overlap of molecular orbitals during bonding leads to the splitting of discrete orbital energies, and depending on the number and proximity of neighboring molecules, the dispersion of energy levels may form an energy continuum, or an energy band. In inorganic semiconductors such as single-crystalline silicon and gallium arsenide (GaAs), atoms are bound by relatively strong covalent bonds resulting in very strong electronic orbital overlap. As a result, electronic wave functions (Bloch waves) are delocalized over large distances in the lattice and broad energy bands are formed and the charge-transport mechanism can be described as band transport. During band transport, the charge-carrier mobility decreases with increasing temperature due to phonon scattering, and transport is characterized by the temperature dependence of mobility following a power law behavior:

$$\mu \propto T^{-n} \dots (n > 0)$$

In contrast, organic semiconductors form largely polycrystalline or even amorphous layers with molecules bound by relatively weak Van der Waals forces with greater intermolecular distances and weaker intermolecular orbital overlaps. Moreover, organic semiconductor thin films are characterized by a lot of defects, impurities and most importantly, grain boundaries which break the periodicity of the lattice, and formation of energy bands or band transport are highly unlikely. Instead, all impurities and defects form localized states in the HOMO-LUMO energy gap, which act as charge-carrier traps. Charge transport in organic semiconductor films is therefore largely dominated by the presence and distribution of localized states.

In the case of completely amorphous organic solids, charge transport occurs by a *hopping* mechanism, where charge carriers are transported through the solid by hopping from one localized state to another. In this case, the charge-carrier mobility increases with increasing temperature and the temperature dependence of mobility in this case follows an Arrhenius equation:

$$\mu \propto exp \frac{-\Delta E_a}{k_B T}$$

where $\Delta E_{\rm a}$ is the activation energy, i.e the energy that a localized charge carries has to absorb in order the leave the localized state and contribute to charge transport, k_B is the Boltzmann constant and T is the temperature.

Charge transport in polycrystalline solids, which is the form most organic semiconductor films assume, is thus a combination of coherent transport within a delocalized transport band (as observed exclusively in single crystals) and incoherent transport by thermallyactivated hopping between localized states (as observed in disordered or amorphous solids).

Charge Transport Models

In order to understand and favorably tailor the electrical properties of organic thin-film transistors, it is important to develop a theoretical framework of charge transport in organic semiconductor films. However, this is extremely challenging due to the complex structure-property relationships and a wide variety of available organic semiconductors [75]. Furthermore, different fabrication processes and slight variations in the process parameters form organic semiconductor films with largely varying properties and a varying degree of impurities or defects. Therefore, no single coherent model has been developed yet to completely understand and explain the charge transport in organic semiconductor films, like the band theory of transport for inorganic semiconductors. Nevertheless, certain models have been proposed which explain certain aspects of charge transport in specific conditions quite well, and are detailed below.



Figure 2.3: Schematic description of (a) the Multiple Trap and Release (MTR) and (b) the Variable Range Hopping (VRH) charge-transport models

Multiple Trap and Release (MTR)

The multiple trap and release model is best suited to systems with a relatively large degree of order present, i.e. a largely polycrystalline semiconductor thin film, with a minority presence of impurities, defects and grain boundaries. The model combines aspects of delocalized band transport and the presence of localized states. It assumes that charge transport takes place in an extended, narrow transport band formed by the large majority of delocalized states with localized states present in the HOMO-LUMO energy gap acting as trap states. Charge carriers in the transport band can be trapped in the localized states but with enough thermal stimulation can be released back into the transport band to contribute to charge transport, as shown schematically in Fig.2.3(a). This model was originally developed to describe the charge transport in amorphous or polycrystalline silicon TFTs and was later extended to organic semiconductors as well [76].

Variable Range Hopping (VRH)

The variable range hopping model is ideally suited to describe systems with increased disorder in the lattice, i.e. largely disordered or amorphous semiconductor films [77] in which a majority or all states are localized and there is no large-scale delocalization over several molecules or the entire film. In this case, charges hop from one localized state to another and are transported through the solid, as represented in Fig.2.3(b).

The temperature dependence of charge transport, or the charge-carrier mobility, is closely related to the degree of molecular order and the energy distribution of electronic states in the solid. In highly pure organic single crystals, similar to inorganic semiconductors, the charge-carrier mobility increases with a decrease in temperature. This is primarily because charge-carrier transport is predominantly coherent and the influence of phonon scattering on the mean free path of charge carriers decreases with decreasing temperature. On the other hand, in amorphous layers, the charge-carrier mobility decreases with decreasing temperature because charge transport takes place exclusively by thermally activated hopping between localized states. For organic semiconductors which are most commonly fabricated into polycrystalline layers, different temperature dependencies are observed based on the degree of molecular order and amount of defects or impurities present. The temperature dependence of the mobility observed in organic solids can be described by the following equation:

$$\mu(T) = \mu_o \left(\frac{T}{T_o}\right)^{-n} exp\left[\frac{-E_a}{k_B T}\right]$$

where E_a is the *activation energy*, i.e. the energy that a localized charge carrier has to absorb in order to leave the localized state and contribute to charge transport, and μ_0 is the intrinsic charge-carrier mobility of the semiconductor.

The morphology of the organic semiconductor films and the orientation of molecules within is thus of extreme importance as the crystallinity of the film determines the degree of π orbital overlap and efficiency of charge transport. Different approaches can be utilized towards obtaining a suitable film morphology, and for a particular organic semiconductor, optimizing the deposition process plays a very important role. The influence of different parameters during the deposition process and their suitable manipulation towards high-quality semiconductor films is explained in Section 2.5.4. A second approach entails designing and synthesizing new organic semiconductor molecules by modifying their chemical structure to suitably tailor the crystalline nature and thereby the electronic properties of organic semiconductor films. Examples of this approach are given



Figure 2.4: Schematic representation of the different molecular orientations: (a) edgeto-face and (b) face-to-face orientations of the organic semiconductor molecule pentacene

- The organic semiconductor molecule pentacene has two different possible molecular orientations, viz. 'edge-to-face' and 'face-to-face', in which the the face-to-face orientation leads to more efficient charge transport due to a larger degree of π orbital overlap. The molecule pentacene was modified by introducing bulky side groups at the 6, 13 positions to synthesize the molecule TIPS-pentacene (6,13bis(2-(tri-isopropylsilyl)ethynyl)pentacene) [78], shown in Fig.2.2. The targeted synthesis of TIPS-pentacene suppressed the occurrence of edge-to-face molecular orientations and increased the degree of π -orbital overlap, improving the chargetransport properties [79]. However, the preferred 'face-to-face' packing mode is only obtained by a very specific solution-processing technique and TIPS-pentacene cannot be used to fabricate organic TFTs in which the organic semiconductor is vacuum-deposited.
- The ionization potential refers to the energy difference between the HOMO energy level and the vacuum level, i.e. the energy necessary to withdraw an electron from the HOMO energy level to ionize the molecule. Increasing the ionization potential of an organic semiconductor increases its stability against undesired and often irreversible oxidation reactions, and the loss of delocalized molecular orbitals due to oxidation is prevented. Within the family of unsubstituted acenes, the ionization potential increases as the number of aromatic rings decreases, i.e. the size of the molecule decreases. However, a reduction in the size of the molecule is unsuitable as it significantly reduces the molecular interactions, the quality of crystal formation and the semiconducting property of the molecule. A better approach

to increase the ionization potential of organic semiconductor molecules without decreasing their size is to synthesize molecules with several weakly coupled aromatic π systems as shown in Fig.2.2, e.g. the organic semiconductor molecule DPVAnt (di(phenylvinyl)anthracene) [80,81]. Similarly, the molecules DNTT and BTBT were synthesized to obtain a larger ionization potential and hence better airstability along with an improved crystal structure as compared to pentacene. Several derivatives of DNTT (dinaphtho-[2,3-b:2',3'-f]thieno[3,2-b]thiophene) [72] and BTBT (benzothieno[3,2-b][1]benzothiophene) [73] have comparable, if not larger, charge-carrier mobilities than pentacene and are among the most popular smallmolecule organic semiconductors today, and will also feature in this work.

2.3 Organic thin-film transistors

Organic transistors are fabricated as thin-film transistors (TFTs), which are field-effect transistors with all the different components deposited sequentially as thin films. In an organic TFT, the active component is an organic semiconductor (discussed in Section 2.1), with the organic semiconductor molecules usually deposited as a polycrystalline thin film. To establish field-effect behavior, the organic semiconductor film is separated from the gate electrode by an insulating layer, called the gate dielectric. There are two other electrodes in contact with the semiconductor layer, namely the source contact and the drain contact, for the injection and extraction of charge carriers respectively. Depending on the order of depositing the various components and their relative arrangement with each other, there are four different TFT architectures possible, schematically shown in Fig.2.5: (a) bottom-gate (inverted) staggered (b) bottom-gate (inverted) coplanar (c) top-gate staggered and (d) top-gate coplanar.

The choice of the TFT structure influences the available area for charge carriers to be transported and subsequently other TFT performance characteristics. This thesis includes the two bottom-gate (inverted) structures, either in top-contact (invertedstaggered) or in bottom-contact (inverted-coplanar) architectures, shown schematically in Fig.2.5(a) and (b) respectively. The different components and the basic parameters involved in TFT operation can be seen in Fig.2.6. Typically, by applying a voltage V_{GS} to the gate electrode with respect to the grounded source electrode, an electric field is created in the semiconductor due to charge polarization in the gate dielectric. Thus,



Figure 2.5: Schematic representation of organic TFT architectures: (a) bottom-gate (inverted) staggered (b) bottom-gate (inverted) coplanar (c) top-gate staggered and (d) top-gate coplanar

the operation of the transistor resembles the charging of a capacitor which results in the accumulation of mobile charge carriers in the semiconductor layer to form a conductive path, or the conducting channel in the TFT. The formation of the channel and operation of the transistor are thus controlled by the application of the voltage V_{GS} , establishing the switching characteristic of the transistor. The flow of charge carriers from the source contact to the drain contact, i.e. the drain current I_D , is controlled by a second voltage V_{DS} applied to the drain contact with respect to the grounded source contact. The magnitude of the drain current and thereby the different operation regimes are controlled by the voltage V_{DS} .



Figure 2.6: Schematic representation of an organic TFT with the basic components, electrical parameters and formation of the conduction channel displayed

In contrast to transistors with inorganic semiconductors (e.g. MOSFETs), organic transistors are not specifically doped to facilitate the transport of one type of charge carrier in the channel. The semiconductor films are usually intrinsic, both in the channel as well as at the contacts. Instead, the type of charge carriers and thereby the polarity of the transistor is decided by the relative position of the Fermi energy level (E_F) of the contact material and the HOMO/LUMO energy levels of the organic semiconductor, as shown in Fig.2.7. If the energy barrier between the Fermi energy level of the contact and the HOMO energy level of the organic semiconductor is sufficiently small, holes can be efficiently injected and extracted and the transistor is a p-channel transistor. Similarly, if the energy barrier between the Fermi energy level of the contact and the LUMO energy level of the organic semiconductor is sufficiently small, it is easier to inject and extract electrons in the channel, and the transistor is an n-channel transistor. To avoid undesired charge injection of the opposite charge carriers at the drain contact, it is preferred to have the largest possible energy barrier between the contacts and the opposite energy level of the semiconductor. That is, for a p-channel transistor, a large energy barrier between the LUMO level and the contacts is desired and for an n-channel transistor, there should a large energy barrier between the HOMO level of the semiconductor and the contacts. For many organic semiconductors, the energy difference between the HOMO



Figure 2.7: Relative alignment of energy levels between the Fermi energy level (E_F) of the contact material and HOMO-LUMO energy levels of the organic semiconductor in p-channel and n-channel organic TFTs

and LUMO energy levels is reasonably large so that only one type of charge carrier can be efficiently injected and extracted. For example, the HOMO level for pentacene is approximately -5 eV [61–63] and approximately -5.44 eV for DNTT [72], when used with gold contacts, where gold has a normal work function of around 5.2 ± 0.1 eV, with possible variations due to exposure to air or other layers in the device [82] and makes this combination suitable for a p-channel transistor.

2.3.1 Current-voltage characteristics



Figure 2.8: Current-voltage characteristics of a p-channel organic transistor with different operation regimes (a) A sample transfer curve, i.e. the drain current I_D as a function of the gate-source voltage V_{GS} at a constant drain-source voltage V_{DS} . The gate current is shown in red. (b) A sample output curve, i.e. the drain current I_D as a function of the drain-source voltage V_{DS} for a set of constant gate-source voltage V_{GS} values

Since only p-channel TFTs will be investigated in this thesis, the transistor operation of a p-channel TFT will be discussed in detail in this section, but the same equations also apply to n-channel TFTs by reversing the polarity of the charge carriers. On applying the voltage V_{GS} between the gate electrode and the source contact, charge carriers are accumulated in the semiconductor due to charge coupling across the dielectric close to the semiconductor-dielectric interface. Since organic semiconductors are normally intrinsic and the polarity of the channel is decided according to the relative energy levels of the semiconductor and the contact material, the accumulation of charge carriers is the formation of the conduction channel of the transistor. This conducting channel is situated only in the first few monolayers of the organic semiconductor layer, close to the semiconductor-dielectric interface [83, 84], i.e. the thickness of the channel relative to the semiconductor layer thickness is extremely small. The charge-sheet model assumes the thickness of the channel is zero [85] and simplifies the problem so that the metalinsulator-semiconductor structure is modeled as a parallel-plate capacitor, with the gate electrode and the semiconductor layer acting as the two *plates*, separated by the dielectric insulator.

The voltage V_{GS} creates an electric field, which modulates the resistance at the interface between the organic semiconductor and the dielectric. Considering a resistance dR over a short distance dx in the channel :

$$dR = \frac{dx}{W\mu Q(x)} \tag{2.1}$$

where W is the channel width and Q(x) is the surface-charge density. For the organic TFT, the potential at the source contact is V_{GS} , and on application of the voltage V_{DS} , the potential at the drain contact is V_{GS} - V_{DS} . In general, the potential in the channel at any position x will be V_{GS} -V(x). Thus, the charge density in the channel $Q_s(x)$ can be expressed as a function of the capacitance and the potential in the channel as :

$$Q_{\rm s}(x) = C_{\rm diel}[V_{\rm GS} - V(x)]$$

In real devices, the semiconductor layer is often characterized by impurities, defects and grain boundaries which act as charge-trapping sites. In addition, a work-function difference and donor/acceptor states at the interface result in an offset voltage required to be overcome for the accumulation of mobile charge carriers in the channel. This is termed as the **threshold voltage V**_{th}. Thus, to describe the charge density in real devices, $Q_s(x)$ is expressed as:

$$Q_{\rm s}(x) = C_{\rm diel}[V_{\rm GS} - V_{\rm th} - V(x)]$$

$$(2.2)$$

Substituting dR = dV/dI, and Eqn.2.1 in Eqn.2.2, and integrating x over the channel length L:

$$I_D \int_0^L dx = \mu C_{\rm diel} W \int_0^{V_{\rm DS}} dV [V_{\rm GS} - V_{\rm th} - V(x)]$$

Under the given potentials at the source and drain contacts respectively as discussed above, and integrating the above equation, we obtain the expression for the drain current I_D in this operation regime:

$$I_{D,lin} = \mu C_{\rm diel} \frac{W}{L} \left(V_{\rm GS} - V_{\rm th} - \frac{V_{\rm DS}}{2} \right) V_{\rm DS}$$

for $|V_{\rm GS} - V_{\rm th}| > |V_{\rm DS}| > 0$ (2.3)

The following assumptions have also been made in deriving the above equation:

- 1. Gradual Channel Approximation : The variation of the electric field along the channel (which drives the drain current) is assumed to be much smaller in comparison to the variation of the electric field perpendicular to the channel from the gate to the semiconductor (which determines the accumulation of charge carriers). [85, 86].
- 2. The charge-carrier mobility μ is independent of the drain-source voltage V_{DS}.

The above derivation of the transistor's current-voltage characteristics is a simpler version allowed by the assumptions made [87]. A more complex derivation of transistor characteristics can be found in [88,89].

2.3.2 Operation Regimes

Linear Regime

Eqn.2.3 describes the drain current I_D for the condition: $|V_{GS} - V_{th}| >> |V_{DS}|$, in which the conducting channel is nearly uniform and can be modeled as a resistor as described above. Considering Eqn.2.3 and the assumed condition, the magnitude of $\frac{V_{DS}}{2}$ is negligible in comparison to the magnitude of $(V_{GS}-V_{th})$ and the equation can be simplified to assume a linear dependence of I_D on V_{DS} and operation under this condition is termed as the linear operation regime. For practical measurements, for larger values of V_{DS} at the end of the linear regime, I_D has a parabolic dependence on V_{DS} at a constant V_{GS} , as can be observed in Fig.2.8(b).

Saturation Regime

When V_{DS} is increased further in magnitude than in the linear regime, the slope $\frac{\partial I_D}{\partial V_{DS}}$ gradually decreases due to a smaller number of accumulated charges on the drain side, until I_D reaches a maximum value at the condition $V_{GS}-V_{th}=V_{DS}$. At this point, there are almost no accumulated charges at the drain-contact side, which is termed as the *pinch-off point*. With a further increase in V_{DS} , the pinch-off point moves slightly further away from the drain contact but still in close proximity as compared to the source contact, and there is no accumulation of charge carriers at the drain contact in this state. By substituting $V_{DS}=V_{GS}-V_{th}$ in Eqn.2.3, the expression for the drain current in the saturation regime can be obtained:

$$I_{\rm D,sat} = \mu C_{\rm diel} \frac{W}{2L} (V_{\rm GS} - V_{\rm th})^2$$

for $|V_{\rm DS}| > |V_{\rm GS} - V_{\rm th}| > 0$ (2.4)

This operation regime is called the saturation regime because the drain current is *satu*rated, i.e. it no longer depends on the drain-source voltage V_{DS} , as seen in Fig.2.8(b).

Off-state Regime

The equations for the drain current in the linear and the saturation regimes are only valid upon formation of a charge-sheet layer, i.e. only if $V_{GS} > V_{th}$, since V_{th} is the voltage required to overcome all the obstacles to accumulate charge carriers in the channel. When V_{GS} is much smaller than V_{th} , the transistor is in the *off-state* and the current in this operating condition is determined by the band gap or the doping condition. The current in this transistor operation regime, the *off-state current* I_{Off} should be as small as possible to reduce the power consumption of the transistor.

Subthreshold Regime

When the gate-source voltage V_{GS} is slightly smaller than the threshold voltage V_{th} , but larger than the switch-on voltage, the free carrier concentration in the channel is quite small. The switch-on voltage V_{so} is the gate-source voltage at which the drain current starts increasing beyond the off-state. The drain current in the subthreshold regime is due to charge carriers which are injected in the semiconductor layer by thermal excitation and have crossed the energy barrier at the source contact and have diffused to the drain contact. The drain current depends exponentially on V_{GS} in this regime, given by the following expression:

$$|I_{\rm D}| = I_{\rm o} exp\left(\frac{e|V_{\rm GS}|}{nk_BT}\right) \tag{2.5}$$



Figure 2.9: Extraction of the charge-carrier mobility and the threshold voltage from the slope and the intercept of experimentally measured transistor characteristics in (a) the linear operation regime and (b) the saturation operation regime

2.3.3 TFT Operation Parameters

Charge-carrier mobility

The charge-carrier mobility can be extracted by fitting the measured transfer curves according to Eqn.2.3 and Eqn.2.4, if the curves for $I_D(V_{GS})$ in the linear regime and $\sqrt{I_D}(V_{GS})$ in the saturation regime are straight lines. In that case, the mobility can be calculated from the slope of the curves as shown in Fig.2.9.

In case the extracted curves are not straight lines, it is better to calculate the differential charge-carrier mobility pointwise in the respective regimes according to the following equations:

$$\mu_{\rm lin}(V_{\rm GS}) = \frac{L}{W C_{\rm diel} V_{\rm DS}} \left(\frac{\partial I_D}{\partial V_{\rm GS}}\right)$$
(2.6)

$$\mu_{\rm sat}(V_{\rm GS}) = \frac{2L}{WC_{\rm diel}} \left(\frac{\partial\sqrt{I_D}}{\partial V_{\rm GS}}\right)^2 \tag{2.7}$$

The charge-carrier mobility μ is one of the most important parameters for quantifying the performance of organic TFTs. The mobility can depend on a wide range of parameters such as the degree of π orbital overlap between neighboring molecules, the chemical purity of the organic semiconductor, the degree of molecular order in the semiconductor layer and the density and energy distribution of trap states. The influence of some of

these factors as modulated at the dielectric-semiconductor interface are examined in detail in this thesis.

Threshold voltage

An ideal gate dielectric has a large energy band gap with no electronic states present either within the gate dielectric layer or at the interface with the semiconductor. However, in reality, the structural and chemical defects in the dielectric layer lead to the formation of trap states in the semiconductor layer as well as at the dielectric-semiconductor interface. Along with this, there is often a mismatch in energy levels between the semiconductor and the gate-electrode metal. A non-zero gate-source voltage, defined as the threshold voltage, must be applied to overcome these effects to generate charge carriers in the channel, i.e. to fill the trap states and shift energy levels to reduce charge injection barriers. More detailed accounts of the threshold voltage and its significance can be found in the following research articles [90–92]. There are several methods proposed to determine the threshold voltage for organic TFTs [93–96]. Commonly, the threshold voltage can be extracted from the transfer curves by fitting the Eqn.2.3 and Eqn.2.4, as shown in Fig.2.9. The threshold voltage will be an important parameter in discussing the function of self-assembled monolayers in hybrid dielectrics in Chapter 6.

Subthreshold swing

As mentioned in subsection 2.3.2, the subthreshold regime is characterized by a diffusiondominated current in the channel and the drain current depends exponentially on the gate-source voltage. The reciprocal of the slope of the transfer curve in this region, i.e. $\frac{\partial V_{\text{GS}}}{\partial \log I_{\text{D}}}$ is defined as the subthreshold swing.

$$S = \frac{\partial V_{\rm GS}}{\partial \log I_{\rm D}}$$

= $\ln 10 \frac{k_B T}{e} \left(1 + \frac{e N_{\rm it}}{C_{\rm diel}} \right)$ (2.8)

where N_{it} is the concentration of electrically active impurities at the gate dielectricsemiconductor interface. The subthreshold swing is an indication of the quality of the gate dielectric and the dielectric-semiconductor interface and is considered as an important parameter to qualify the dielectric in this thesis.

Current on-off ratio

The current on-off ratio is the ratio of the drain current in the on-state to the drain current in the off-state. The on-state current is the drain current at the maximum value of the gate-source voltage (in magnitude) and the off-state current is measured at the start point of the V_{GS} sweep. A large value of the current on-off ratio is desired for low power consumption and large noise margins of circuits with the organic TFTs.

Gate current

The gate current is a result of leakage through the gate dielectric, shown in red in Fig.2.8 and should be as small as possible. The gate current is quantified at the largest value of the gate-source voltage applied (in magnitude), as often the gate leakage current is higher with higher values of the gate-source voltage applied. The gate current will be used as an important parameter to determine the quality of the gate dielectric in this thesis.

2.4 Dielectric properties

A large part of this thesis focuses on the gate dielectric component of the organic TFT and the effect of dielectric properties on TFT performance. For this purpose, it is important to review some basic principles of dielectric materials and their properties. The gate dielectric material employed in Si-MOSFETs for the longest time was SiO₂ and the strategy for improving circuit performance was circuit element shrinkage till the fundamental limits of the material were reached, leading to a requirement for new materials. Particularly in the case of SiO₂, the need for high gate-dielectric capacitance for submicron devices and circuits required the aggressive scaling down of the dielectric thickness because of the relatively small dielectric constant of SiO₂ (k = 3.9). Unfortunately in ultrathin SiO₂ films, the tunneling current increases exponentially. It dominates the leakage current, heavily compromises the device performance and leads to a large power dissipation [97]. Similarly for organic TFTs, the strategy towards low operating voltages is to enhance the gate-dielectric capacitance by using high-*k* materials. This section will examine basic dielectric properties and their operating principles. A more detailed explanation about the requirements of dielectric materials for organic TFTs, the different materials used and strategies explored in this thesis are given in Chapter 3.

A dielectric or insulator is defined as a material with an absence of charge transport, i.e. a material with ideally a large band gap and in principle, no electronic states in the band gap for valence electrons to be excited into. An ideal insulator environment can only be found in a vacuum dielectric. When a reasonable amount of electric field is applied to non-ideal insulators, there is a shift in charge distribution which leads to field-induced polarization that characterizes dielectric behavior. The figure of merit for a dielectric material is the capacitance per unit area (C_{diel}), i.e. the ability to store electrical charge at a given voltage. On applying a voltage V to two electrodes separated by a distance d in vacuum, an electric field E is created :

$$E = \frac{V}{d}$$

The charge per unit area Q created is then proportional to the electric field:

$$Q = \varepsilon_0 E$$
$$= \frac{\varepsilon_0 V}{d}$$

where ε_0 is the absolute dielectric permittivity in vacuum. If a dielectric material with a relative permittivity (also know as the dielectric constant) of k is introduced between the electrodes, the capacitance can then be defined as :

$$C_{\text{diel}} = \frac{Q}{V}$$

$$= \frac{\varepsilon_0 k}{d}$$
(2.9)

(The relative permittivity k for vacuum is 1)

The dielectric properties of a material and its dielectric response to an electric field depend on the frequency of the electric field. The charge-density fluctuations developed as a response depend on the nature of the material and the dominant polarization mechanism operating according to the frequency of the applied electric field. The main types of polarization mechanisms are as follows.

• Electronic polarization occurs at frequencies up to 10^{15} Hz and occurs when

the applied electric field distorts the negative electronic cloud around the positive atomic nuclei in the direction opposite to the field. For crystals in which the atoms are well locked into the lattice structure, electronic polarization dominates the polarization mechanism.

- Ionic polarization, as the name suggests, typically occurs in ionic compounds and responds to field frequencies up to 10¹² Hz. Upon applying an electric field, the cations and anions are displaced in opposite directions, leading to charge separation and field-induced polarization.
- Orientational polarization arises when the material contains a permanent dipole moment, typically found in polymers and dipolar moieties in polymers as well as in complex oxides, which rearranges on application of an electric field. For most polymers, a combination of orientational polarization and chain relaxation dominate the dielectric behavior. The relaxation frequencies for orientational polarization range from 10⁸-10¹⁰ Hz.
- Interfacial polarization contributes to dielectric behavior at relatively lower frequencies (10⁶ Hz) and is usually observed in polycrystalline solids and hybrid systems. It occurs when there is charge accumulation at an interface in a heterogeneous system and affects both bound as well as free charges. Interfacial polarization is most relevant to the dielectric materials and material systems discussed in this thesis.
- Electrode or electric-double-layer (EDL) polarization also occurs at extremely low frequencies, lower than 10³ Hz. All electric or ionic conducting systems contain free charges and/or ions which move toward the electrode or the sample interface upon applying an electric field. This movement leads to the formation of electric double layers (EDLs) in these regions which constitute the thickness of the capacitor. This form of polarization is most relevant for electrolyte insulators, which are frequently used in organic transistors aimed towards bioelectronics applications.

The frequency response of areal capacitance depends on the relative permittivity (k) of the material, as shown in Fig.2.10 [98]. In high-quality low-k materials $(k \leq 3.9)$, the capacitance is stable over a large range of frequencies because the major contribution



Figure 2.10: Comparison of the frequency-dependent areal capacitance of high-k and low-k dielectric films. Adapted from [98]

comes from electronic polarization. However, for high-k materials (k > 3.9), the capacitance may decrease at higher frequencies because the polarization in high-k materials may be unable to follow the change of a high-frequency electric field. This is particularly important when choosing dielectric materials for high-frequency applications of organic transistors.

For inorganic dielectrics, such as metal oxides or nitrides, the band gap (E_g) is approximately inversely proportional to the relative permittivity (k), which plays a crucial role in choosing an appropriate material for the dielectric. For higher capacitances, i.e. higher charge-storing abilities at lower voltages, a larger dielectric constant k is desired, but this might mean choosing a dielectric material with a narrower band gap. Although a smaller band gap can be chosen to enhance k, a large band gap is desired to suppress charge injection from the electrodes and to reduce charge generation due to thermal and photoexcitation processes.

2.4.1 Capacitance Measurement

A schematic representation of the method used to measure the gate-dielectric capacitance of gate dielectrics in organic transistors employed in this thesis is shown in Fig.2.11(a). A small AC signal of known frequency and amplitude is applied to the capacitor. The displacement current (I_{disp}) is converted to an AC voltage using a low-noise amplifier,



Figure 2.11: (a)Schematic representation of the capacitance measurement setup (b) An example of a frequency-dependent capacitance measurement performed on an $AlO_x/phosphonic$ acid-SAM hybrid gate dielectric with an optical image of a capacitor (inset)

and the capacitance (C) can be calculated using the following equation:

$$I_{\rm disp} = C \frac{dV}{dt} \tag{2.10}$$

It is important to note here that the displacement current does not actually flow through the gate dielectric due to a potential difference such as in a resistive current, but instead is a result of a change in the potential difference and flows only between the gate dielectric and the power source, as indicated by Eqn.2.10. The measurement is performed multiple times by varying the frequency over a defined range suitable for the dielectric, and the calculated values of the capacitance are averaged over this range and normalized for the area of the capacitor. A sample measurement performed on a hybrid gate dielectric can be seen in Fig.2.11(b).

Ideally, when dielectric materials are used in electrical devices like transistors, they should be completely insulating against electrical conduction, because of the absence of electronic states in the band gap for valence electrons to be excited into. However, in reality, dielectric materials have localized states near the Fermi energy level (E_F) , with the electronic wave functions decaying exponentially over the scale of the localization length. In real devices, upon applying an electric field, these states contribute to a current flow through the dielectric material known as the leakage current, or gate current I_G in transistors. Conduction through the dielectric material occurs through certain mechanisms, briefly described below:

- Direct or quantum tunneling occurs through very thin dielectric layers where the thickness of the dielectric layer is comparable to the localization length and charge carriers tunnel directly across the dielectric. The rate of coherent tunneling decreases exponentially with an increase in the dielectric thickness, and this process is independent of the temperature. Under high electric fields, which exceed the barrier height, the tunneling rate increases and field emission is induced. This specific type of mechanism is referred to as Fowler-Nordheim tunneling and the rectangular barrier in direct tunneling is modified to a triangular shaped one. direct tunneling is usually applicable for transport through self-assembled monolayers, which are considered in gate dielectrics in this thesis.
- Indirect or diffusive tunneling occurs in the case of thick dielectric barriers and a high density of localized states, in which electrons tunnel through consecutive states in a series of discrete steps. This tunneling mechanism is also temperature independent and is predominant in thick dielectric layers at extremely low temperatures.
- Hopping is a thermally activated electron transfer process which is dominant at low electric fields and moderate temperatures. Similar to diffusive tunneling, electrons travel between one or more sites and the process follows a classic Arrhenius temperature dependence. The rate of conduction by hopping depends inversely on the thickness of the dielectric. For thicker dielectrics, the distance is too great for cohesive tunneling, but electrons can travel more efficiently by hopping between sites.
- The **Poole-Frenkel effect** accounts for charge-carrier traps in hopping electron transport and is attributed to the lowering of Coulombic trapping barriers within the molecule by the applied electric field. The Poole-Frenkel effect explains electrical conduction in organic semiconductors when trapped electrons contribute to the electric current. This effect usually occurs at high temperatures and intermediate electric fields.
- Schottky emission explains the electron transfer mechanism at interfaces. A Schottky barrier is formed due to misaligned energy levels of two materials at an interface and a partial charge transfer occurs from one layer to the other. The formation of a Schottky barrier leads to the generation of a depletion layer and

electrostatic charges in which electrons can be injected through the interface once they have sufficient thermal energy to surmount the potential height.

When the voltage applied across a dielectric is increased beyond a certain threshold, it imparts sufficient energy for the electrons to be excited into the band. This voltage is called the breakdown voltage and the capacitor loses its dielectric properties at this voltage. For certain dielectric materials, a sharp breakdown voltage cannot be observed and the traditional definition of the breakdown voltage is not a useful metric. In these cases, the breakdown voltage is determined when the current through the dielectric crosses a certain threshold, at which point the dielectric loses its insulating properties. The desired threshold for the current through the dielectric and the voltage at which this phenomenon occurs must be taken into account while choosing a material for certain device and circuit design applications.

2.5 Growth and morphology of organic semiconductor films

Similar to inorganic semiconductors, organic semiconductors in the form of aromatic hydrocarbons, can also be synthesized as molecular crystals. However, the intermolecular Van der Waals bonding in organic single crystals is much weaker and negatively impacts their periodic arrangement and size. As a result, the chemical purity and crystalline quality of even the best organic single crystals cannot be compared to those of inorganic semiconductors. Instead, organic semiconductors are best suited for the production of polycrystalline layers, either from solution or by vacuum-deposition methods. Particularly small-molecule organic semiconductors, such as unsubstituted acenes, are practically insoluble in organic solvents and are almost always deposited as thin layers with relative ease by thermal sublimation and de-sublimation in vacuum. Since only vacuum-deposition techniques for small-molecule organic semiconductor films are discussed in this thesis, only those will be considered for an in-depth analysis.

As discussed in section 2.2, charge transport is promoted by molecular orbital overlap between neighboring molecules, and therefore, the morphology of organic molecules in a polycrystalline film must be conducive to molecular orbital overlap and superior charge transport properties. The underlying surface for organic semiconductor deposition is of utmost importance in this respect, as depicted schematically in Fig.2.12.



Figure 2.12: Schematic representation of the orientation of organic semiconductor molecules on (a) an electrically conducting and (b) an electrically insulating substrate with the direction of π -orbital overlap in each case depicted in red

If we consider the organic semiconductor pentacene, the molecules lie flat on an electrically conductive substrate (e.g. metals) due to the strong interactions between the molecules and the substrate. However, on an electrically non-conductive substrate, the molecule-substrate interactions are relatively weaker and the longitudinal axis of the molecules is now perpendicular to the substrate surface. In this configuration, where the molecules are standing upright on the surface, the direction of π overlap is parallel to the substrate, and this configuration of molecules is most suitable for charge transport. Since the organic semiconductor is deposited on the gate dielectric surface, the dielectric-semiconductor interface and the surface properties of the dielectric play a vital role in the growth and morphology of the semiconductor film. The nucleation and growth processes of these films and thereby their thin-film morphology are crucial to the charge-transport properties of organic transistors. This section will examine the different growth mechanisms, influential factors for suitable film morphology and the relationship between the structural and electronic properties of organic semiconductor films.

2.5.1 Thermodynamic considerations of vapor nucleation and growth

Considering the nucleation of organic semiconductor molecules for film growth, the basic thermodynamic driving force is the difference between the chemical potential $(\Delta \mu)$ of molecules in the vapor phase (μ_v) and the chemical potential of molecules in the crystalline phase (μ_c) . For nucleation to occur without a heterogeneous substrate to catalyze crystal formation, the vapor phase must be supersaturated, i.e.: $\mu_v > \mu_c$.

The nucleation process is a competition between the thermodynamic driving force and the energy costs associated with the creation of new surfaces. The thermodynamic driving force is also influenced by volume effects and the enthalpic free energy lowering by beneficial intermolecular interactions. The free energy required to form a finite-sized crystal formed of j molecules can be expressed as:

$$\Delta G(j) = -j\Delta\mu + j^{2/3} \sum_{i} \gamma_i A_i \tag{2.11}$$

where γ_i is the surface energy of a surface with area A_i and $\Delta \mu = \Delta \mu_c - \Delta \mu_v$.

Since organic molecules are usually highly anisotropic, there are many different possible surfaces with different surface energies, and the above equation represents only the macroscopic relationship between the total free energy, crystal size and surface energies. Eqn.2.11 highlights the importance of the surface and the relevant interfacial energies which lead to differences in the free energy and thereby the different growth modes of the film. The barrier to nucleation ΔG^* can be obtained by:

$$\left(\frac{\partial \Delta G(j)}{\partial j}\right)_{\rm T,P} = 0$$

where ΔG^* is $\Delta G(j) = \Delta G(i)$ and i is the critical cluster size, i.e. the size of the cluster at which the addition of one more molecule stabilizes the cluster. Therefore, ΔG^* is the barrier to nucleation at which the surface energy effects are the greatest. At this point, the addition of more molecules to the cluster will increase the intermolecular enthalpic interactions and lower the total energy. In other words, the intermolecular effects will dominate the surface effects to create a stable island. The importance of ΔG^* can be expressed by considering its strong influence on the density of stable nuclei which can be expected to survive long enough to contribute to crystal growth by the following association:

$$N^* = n_{\rm s} exp\left(\frac{-\Delta G^*}{k_{\rm B} T_{\rm s}}\right) \tag{2.12}$$

Even though thermodynamic models are valid for treating nucleation and growth mechanisms, several kinetic and scaling models adopted from classic nucleation theories developed for inorganic materials have also been applied to model the growth of organic semiconductors like pentacene.

2.5.2 Kinetics of nucleation and growth

Thermodynamic equilibrium is defined by the condition of detailed balance, in which all participating processes are proceeding at equal rates in opposite directions. In such a condition, there is no net growth. Although the system can be described by unchanging macroscopic variables such as the pressure or temperature, microscopically the system is continuously changing via processes which balance each other out.



Atomistic processes

Figure 2.13: The various atomistic processes contributing to the adsorption and crystal growth process during vapor deposition

Fig.2.13 shows the different atomistic processes contributing to the adsorption and crystal growth processes during vapor deposition. Atoms arrive at the surface with a rate of arrival R, as determined by the evaporation source and the experimental conditions. Considering a substrate with N_0 sites per unit area, and a number density of $n_1(t)$ single atoms arriving on the surface, the single-atom concentration is n_1/N_0 . Upon arrival at the surface, the single atoms diffuse over the surface until they are captured by one of the several processes, such as re-evaporation, nucleation of clusters, capture by existing clusters, dissolution into the substrate or capture at special sites which include defects. The probability of each of these processes depends on the single-atom concentration and the coverage, and in cases of thermally-activated processes, on the activation energies (e.g. re-evaporation energy $E_{\rm a}$, diffusion energy $E_{\rm d}$). The formation of clusters of critical size and their growth thereafter to larger clusters depends on these individual processes. Considering the different atomistic processes, it is clear that the two independent experimental variables, viz. the rate of condensation R and the temperature T are extremely important towards influencing the nucleation and growth of atomic clusters on the surface. In addition, the presence of defects on the substrate surface in forms of ledges,

kinks, dislocation and point defects is also very significant and these defects have a strong influence on the binding of single atoms and formation of small clusters on the surface.

Crystal growth is a non-equilibrium kinetic process, and the kinetics of the growth process must be considered. The various reaction processes indicated above and their relative influence determine the final macroscopic state kinetically, although it may not be the most stable state. The rate of nucleation is therefore dependent on various factors such as the rate of deposition, the substrate temperature, surface properties of the substrate, intermolecular interactions, and molecule-surface interactions, which will be explored in detail throughout this thesis.

The rate of nucleation is the number of nuclei of a critical size forming on a substrate per unit area and per unit time and depends on the rate at which the adsorbed atoms attach to it, given by the expression:

$$N = N^* A^* \omega \tag{2.13}$$

where N^* is the equilibrium concentration of stable nuclei, A^* is the critical nucleus area and ω is the rate of atom impingement onto critical nuclei.

Considering the individual dependencies of ω on the re-evaporation energy $E_{\rm a}$ and the diffusion energy $E_{\rm d}$, as well as the dependency of N^* on ΔG^* according to Eqn.2.12 and the critical area, the nucleation rate can be expressed as:

$$N = 2\pi r^* a_0 sin\theta \frac{PN_{\rm A}}{(2\pi MRT)^{1/2}} n_{\rm s} exp\Big(\frac{E_{\rm a} - E_{\rm d} - \Delta G^*}{k_{\rm B}T_{\rm s}}\Big)$$
(2.14)

where r^* is the critical nucleus radius, a_0 is the lattice parameter, θ is the contact angle, n_s is the total nucleation site density, E_a is the re-evaporation energy barrier, and E_d is the surface diffusion energy.

Evidently, three important energetic barriers directly influence the nucleation density. A larger degree of surface mobility and thereby a greater chance for molecules to encounter each other and form a stable cluster are achieved by a smaller barrier for diffusion. In general, the surface chemistry and roughness determine the lateral corrugation in the molecule-substrate interaction potential which influences the barrier to diffusion. A

larger barrier for desorption or re-evaporation means that the molecule is more tightly bound to the surface and has more time to form n-mers with other molecules to form a stable cluster. Each of these molecular processes is also a function of the moleculesubstrate interaction strength, and the implications of these factors will be considered in section 2.5.4.

2.5.3 Growth modes



Figure 2.14: The three most common growth modes during thin-film growth: (a) 2D layer-by-layer (Frank-van der Merwe) growth, (b) 2D monolayer followed by 3D growth (Stranski–Krastanow) and (c) 3D island growth (Volmer-Weber)

As previously mentioned, the growth mode of the organic semiconductor films must be conducive to charge transport by increasing the degree of π orbital overlap between neighboring molecules. The growth mode of films can be, in general, differentiated into 2D-like layer-by-layer growth, 3D island growth or a combination of island-layer growth, as shown in Fig.2.14. Layer growth (also called Frank-van der Merwe type of growth) occurs when the molecules are more strongly bound to the substrate than to each other and the first molecules to condense from the vapor phase form a complete monolayer on the surface. During island growth (also known as Volmer-Weber type of growth), small clusters are nucleated directly on the substrate surface and these clusters grow into islands of the condensed phase. This growth mode occurs when the deposited molecules are more strongly bound to each other than to the substrate. For organic transistors with large charge-carrier mobilities, a 2D-layer growth mode is preferred for organic semiconductor films since a 3D-island growth mode can give rise to voids and a larger number of grain boundaries in the film. Usually, organic semiconductor molecules form films with an intermediate growth mode, the island-layer (or Stranski-Krastanov) type of growth, in which layer growth after the formation of the first monolayer (or a few monolayers) is unfavorable. Layer growth is maintained if the binding of each subsequent layer to the bulk of the crystal decreases monotonically.

A mismatch in lattice parameters or certain molecular orientations may result in a high free energy at the intermediate layer interface and disturb this monotonic decrease in binding energy in consecutive layers, causing subsequently deposited molecules to form islands on the top of the intermediate layer. The growth mode is determined by competing interlayer interaction energies and molecule-substrate energies. The stronger the molecule-substrate interactions, the greater the tendency for 2D growth. It must be noted that each type of semiconductor and the type of surface have unique interactions, and any generalizations made about the hydrophobicity, surface energy or the surface chemistry may not be entirely accurate.

2.5.4 Factors influencing growth and morphology of organic films

Based on the theoretical considerations for the optimum growth of organic semiconductor films, this section details the experimental factors and their influence during fabrication.

• Rate of deposition

The rate of deposition is an important parameter to control during thermal sublimation in vacuum. Generally, an increase in the deposition rate increases the rate of nucleation in the film growth process, leading to smaller crystallites, smaller grains and thereby a larger density of grain boundaries in the polycrystalline film. A drastic increase in the rate of deposition greatly increases the tendency for supersaturation, and thermodynamic considerations for the nucleation in films may no longer be applicable. At extremely high deposition rates, film growth is completely kinetically dominated and the films fabricated may become completely amorphous with a glass-like nature. Therefore, in general, a relatively small deposition rate is preferred for organic semiconductor film growth to reduce the nucleation density and increase the overall grain size.

• Substrate temperature

The overall barrier to heterogeneous nucleation decreases with an increase in the substrate temperature. Although the free energy G is only weakly dependent on the temperature, all individual energy barriers (e.g. for desorption, diffusion) follow an Arrhenius temperature dependence. An increase in the substrate temperature motivates surface mobility and molecules to diffuse on the substrate to find

low-energy sites and promotes the formation of larger, circular grains. However, when exceeded beyond an optimum limit, an increased substrate temperature can lead to increased desorption or re-evaporation of molecules from the surface. In an extreme case, when the substrate temperature exceeds the sublimation temperature, nucleation is thermodynamically impossible. Therefore, the ideal substrate temperature is balanced between the temperatures required for increasing surface diffusion and for decreasing desorption. Within this optimum range for the substrate temperature, in which the sticking coefficient for molecules is sufficiently high for nucleation, an increase in the substrate temperature decreases the chemical potential difference ($\Delta \mu = \mu_c - \mu_v$), and an overall tendency for the preferred 2D-layer growth mode is higher. Often new organic semiconductor molecules synthesized require the investigation of an optimum substrate temperature on different substrates. An example of such an investigation can be seen in Appendix C.

• Substrate surface properties

The effect of surface properties of the gate dielectric in the organic transistor is of great importance for the fabrication of high quality semiconductor films, particularly, the effect of the surface energy and the surface roughness of the gate dielectric surface. The surface energy characterizes the wettability of the surface which has a possible influence on the probability and size of the formation of nucleating clusters from the deposited molecules. The surface roughness of the dielectric can be instrumental in suitably increasing the coordination between the deposited molecules and the surface. The rough areas on the surface decrease the barrier for nucleation and increase the barrier to desorption. However, on extremely rough surfaces, the grain quality of the films is negatively affected, leading to small, mis-oriented grains and poor device performances. The surface energy and surface roughness are considered in detail, particularly in Chapter 6 to establish a relationship between the interfacial properties at the dielectricsemiconductor interface and the electrical properties of the organic transistors.

2.5.5 Thickness of organic semiconductor films

The thickness of the organic semiconductor film has potentially a strong influence on their growth and morphology and on the device performance. Depending on the device configuration and the relative position of the contacts to the organic semiconductor, the thickness of the semiconductor layer influences the injection and extraction of charge carriers. A majority of the charge transport in organic transistors takes place in the first one or two monolayers of the organic semiconductor layer, close to the dielectric-semiconductor interface. Typically, organic semiconductor films are deposited with a thickness corresponding to multiple (10-20) monolayers to stabilize the initial charge-transporting monolayers. The impact of the film thickness on its stability and morphology is discussed in detail in Chapter 7.

2.6 Thin-film characterization techniques

Apart from the electrical characterization of organic transistors, the semiconductor and dielectric layers were also characterized for various surface and film properties. This section briefly describes the basic working principle of the various characterization techniques used in this thesis.

2.6.1 Contact angle goniometry

The surface energy was measured by static contact angle goniometry. In this method, a liquid drop is placed on the surface of interest, and the static contact angle is measured using an optical microscope. By measuring the wettability of two different liquids, a polar and a non-polar liquid, the surface energy is calculated using the Owens-Wendt method [99].

2.6.2 Atomic force microscopy

Atomic force microscopy (AFM) is a type of a scanning probe microscopy technique used for topographic imaging with a resolution of the order of a fraction of a nanometer. A schematic of the basic operational setup can be seen in Fig.2.15. A cantilever beam with a sharp tip oscillates at a resonance frequency, and a laser beam is reflected off the top of the cantilever to a split photodetector. The sharp tip probes the sample surface and the forces between them change the amplitude of oscillation, with the deflection in the cantilever corresponding to the topography of the surface. In this work, AFM has



Figure 2.15: Schematic representation of (a) an Atomic Force Microscopy (AFM) measurement and (b) the R_{RMS} surface roughness parameter, adopted from www.olympusims.com

been used to collect morphology information of organic semiconductor films as well as observe any changes in the morphology with time by using repeated scans of the same area. In addition, surface roughness information from the dielectric and semiconductor surfaces has been collected using AFM.

$\mathbf{R}_{\mathrm{RMS}}$ surface roughness

The parameter used to characterize the roughness of the surface from the topological information collected by the AFM is the Root-Mean-Square Roughness (R_{RMS}), as shown in Fig.2.15(b). The root-mean-square surface roughness is calculated as follows. All topographical features collected by the AFM are converted to profiles by considering the absolute values of all amplitudes or the profile height from the mean plane (\mathbf{Z}). These profile heights are squared, which accentuates all prominent features on the sample surface. The squared values are integrated over the entire surface and averaged to obtain a mean value. Finally, a square root value is obtained to compensate for the initial squaring of profile height values, which determines the R_{RMS} surface roughness. By selecting R_{RMS} as the parameter of choice to characterize the surface roughness of films, any contribution from features and surface elements contributing to rough surfaces is appropriately measured.



Figure 2.16: Schematic representations of (a) Scanning Electron Microscopy (SEM) and (b) Transmission Electron Microscopy (TEM) techniques

2.6.3 Scanning Electron Microscopy

Scanning Electron Microscopy (SEM) utilizes a fine beam of focused electrons to scan a sample surface, and the interaction between the electrons and the sample surface is recorded to create a highly magnified image (with a resolution of 1-20 nm). A basic schematic of the inside of an SEM with its different components can be seen in Fig.2.16(a). The electron source emits a beam of high-energy electrons which is focused using electromagnetic lenses and is incident on the sample surface. Upon interaction with the sample surface, secondary electrons, backscattered electrons and X-rays are emitted, collected by various detectors and can be used to access information about the surface topography and elemental composition. The contrast in SEM images is created due to a difference in atomic weights of different elements on the sample surface, creating a stark contrast between the organic semiconductor and the dielectric films. SEM images have been used to analyze the growth of organic semiconductors on dielectric surfaces in this thesis.

2.6.4 Transmission Electron Microscopy

Transmission Electron Microscopy (TEM) uses an electron beam, as the name suggests, transmitted through an atomically-thin sample to create an image of the internal structure to collect information about the morphology, composition and crystal structure. A basic schematic of the functioning principle of a TEM with its different components can be seen in Fig.2.16(b). Samples investigated in the TEM must be extremely thin (≤ 150 nm) to allow electrons to pass through them to be collected at a detector and analyzed to create a 2D image, often with a resolution below 1 nm. In this thesis, TEM has been used to measure the thickness of ultrathin gate oxide films.

2.6.5 X-ray Diffraction



Figure 2.17: Schematic representation of a grazing incidence X-ray diffraction measurement with an incident X-ray beam (k_i) at an incident angle α_i , and the diffracted beam (k_f) at a diffraction angle of 2θ , used to measure the interlayer spacing d of the organic semiconductor layer

Grazing incidence X-ray diffraction is a surface sensitive technique used to analyze the crystallinity of the sample. An X-ray beam probes the sample at an extremely small angle of incidence with respect to the sample plane to limit the penetration of the beam to the surface layers. The angles and intensities of the diffracted X-ray beams give information about the position of the molecules and thereby the crystalline nature of the layer. In this thesis, grazing incidence X-ray diffraction is used to analyze the d-spacing between the semiconductor layers, the orientation of the molecules with respect to the surface and the quality of the layer-by-layer packing in organic semiconductor layers.

Chapter 3

Hybrid gate dielectrics in organic thin-film transistors

A majority of the efforts in the development of organic thin-film transistors (TFTs) have been focused on improving the electrical performance in the conducting channel of the transistor. One of the ways this was achieved was by synthesizing organic semiconductors of higher quality, i.e. with higher intrinsic mobilities and better air stability [100–104]. Another important area of focus was the development of deposition methods of organic-semiconductor films with a sufficiently low density of structural and electronic defects [105, 106]. However, in addition to the semiconductor, the gate dielectric also plays a vital role in the organic transistor, with a primary function of isolating the organic semiconductor from the gate electrode, and has several requirements for desirable organic TFT operation.

As detailed in Chapter 2, the gate dielectric is polarized upon applying a voltage across it, and the internal charge separation and electric field buildup lead to the accumulation of charge carriers in the organic semiconductor layer. The principle function of the gate dielectric is thus, to effectively couple the charge applied to the gate electrode to form the conduction channel in the transistor. The thickness and the relative permittivity of the gate-dielectric layer determine the gate-dielectric capacitance (given by Eqn.2.9), which in turn determines the operating voltage of the transistor. In order to enable several portable and lightweight device applications, it is necessary to operate organic TFTs at a relatively low voltage, which can be achieved by a high gate-dielectric capacitance. The primary ways to obtain a high gate-dielectric capacitance are to either reduce the thickness of the gate-dielectric film or to use gate-dielectric materials with a large relative permittivity. Several potential device applications based on organic transistors are designed to be powered by small batteries or solar cells, which require an overall low power consumption along with low-voltage operation. It is important that the gate dielectric is able to withstand the electric field generated on applying the gate-source voltage, without incurring a significant leakage current density to maintain an overall low power consumption.

Considering organic TFTs in the inverted-device architecture (as shown in Fig.2.4), the gate dielectric provides a surface on which the organic semiconductor is deposited. The conducting channel of the transistor is confined to the first few monolayers of the organic semiconductor layer, located in close proximity to the dielectric-semiconductor interface [83, 84, 107]. Surface properties of the gate dielectric are a potential factor influencing the growth and morphology of the organic semiconductor layer and thereby the charge-transport characteristics of the organic TFT. As an example, the surface roughness of the gate dielectric can be correlated with the undesirable formation of nanoscopic interfacial grain boundaries and sites for charge scattering [108]. As another example, a gate-dielectric surface with an increased surface energy may facilitate semiconductor deposition by certain solution-processed techniques by providing a favorable hydrophilic surface for that particular technique [109]. On the other hand, chemical functionalities contributing to high surface energy may act as charge-trapping sites, negatively impacting the charge transport in the organic TFT [110]. The surface properties of the gate dielectric are thus a crucial aspect in correlating the material properties of the organic semiconductor to the electrical properties of the organic transistor, and will be a key consideration in this thesis to analyze different aspects of TFT performance based on their material and design configuration.

The gate dielectric also has an influence, either directly or indirectly, on several transistor operation characteristics including but not limited to the charge-carrier mobility, the threshold voltage and the threshold-voltage shift. For example, molecules with fluorine or amino groups are used in self-assembled monolayers in the gate dielectric to shift the threshold voltage to more positive or more negative values respectively [111]. The design and choice of the gate dielectric thus becomes crucial towards operation of the transistor and circuit design aimed for specific applications.

The long-term stability of organic TFTs remains an ongoing challenge and a key concern towards the development and integration of organic TFTs in commercial electronic
applications. Similar to other components in the organic TFT, the gate dielectric must also be compatible with the entire fabrication process and be stable against electrical or environmental degradation. Hence, along with the electrical properties of the dielectric, the mechanical and chemical properties also play a role in the choice of the gate dielectric for organic TFTs.

The most popular and common choice for the gate dielectric in organic TFTs is thermally grown silicon dioxide (SiO_2) on silicon wafer substrates. This choice is particularly useful for testing new organic semiconductors because of the uniformity of the gate dielectric and the simplicity of the TFT-fabrication process in which no additional steps are required to deposit and pattern the gate dielectric. However, the fact that highquality silicon dioxide can be produced only on single-crystalline silicon wafers, unsuitable for flexible TFT applications, as well as the low dielectric constant of SiO_2 (~3.9) and the inability to pattern the dielectric prevent the fabrication of high-performance, advanced devices and circuits. Apart from thermally grown SiO_2 , other gate insulators employed in organic TFTs can be broadly classified into the following categories based on their material classes: inorganic metal oxides, insulating polymers and electrolyte insulators. Metal oxides such as Al_2O_3 [112,113], TiO_2 [114,115], ZrO_2 [116,117], HfO_2 [118–120], Ta_2O_5 [121] and even perovskites such as $Ba_xSr_yTiO_3$ [122, 123] have been widely investigated. Deposition techniques such as sputtering, atomic layer deposition, pulsed laser deposition, anodization, electron-beam evaporation and even several solution-processing techniques such as spin/spray/bar-coating are popularly employed, with the method of deposition chosen to suit the other materials and techniques used in the TFT fabrication process. Vapor deposition techniques are extremely successful in producing high-quality, defect-free films, while solution-processing techniques offer advantages such as large-area fabrication and low processing temperatures suitable for alternative substrates. Polymers investigated most frequently as gate dielectrics for organic TFTs include poly(vinyl alcohol) [124], polyimide, poly(vinylphenol) [125, 126], poly(methyl methacrylate) [127, 128], photoresists, etc. There are two main approaches for fabricating polymer dielectrics, either by depositing the polymer solution followed by evaporating the solvent, or formation of the polymer from the monomer directly on the gate electrode. Polymer dielectrics are most commonly deposited by solution-processing techniques such as spin or spray coating, inkjet printing, etc. Although ideal for a lot of TFT configurations and applications, polymer dielectrics often suffer from a lack of thermal and chemical stability for patterning the dielectric and therefore hinder the possibility of fabricating complex devices and circuits. Polymer dielectrics are most suited for applications in which a solution-processing approach is adopted and benefit from advantages like roll-to-roll and scalable fabrication.

An emerging class of gate dielectrics for organic TFTs is electrolyte insulators, particularly developed towards bioelectronics applications [129, 130]. The primary types of electrolytes employed as gate dielectrics include polymer electrolytes, polyelectrolytes, ionic liquids and ion-gels [131–133]. Electrolyte insulators function by the formation of electric double layers (EDLs), i.e. the displacement of oppositely charged ions at the electrolyte/electrode interfaces in response to an electric field. Since the capacitance of the electrolyte layer depends on the formation of the EDLs and not the physical thickness of the electrolyte layer, very large capacitances (1-10 μ F/cm²) can be achieved. Consequently, a large charge-carrier density can be generated at the interface, which is the primary motivation [134]. However, it must be noted that the capacitance of the electrolyte insulator is frequency-dependent and often charge-carrier mobilities reported are largely overestimated [135]. Besides, thick films of the electrolytes (1-10 μ m) are required to prevent undesired leakage currents, which remains a major issue.

Considering the advantages of each of the above mentioned dielectric material classes, a unique approach is to employ "hybrid dielectrics", which comprise of more than one type of dielectric material. For example, blends of low-k polymers with high-k fillers such as perovskites [barium titanate (BT), barium zirconate (BZ), aluminum titanate (AT), barium strontium titanate (BST)] and metal oxides [ZrO₂, HfO₂, Ta₂O₅, Y₂O₃, TiO₂] are an effective method to increase the average relative permittivity of the composite [136, 137].

This thesis explores the use of hybrid gate dielectrics with two components, i.e. an ultrathin metal-oxide film in combination with an organic self-assembled monolayer. An important aspect regarding hybrid dielectrics is the concept of self-assembled monolayers (SAMs), which will be further elaborated on in Section 3.1. The materials, fabrication techniques and the choices thereof have been detailed in Section 3.2, along with the multi-fold advantages and benefits of this approach.

3.1 Self-assembled monolayers

George Whitesides and Bartosz Grzybowski defined self-assembly as "the autonomous organization of components into patterns and structures without human intervention" [138]. A self-assembled monolayer (SAM) is a molecular structure formed spontaneously on the surface by adsorption and organization into more or less ordered domains, with its thickness chemically limited to one monolayer. Molecules which are used to form SAMs are usually composed of three main components (as shown in Fig.3.1): a head group, which binds to the underlying surface, a backbone made up of either an aliphatic chain or aromatic oligomers and responsible for molecular ordering, and a terminal or tail group which determines the topography and surface properties of the SAM. The process of molecules assembling into a SAM proceeds usually through the adsorption of the head group on the substrate, followed by the slow organization of backbone and tail groups. The individual components of the molecules have a strong influence on determining the properties of the SAM. For example, a strong chemisorption of the head group brings about a high stability of the SAM, and the interactions between the aliphatic chain backbone determine the molecular ordering and packing density of the SAM.



Figure 3.1: Schematic representation of molecules assembling into self-assembled monolayers (SAMs) with the different components of the molecules contributing to self assembly

In this section, we will focus on SAMs used for gate dielectrics in organic TFTs. The most popular molecules used to form SAMs and particularly the molecules discussed in detail in this thesis are shown in Fig.3.2, denoted by abbreviations either commonly used in literature or as abbreviated in this thesis (complete list in the Abbreviations section).



Figure 3.2: Commonly used molecules for self-assembled monolayers (SAMs) in gate dielectrics in organic TFTs

The potential application of SAMs for gate dielectrics in organic TFTs was first developed by the Vuillaume group by fabricating metal-insulator-semiconductor (MIS) structures with an octadecyltrichlorosilane (ODTS) SAM assembled on the native SiO_2 surface as well as alkyltrichlorosilanes with three different chain lengths (12, 16 and 18 carbon atoms) as the insulator [139, 140]. It was demonstrated that the leakage current density $(10^{-8}-10^{-11} \text{ A/cm}^2)$ through the MIS structure was greatly reduced, by about 4-5 orders of magnitude in comparison to an oxide or nitride insulator of similar thickness. The beneficial effect of the SAM was ascribed to an increase in the energy barrier and the packing density of the SAM. This approach was further extended to fabricate thin-film transistors with alkyltrichlorosilane SAMs with different functional terminal groups assembled on native SiO_2 as the gate insulator [141, 142], exhibiting very low leakage current densities $(10^{-5}-10^{-8} \text{ A/cm}^2)$. This seminal work established the advantage of using SAMs to achieve low operating voltages in organic TFTs by reducing the thickness of the gate dielectric without incurring significant leakage current losses. This encouraged the wide investigation of alkyltrichlorosilane SAMs for gate dielectrics towards fabrication of low-voltage organic TFTs, exploring the different aspects of the SAM and its impact on the performance of the organic TFT. Halik et. al investigated a series of oligothiophene-organic semiconductor transistors with an ODTS dielectric by varying the number of thiophene units and alkyl side-chain substitutions on the semiconductor [143]. A phenoxy-terminated SAM, PhO-ODTS was used as the gate dielectric to fabricate pentacene TFTs, with the strong $\pi - \pi$ interactions of the terminal phenyl groups preventing the penetration of the pentacene molecules in the

SAM and defining a better semiconductor-dielectric interface [144]. The use of SAMs with solution-processed organic semiconductors was also demonstrated by Park et. al by fabricating a regioregular poly(3-hexylthiophene) (P3HT) organic TFT with a docosyltrichlorosilane SAM as the gate dielectric demonstrating a significant reduction of the operating voltage and power consumption [145].

Apart from alkyltrichlorosilane SAMs, another major category of molecules for SAMs as gate dielectrics are phosphonic acids, which were introduced by Klauk et. al by fabricating both p-channel and n-channel organic TFTs using the organic semiconductors pentacene and hexadecafluorocopperphthalocyanine ($F_{16}CuPc$), as well as low-power complementary circuits [35]. The gate dielectric in use was an octadecylphosphonic acid (ODPA) SAM with a thickness of 2.1 nm in combination with a plasma-grown ultrathin aluminum oxide (AlO_x), which was determined to be 3.6 nm thick. Despite the overall small thickness of the dielectric, the leakage current density was reduced by 3 orders of magnitude in comparison to a standalone oxide dielectric without the SAM. In addition, the ODPA SAM demonstrated a lower leakage current density compared to the ODTS SAM which was assigned to its better packing density, as measured by X-ray photoelectron spectroscopy measurements.

Molecules are assembled into SAMs on the surface of interest most commonly by immersion of the substrate into a solution of the molecules with a suitable solvent. Recently, formation of SAMs by vacuum-evaporation techniques has also been reported [146,147]. The adsorption of alkylphosphonic acids on aluminum oxide surfaces proceeds through the interaction of the phosphonic acid head group with the hydroxyl groups (-OH) present on the aluminum oxide surface. A mono-, bi- or tri-dentate adsorption mechanism anchors the phosphonic acid group via one, two or three P-O-Al bonds via a stepwise condensation reaction as shown in Fig.3.3 [148–150].



Figure 3.3: Assembly of phosphonic acids on aluminum oxide via mono, bi or tridentate bonding mechanisms through condensation reactions

Several factors influence the adsorption of the phosphonic acid molecules and formation

of the SAM which include the presence of hydroxyl groups, surface roughness of the substrate, temperature of the substrate, the chain length and chain type of the molecule, the type, concentration and incubation time of the solvent and suitable preparation of the substrate surface. Of the various influencing factors, the influence of the backbone chain length of alkyl- and fluoroalkyl phosphonic acid molecules will be discussed in detail in Chapter 6.

SAMs employed as gate dielectrics influence the functionality and performance of organic TFTs in several ways. Primarily, as described above, SAMs enable low-voltage operation of organic TFTs by reducing the gate-dielectric thickness while maintaining a significantly low leakage current, which would otherwise require oxide dielectrics of a large thickness. In organic TFTs with hybrid gate dielectrics, SAMs introduce a buffer layer between the oxide and the organic semiconductor, thereby suppressing the formation of Fröhlich polarons and charge trapping at the interface [151]. This phenomenon will be demonstrated and discussed in detail in Chapter 4. A major contribution of SAMs lies in modifying the surface reactivity, preventing water adsorption, passivating the surface from undesired reactions and thereby defining the surface energy of the dielectric. This has been instrumental in improving the bias-stress stability of organic TFTs [152, 153]. A predominant influence of the SAM in modification of the dielectric surface properties is linked with the deposition of the organic semiconductor on the top of the SAM in bottom-gate TFT architectures. The choice and quality of the SAM has potentially a strong impact on the nucleation, growth and morphology of the organic semiconductor layer, and thereby the charge-transport properties of the organic TFT. This will be further elaborated in Chapter 6. A number of organic-TFT characteristics can be suitably tailored by customizing the type of the SAM. For example, a well-ordered and densely packed SAM promotes an optimum organic-semiconductor morphology and a large charge-carrier mobility [154]. By nature of their dipole character and suitable terminal groups, SAMs can be instrumental in altering the threshold voltage of organic TFTs significantly and purposefully, according to operation requirements [155, 156]. An example of this phenomenon will be discussed in detail in Chapter 6 with fluorinesubstituted phosphonic acid molecules used to form the SAM. Furthermore, a mixture of alkyl and fluoroalkyl phosphonic acid molecules can be used to tune the threshold voltage of organic TFTs suitably for optimum circuit design of complementary inverters [157, 158].

3.2 Hybrid gate dielectrics for low-voltage organic TFTs

High-k metal oxides were sought as an alternative to SiO_2 to overcome the constraint related to its low dielectric constant. However, most metal-oxide films, including SiO_2 , have certain unfavorable properties which can have adverse effects on device performance. These properties include the presence of surface hydroxyl groups (-OH) which act as charge-carrier traps, ionic polarization between the charge carriers in the semiconductor and the metal-oxide lattice and a relatively high surface roughness. Therefore, the concept of surface modification of metal-oxide films by an additional dielectric layer was introduced as a method to counter these adverse effects. Several reports in literature refer to organic layers such as polymers or self-assembled monolayers (SAMs) employed for this purpose as *surface-modification layers* for metal-oxide gate dielectrics. Since the thickness of the polymer layers or SAMs is comparable to the oxide thickness, the combination of inorganic/organic bilayer dielectrics will be referred to as hybrid gate dielectrics in this thesis.

A wide variety of polymers were used in combination with metal oxides to form hybrid gate dielectrics to improve the performance of organic TFTs in terms of increased charge-carrier mobilities, increased on-off current ratios and decreased leakage currents. Some notable examples include PMMA/Ta₂O₅ [159], PMMA/Al₂O₃ [160], PVA/SiO₂ [161], crosslinked PVP/Ta₂O₅ [162], crosslinked PVP/TiO₂ [163], etc. A more in-depth review of polymer/metal oxide hybrid gate dielectrics can be found in [164]. Although the addition of the polymer layers improved the device performance in most cases and increased the charge-carrier mobilities by the action of the polymer (for example, by reducing the surface roughness), the polymer layers deposited required relatively large thicknesses to form a uniform, conformable layer. This resulted in large operating voltages as compared to single-layer oxide dielectrics, which was a considerable disadvantage in spite of the advantageous larger charge-carrier mobilities.

Self-assembled monolayers, by virtue of their monolayer thickness, are a suitable substitute to polymers as surface modification or the second component of hybrid gate dielectrics in addition to metal oxides. Hybrid gate dielectrics composed of a metal oxide and a self-assembled monolayer are instrumental in simultaneously achieving a low operating voltage and a low leakage current density, which is difficult to attain through the manipulation of single-material gate dielectrics. The comparison of thick metal oxide, thin metal oxide and hybrid gate dielectrics is shown schematically in Fig.3.4. In the case of single-material gate dielectrics, such as metal oxides, a small thickness can successfully achieve a low operating voltage but might lead to large leakage currents and vice-versa, a large thickness suppresses large leakage currents but increases the operating voltage of the transistor. The advantage of the hybrid dielectric is then apparent, where a thin metal oxide layer is used in combination with the organic SAM, providing multi-fold benefits of a low operating voltage, a reduced gate-leakage current as well as the possibility of surface property modification by the SAM.



Figure 3.4: Schematic representation and comparison of thick and thin oxide dielectrics with hybrid gate dielectrics composed of a thin oxide layer and a self-assembled monolayer

As previously mentioned in Section 3.1, the two primary categories of molecules used to form SAMs for gate dielectrics in organic TFTs are silane-based molecules and phosphonic acid-based molecules. Organosilane molecules form SAMs on surfaces bearing -OH groups, e.g. silica, aluminum oxide, zinc oxide, etc. The most popular and widely studied molecules include ODTS (octadecyltrichlorosilane), OTS (octadecylsilane), OTMS (octadecyltrimethoxysilane), HMDS (hexamethyldisilazane), and a wide variety of techniques have been used to form SAMs such as dipping [165,166], immersion [167], stamp printing [168], Langmuir-Blodgett technique [169], spin-casting [170], and even vapor deposition [171,172]. Silane-based SAMs have shown a targeted improvement in organic-TFT performance by influencing the growth mode of organic semiconductors like pentacene, C_{60} , etc. and improving charge-transport properties.

Phosphonic acid SAMs assemble on a large range of inorganic oxide dielectrics by multidentate binding of strong covalent P-O-M anchoring where M is the metal atom. The most popular combination for phosphonic acid SAMs is with an AlO_x film as the metal oxide component, but phosphonic acid molecules have also been used in combination with HfO₂ films prepared by a sol-gel technique [119,120,173,174]. The underlying AlO_x film can be prepared by a variety of methods such as atomic layer deposition (ALD) [175], anodization [176], spin coating [177, 178], UV/ozone assisted oxidation [179, 180] and plasma-assisted oxidation [37, 181–184]. An extremely detailed review can be found in [185].

As mentioned above, this thesis investigates an ultrathin, hybrid gate dielectric composed of an inorganic metal oxide and an organic self-assembled monolayer (SAM). This approach utilizes the benefits of both components to provide a unique advantage for low-voltage organic TFTs. For the metal oxide component, an aluminum oxide (AlO_x) film is fabricated by oxidizing the surface of the underlying aluminum gate electrode in a radio-frequency generated oxygen plasma. The SAM component constitutes phosphonic acid molecules with an aliphatic chain, either an unsubstituted alkyl chain or a substituted fluoroalkyl chain, assembled from solution.

3.3 Importance of individual components of a hybrid gate dielectric

After reviewing the different types of gate dielectrics from literature, this section shows the original work in this thesis of fabricating organic TFTs with different gate dielectric configurations. The aim is to highlight the importance and action of each of the two components in a hybrid gate dielectric, i.e. the metal oxide and the self-assembled monolayer, and outline their contributions to the overall benefit and application of hybrid gate dielectrics in organic TFTs.

Fig.3.5 and Fig.3.6 show transfer characteristics and gate currents of TFTs employing the following two small-molecule organic semiconductors:

- dinaphtho[2,3-b:2',3'-f]thieno[3,2-b]thiophene (DNTT)
- 2,7-diphenyl[1]benzothieno[3,2-b][1]benzothiophene (DPh-BTBT)

Three different gate-dielectric configurations were selected for the organic TFTs with each of the organic semiconductors-

• (a) SAM dielectric: the gate dielectric was formed only using the solutionprocessed SAM directly on the native oxide of the aluminum gate electrode, i.e. without exposing the aluminum gate electrode to an oxygen plasma.

- (b) Oxide dielectric: the gate dielectric was formed by exposing the aluminum gate electrode to an oxygen plasma, but without the solution-processed SAM.
- (c) Hybrid gate dielectric: the gate dielectric was formed by both, the oxygen plasma treatment to form the aluminum oxide film as well as the solution-processed SAM.

An alkyl phosphonic acid molecule with 14 carbon atoms, n-tetradecylphosphonic acid $(HC_{14}PA)$, was chosen to form the SAM for the DNTT TFTs and a fluoroalkyl phosphonic acid molecule with 10 carbon atoms, 1H,1H,2H,2H-perfluorodecylphosphonic acid $(FC_{10}PA)$, was chosen to form the SAM for the DPh-BTBT TFTs. The choice of the molecule for the SAMs for the respective organic semiconductor TFTs is further explained in Chapter 6 and Appendix D. By fabricating transistors with individual components of the gate dielectric and comparing them with the hybrid gate dielectric-TFTs, the importance and significance of each of these is established.



Figure 3.5: Transfer characteristics and gate currents of DNTT TFTs fabricated with either (a) a SAM dielectric (without oxygen plasma), (b) a bare-AlO_x dielectric (without SAM) and (c) a hybrid AlO_x/SAM gate dielectric. The molecule chosen to form the SAM is $HC_{14}PA$

3.3.1 The oxide component of the hybrid gate dielectric

Considering the DNTT TFTs in Fig.3.5, the importance of the oxygen-plasma process for the formation of the AlO_x film as well as towards the overall functioning of the gate dielectric is evident. It can be observed in Fig.3.5(a) that the gate leakage current is several orders of magnitude higher when the oxygen-plasma process was skipped. In the absence of the oxygen-plasma process, the native AlO_x film serves as the gate oxide and



Figure 3.6: Transfer characteristics and gate currents of DPh-BTBT TFTs fabricated with either (a) a SAM dielectric (without oxygen plasma), (b) a bare-AlO_x dielectric (without SAM) and (c) a hybrid AlO_x/SAM gate dielectric. The molecule chosen to form the SAM is $FC_{10}PA$

density of surface hydroxyl groups (-OH) is low. In such a case, the adsorption of the phosphonic acid molecules is also hindered and the overall thickness of the gate dielectric is extremely small, essentially limited to that of the native AlO_x film. The small thickness and poor quality of the gate dielectric in this case explains the large leakage current (10⁻⁷ A) as seen in Fig.3.5(a). For the DNTT TFT shown in Fig.3.5(b), the plasma-oxidized AlO_x film acts as the gate dielectric, and even without the solution-processed SAM, the gate current through the gate dielectric can be limited to approximately 10⁻¹¹ A. Both these instances establish the importance of the oxygen-plasma process in forming a high quality AlO_x film as well as creating a sufficient density of surface hydroxyl groups for the formation of the SAM.

The importance of the oxygen plasma and the AlO_x component for DPh-BTBT TFTs, employing a fluoroalkyl phosphonic acid SAM, is not as significant as the DNTT TFTs, but nevertheless also important enough to cause a variation in the gate current. Without the oxygen-plasma grown AlO_x film, the gate current in the transistor (as seen in Fig.3.6(a)) is almost 10^{-9} A, which is one to three orders of magnitude higher than the TFTs employing the oxygen-plasma grown AlO_x film in the gate dielectric.

The AlO_x film in these DNTT and DPh-BTBT TFTs was formed by exposing the underlying aluminum gate electrode to an oxygen plasma with the following plasma process parameters: oxygen flow rate 30 sccm, oxygen partial pressure 0.01 mbar, RF power 200 W and duration 30 seconds. Chapter 4 investigates the effect of varying the plasma power and plasma duration between 10 W-300 W and 10 s-1800 s respectively on the resulting AlO_x film properties and thereby the electrical properties of the TFTs. Chapter 5 further investigates the thickness and the composition of the plasma-oxidized AlO_x films.

3.3.2 The SAM component of the hybrid gate dielectric

Fig.3.5(b) shows the transfer characteristics and gate current of a DNTT TFT with a gate dielectric composed of only the plasma-grown AlO_x , i.e. without the solutionprocessed SAM. Comparing the performance of this TFT to the one displayed in Fig.3.5(c), it can be observed that employing the SAM in the hybrid gate dielectric reduces the overall gate current by almost an order of magnitude from 10^{-10} A to 10^{-11} A. More significantly, it can be observed that the TFT with the hybrid gate dielectric has a charge-carrier mobility of $1.9 \text{ cm}^2/\text{Vs}$, which is much larger than the charge-carrier mobility of the TFT with only the AlO_x dielectric, which has a mobility of $0.4 \text{ cm}^2/\text{Vs}$. This is likely due to the action of the SAM, which helps in creating a suitable surface for the growth of the organic semiconductor (DNTT in this case), as well as screening the polaronic charge formation between the AlO_x and the charge carriers in the organic semiconductor channel. These phenomenon will be further discussed in Chapter 4 and Chapter 6.

Considering the DPh-BTBT TFTs with the different gate-dielectric configurations, it can be observed that by employing the fluoroalkyl phosphonic acid SAM in the hybrid gate dielectric, shown in Fig.3.6(c), the gate leakage current through the dielectric is reduced to a minimum value of 10^{-12} A, significantly lower than the gate current of either of the two single-component dielectric TFTs in Fig.3.6(a) or Fig.3.6(b). It can be observed that DPh-BTBT as an organic semiconductor is relatively less sensitive to the dielectric in terms of the charge-carrier mobility of the TFTs. However, an important advantage and function of using functionalized SAMs can be observed. By employing the fluoroalkyl phosphonic acid SAM in the gate dielectric, either by itself or as a part of the hybrid dielectric, as shown in Fig.3.6(a) and Fig.3.6(c) respectively, the threshold voltage of the transistor can be shifted to more positive values. This is an important function of SAMs in hybrid gate dielectrics for organic TFTs and will be explored in depth in Chapter 6.

In general, it can be observed that the AlO_x component formed by the oxygen plasma helps in minimizing the gate current of the transistor by forming a dense oxide dielectric and providing a surface for the SAM to assemble while the SAM contributes not only to reduction in the gate leakage current but also in increasing the charge-carrier mobility in DNTT TFTs and favorably shifting the threshold voltage in DPh-BTBT TFTs.

3.4 Experimental



Figure 3.7: Schematic cross-section of the organic TFTs fabricated in this thesis chapter along with the chemical structures of the organic semiconductors DNTT and DPh-BTBT and the different alkyl and fluoroalkyl phosphonic acid molecules chosen to form SAMs in the hybrid gate dielectrics

The TFTs demonstrated in Fig.3.5 and Fig.3.6 were fabricated on flexible, 125 μ mthick polyethylene naphthalate (PEN) substrates. For the gate electrode of the TFTs, aluminum films with a thickness of 30 nm were deposited by thermal evaporation in vacuum with a rate of about 20 Å/s. AlO_x films were produced by plasma oxidation in pure oxygen with a partial pressure of 0.01 mbar using the capacitively coupled plasma mode with an excitation frequency of 13.56 MHz. The RF power applied to generate the plasma was 200 W and the duration was limited to 30 s. After the plasma oxidation, SAMs were formed by immersing the substrates into a 2-propanol solution with a concentration of 1 mM for a minimum of 3 hours of the respective molecules used to form the SAM.

For the TFTs in Fig.3.5, n-tetradecylphosphonic acid, (HC₁₄PA; PCI Synthesis, Newburyport, MA, USA) molecules was used to form a self-assembled monolayer. The smallmolecule organic semiconductor dinaphtho[2,3-b:2',3'-f]thieno[3,2-b]thiophene (DNTT; Sigma Aldrich) was deposited by thermal sublimation in vacuum with a deposition rate of 0.3 Å/s and with a nominal thickness of 25 nm. During the DNTT deposition, the substrate was held at a constant temperature of 60 °C.

For the TFTs in Fig.3.6, 1H,1H,2H,2H-perfluorodecylphosphonic acid, (FC₁₀PA; Specific

Polymers) molecules was used to form a self-assembled monolayer. The small-molecule organic semiconductor 2,7-diphenyl[1]benzothieno[3,2-b][1]benzothiophene (DPh-BTBT; Sigma Aldrich) was deposited by thermal sublimation in vacuum with a deposition rate of 0.3 Å/s and with a nominal thickness of 25 nm. During the DPh-BTBT deposition, the substrate was held at a constant temperature of 25 °C.

For the source-drain contacts of the TFTs, gold films with a thickness of 30 nm were deposited by thermal evaporation in vacuum with a rate of 0.3 Å/s. The metals and the organic semiconductors were patterned using polyimide shadow masks (CADiLAC Laser, Hilpoltstein, Germany). The TFTs have a channel length of 20 μ m and a channel width of 100 μ m. All electrical measurements were performed in ambient air at room temperature under yellow laboratory light.

3.5 Summary and outlook

This chapter has introduced the hybrid gate dielectric that will be further explored in this thesis, i.e. a combination of aluminum oxide and phosphonic acid SAMs. It has been shown that hybrid dielectrics are hugely successful in forming high-quality gate dielectrics with low leakage-current densities to enable low-voltage and low-power organic TFT operation. The contribution of each of the components forming the hybrid dielectric has been emphasized. Apart from the work in this thesis, a deeper investigation into the characterization of the SAMs (for their packing density and molecular order), precise control of TFT properties using SAMs, multiple combinations of SAMs and organic semiconductors and a comprehensive influence of the SAM on the organic semiconductor would greatly contribute to the understanding of the precise functioning of SAMs and the development of organic TFTs.

Chapter 4

Optimizing plasma oxidation for aluminum oxide films in hybrid gate dielectrics

The first component of the hybrid gate dielectrics investigated in this thesis is an ultrathin aluminum oxide (AlO_x) film fabricated by the plasma-assisted oxidation of the surface of the underlying aluminum gate electrode in organic TFTs. As already mentioned in Chapter 3, there are several methods of forming or depositing metal-oxide films for gate dielectrics in organic TFTs, with techniques such as atomic layer deposition [112, 175], anodic oxidation [176, 186], UV/ozone-assisted oxidation [179, 180], and plasma-assisted oxidation used to fabricate AlO_x films. By exposing the surface of the aluminum gate electrode to an oxygen plasma, the high-energy oxygen ions in the plasma impinge on and diffuse into the aluminum film, and oxidize the surface to form a thin film of aluminum oxide in a self-limiting process. The plasma-assisted oxidation process has several advantages and is extremely well-suited for the purpose of fabricating high-capacitance gate dielectrics in low-voltage organic TFTs. The plasma-oxidation process helps to greatly simplify the fabrication process because unlike in anodic oxidation, electrical contact to the gate metal is not required [187]. Another main advantage is that the plasma-oxidation process is area-selective, i.e. the oxide is formed only where the aluminum film is present and where the oxide is required for TFT operation. This eliminates the need of any additional subtractive patterning process to open vias for

interconnects [188]. The greatest advantage is the formation of a high-quality, native interface between the gate metal and the gate oxide, which minimizes hysteresis in the current–voltage characteristics and the subthreshold swing of the TFTs [189, 190]. In addition, the plasma-oxidation process is also compatible with alternative substrates such as flexible plastic foils and paper, suitable for the potential applications of organic TFTs.

There have been several reports of using plasma-assisted oxidation to fabricate AlO_x films as components of hybrid gate dielectrics in organic TFTs on a variety of substrates such as silicon wafers [183, 184], glass [35, 182, 191], flexible plastic foils [181] and paper [37]. However, in the absence of a direct control of the oxide film properties during plasma oxidation, the influence of the different process parameters on the properties of the AlO_x films and thereby on the organic TFT characteristics needs to be investigated. In this chapter, the process of plasma oxidation to prepare the AlO_x films is optimized to determine the ideal process parameters, specifically the plasma power and the duration of plasma exposure. Different combinations of the plasma power and the plasma duration were selected to fabricate AlO_x films as gate dielectrics in capacitors as well as in organic TFTs based on the small-molecule organic semiconductor DNTT. The influence of these parameters on the material and film properties of the AlO_x films, either as bare- AlO_x gate dielectrics or hybrid AlO_x/SAM gate dielectrics are investigated and are correlated to subsequent effects on the semiconductor morphology and the electrical properties of organic TFTs.

Among the surface properties of the gate dielectric, the surface roughness has a large influence on the morphology of the semiconductor in bottom-gate devices and the overall performance of organic TFTs. Several reports in literature have shown the negative impact of the surface roughness of the gate dielectric on the charge-carrier mobility of organic TFTs, i.e. an increased surface roughness of the gate dielectric decreases the charge-carrier mobility in the transistor [108, 192–196]. This effect is mainly attributed to a lower quality of the semiconductor film on rougher gate dielectric surfaces, for example, the semiconductor film has an increased density of grain boundaries, reduced crystallinity, etc. Several strategies have been employed to tune the surface roughness of the dielectric such as modifying the substrate temperature during deposition of the underlying gate electrode [197], adding a polymeric smoothening layer on the rough dielectric [198], annealing [199], etc [200–202]. In this chapter, the influence of the plasma process parameters on the surface roughness of the gate dielectric, and subsequently on the semiconductor morphology will be investigated in detail.

When an aluminum metal film is exposed to air, a native aluminum oxide film is spontaneously formed on the surface due to air-oxidation. However, this native oxide film is extremely thin (~1-3 nm [203]) and inadequate as a high-quality gate oxide. Fig.4.1(a) shows the transfer characteristics and gate current for an organic TFT with the native aluminum oxide film as the single gate dielectric, i.e. neither an oxygen-plasma process nor a solution-processed SAM was used. There is no observable field-effect behavior in the device and additionally, a large gate leakage current can be observed (10⁻⁷ A at a gate-source voltage of -3 V). Fig.4.1(b) shows an AFM micrograph of the organic semiconductor DNTT deposited on the native AlO_x surface and it can be seen that the semiconductor does not form a complete film, but only individual, disconnected islands instead. This indicates that the native AlO_x film is not sufficiently thick as a gate dielectric and also does not provide a suitable surface for the deposition of the organic semiconductor.



Figure 4.1: Transfer characteristics and gate currents of DNTT TFTs fabricated using native aluminum oxide (obtained without plasma process) either as (a) the gate dielectric or (c) as part of a hybrid AlO_x/SAM gate dielectric. AFM micrographs of DNTT films deposited onto (b) the bare native AlO_x and (d) the hybrid AlO_x/SAM gate dielectric are also shown.

Fig.4.1(c) shows the transfer characteristics and gate current of an organic TFT in which the gate dielectric is a combination of the native AlO_x film and an alkyl phosphonic acid SAM (HC₁₄PA), i.e. no oxygen-plasma process was used but the SAM was formed on the native oxide surface from solution. The transfer curve shows the presence of a field effect in the device, but the gate current is still extremely large (10⁻⁸ A at a gate-source voltage of -3 V). Fig.4.1(d) shows an AFM micrograph of the organic semiconductor DNTT deposited on the surface of the SAM and a connected, albeit with gaps, semiconductor film can be observed. Although the hybrid gate dielectric formed by the native AlO_x and the SAM forms a suitable surface for organic semiconductor deposition and fieldeffect behavior can be achieved in the transistor, the gate dielectric is of extremely poor quality as evidenced by the large gate current. A high density of hydroxyl groups is required to facilitate the adsorption of the SAM. Without the oxygen plasma, there are fewer groups present, which results in a low-quality gate dielectric.

This shows the requirement for the oxygen-plasma process to produce a thin, yet highquality AlO_x film on the surface of the underlying aluminum gate electrode to form high-quality hybrid gate dielectrics in organic TFTs.

4.1 Thickness of plasma-grown AlO_x films



Figure 4.2: Thickness of plasma-grown AlO_x films. (a) Cross-sectional TEM (Transmission electron microscopy) image indicating the thicknesses of five AlO_x films produced sequentially by the plasma-assisted surface oxidation of aluminum using five different combinations of plasma power and plasma duration. (b) Summary of the results

Plasma power (W)	Plasma duration (s)	Aluminum oxide thickness (nm)
10	30	4.3 ± 0.2
10	300	5.8 ± 0.3
200	10	5.8 ± 0.1
200	60	7 ± 0.3
300	30	7.3 ± 0.4

Table 4.1: Thickness of AlO_x films (with error margins) prepared by different combinations of the plasma power and plasma duration as determined by TEM

The thickness of the gate dielectric in a transistor is of extreme importance as it determines the capacitance of the gate dielectric and thereby the operating voltage of the transistor. In certain fabrication techniques there is higher degree of control over the film thickness. For example, during atomic layer deposition (ALD), the thickness of the deposited films can be controlled by the number of cycles in the process. In thermal or electron-beam evaporation, the thickness is controlled by monitoring the rate of deposition using a quartz microbalance. With plasma oxidation, there is no such direct control of the film thickness during the growth process. Hence, there is a need to identify means of controlling the film thickness in an indirect manner. Five combinations of the plasma power and the plasma duration were chosen in five different processes to prepare AlO_x films of varying thicknesses. Fig.4.2(a) shows a cross-sectional TEM (transmission electron microscopy) image of a specimen prepared on a silicon substrate by repeating the deposition of 30-nm-thick aluminum and the plasma-assisted oxidation of the aluminum surface five times, each time with a different combination of the plasma power and plasma duration. The five combinations of process parameters selected were: (1) 10 W, 30 s (2) 10 W, 300 s (3) 200 W, 10 s (4) 200 W, 60 s and (5) 300 W, 30 s. During each transfer of the substrate from the metal-deposition system to the plasma system, the aluminum surface was necessarily exposed to ambient air, causing the spontaneous formation of a native oxide film with a thickness of approximately 3 nm on the aluminum surface [203]. The thickness of the five different AlO_x films along with the error margin as determined from the TEM image are given in Table 4.1. The TEM image indicates that after the plasma-assisted oxidation, the aluminum oxide films have a thickness of approximately 4.3 nm to 7.3 nm, depending on the plasma parameters. (The method of extracting the thickness of the AlO_x films from the TEM image is detailed in Section 4.6). The lower limit of this thickness range (4.3 nm) is thus larger by approximately 1.3 nm than the thickness of the native oxide. Although this increase in the oxide thickness produced by the plasma process is quite small, it is of critical

importance for the proper operation of organic TFTs in which these dielectrics are used as the gate insulator, as seen in the previous section. The relation between the plasma parameters and the thickness of the plasma-grown AlO_x films extracted from the TEM image is also illustrated in Fig.4.2(b). As expected, both a larger plasma power (by virtue of a higher kinetic energy of the oxygen radicals impinging on the oxide surface) and a longer plasma duration (by virtue of a larger number of incident radicals) result in thicker AlO_x films. Thus, an AlO_x film of a desired thickness within the available range of 4.3 nm to 7.3 nm can be fabricated using the plasma-assisted oxidation process by selecting a suitable combination of parameters as seen in Fig.4.2(b).

4.2 Dielectric properties of plasma-grown AlO_x films

One of the most important figures of merit for a gate dielectric is the gate-dielectric capacitance, because a large gate-dielectric capacitance makes it possible to operate the transistor with low operating voltages. This section examines the electrical properties of oxygen-plasma grown AlO_x films employed in metal-insulator-metal capacitors, either as bare- AlO_x dielectrics or as hybrid AlO_x/SAM dielectrics. Fifteen different combinations of plasma power (ranging from 10 to 300 W) and plasma duration (ranging from 10 to 1800 s) were selected to fabricate the AlO_x films. A schematic cross-section of the capacitors, the materials and techniques utilized to fabricate them and a photograph are included in Section 4.6.



Figure 4.3: Measured unit-area capacitance of capacitors with (a) a bare- AlO_x dielectric and (b) a hybrid AlO_x/SAM dielectric as a function of plasma power and plasma duration

The measured unit-area capacitance of the capacitors with a bare- AlO_x dielectric is plotted as a function of the plasma power and the plasma duration in Fig. 4.3(a). Depending on these parameters, the unit-area capacitance varies from 1 to 1.6 $\mu F/cm^2$, with the general trend of a higher power and a longer duration producing AlO_x films with a smaller capacitance. Fig.4.3(a) reveals that the influence of the plasma duration on the capacitance is relatively small, as long as the plasma power is at least 50 W and the duration is at least 30 s. For example, for a plasma power of 200 W, a unit-area capacitance of 1 μ F/cm² is obtained for a plasma duration of one minute, as well as for a much longer plasma duration of half an hour. Although it is possible to use the bare, plasma-grown AlO_x films as the gate dielectric for organic TFTs [156, 204, 205], hybrid gate dielectrics composed of the plasma-grown AlO_x and a phosphonic acid-based self-assembled monolayer (SAM) are far more advantageous, as already discussed in Section 3.3. Fig.4.3(b) shows the unit-area capacitance of the capacitors with a hybrid AlO_x/SAM dielectric, plotted as a function of the plasma power and plasma duration. Due to the additional contribution of the SAM, the capacitance of the hybrid AlO_x/SAM dielectric (0.7–0.8 $\mu F/cm^2$) is smaller than that of the bare-AlO_x dielectric (1–1.6 $\mu F/cm^2$) and shows a notably smaller dependence on the plasma power and duration.

A few important conclusions can be drawn by combining the results from the unit-area capacitance measurements and the TEM results. The thickness of the plasma-grown AlO_x films is observed to saturate at a value of approximately 7 nm after a plasma duration of 30 to 60 seconds, provided the plasma power is at least 50 W. The selflimiting behavior of the growth of the oxide films can be explained by the Cabrera-Mott model [206], in which the low electronic conductivity of aluminum oxide and the small diffusivity of oxygen in aluminum oxide prevent oxygen from reaching the metal surface, once the oxide thickness has reached a certain value determined mainly by the plasma power [207,208]. Compared with the dependence of the capacitance on the plasma duration, its dependence on the plasma power appears to be more monotonic, but overall, the range over which the capacitance of the plasma-grown AlO_x can be tuned is nevertheless quite small (less than a factor of two). One benefit of this small range of accessible capacitances is that it renders the fabrication process more robust by suppressing the effects of unintended process-parameter variations on the resulting TFT characteristics. Fig.4.4 illustrates the unit-area capacitance measured for each of the five parameter combinations for which the AlO_x thickness was determined by TEM (Fig. 4.2) plotted as a function of the inverse of that thickness. The error bars reflect the accuracy of the method by which the oxide thickness was extracted from the TEM image (Section 4.6). Since it was not possible to include all fifteen combinations of the plasma power and plasma duration for the TEM measurement, the five combinations were chosen to cover both the extremes and the center of the range of thicknesses and capacitances as much as possible. By fitting the measurement data with the theoretical relationship between the unit-area capacitance C_{ox} and the oxide thickness t_{ox} :

$$C_{\rm ox} = \varepsilon_0 \varepsilon_{ox} \frac{1}{t_{\rm ox}} \tag{4.1}$$

(where ε_0 is the vacuum permittivity and ε_{ox} the relative permittivity of the plasmagrown oxide) and forcing the linear fit through the origin $(1/t_{ox} = 0; C_{ox} = 0)$, the relative permittivity of the plasma-grown AlO_x is determined to be 8 ± 0.2 . This result is in good agreement with the relative permittivity reported in the literature for aluminum oxide films produced by various methods [209–211].



Figure 4.4: Calculation of the relative permittivity of plasma-grown AlO_x : Unit-area capacitance of bare-AlO_x dielectrics plotted as a function of the inverse of the AlO_x thickness determined by TEM to calculate the relative permittivity of the plasma-grown AlO_x . The red line is a linear fit of Eqn.4.1 that was forced to pass through the origin $(1/t_{ox} = 0; C_{ox} = 0)$.

The total capacitance of the hybrid AlO_x/SAM gate dielectric (C_{diel}) can be considered as a combination of the two individual capacitances in parallel:

$$\frac{1}{C_{\text{diel}}} = \frac{1}{C_{\text{ox}}} + \frac{1}{C_{\text{SAM}}}$$

$$= \frac{t_{\text{ox}}}{\varepsilon_0 \varepsilon_{\text{ox}}} + \frac{t_{\text{SAM}}}{\varepsilon_0 \varepsilon_{\text{SAM}}}$$
(4.2)

(where C is the unit-area capacitance, t is the thickness, with the subscript ox referring to the oxide and SAM referring to the SAM. ε_0 is the absolute permittivity.)

The reported value of the relative permittivity of the SAM (ε_{SAM}) is 2.5 [139], which is much smaller than the relative permittivity of the oxide as calculated above ($\varepsilon_{ox} = 8$). Comparing these values in the context of Eqn.4.2, the term $\frac{1}{C_{\text{SAM}}}$ dominates the value of the total gate-dielectric capacitance, and the influence of the oxide capacitance is minimized. This explains why the hybrid gate-dielectric capacitance does not vary much with the plasma power and plasma duration in Fig.4.3(b).



Figure 4.5: Current density measured as a function of applied voltage through capacitors with (a) a bare-AlO_x dielectric and (b) a hybrid AlO_x/SAM dielectric for various combinations of plasma power and plasma duration.

In addition to the capacitance, the current–voltage characteristics of the capacitors were also measured to analyze the influence of the plasma power and duration on the leakagecurrent density through the dielectrics. The results for the capacitors with a bare-AlO_x dielectric are summarized in Fig.4.5(a). The general trend is similar to the one seen for the capacitance in Fig.4.3(a) and is consistent with the TEM results: a higher plasma power and a longer plasma duration lead to thicker AlO_x films characterized by smaller leakage-current densities. When the plasma power is smaller than about 50 to 100 W and the plasma duration is shorter than about 30 to 60 s, the leakage-current density through the bare-AlO_x films exceeds 10^{-4} A/cm² at ± 3 V. On the other hand, for a plasma power of 200 W and a plasma duration of 60 s, the current density through the bare-AlO_x dielectrics is below 10^{-5} A/cm² at ± 3 V. Fig. 4.5(a) also shows that it is not possible to produce bare- AlO_x dielectrics by plasma oxidation that provide leakage-current densities significantly below 10^{-6} A/cm² at ± 3 V. The results of the measurements of the current density through the hybrid AlO_x/SAM dielectric are summarized in Fig.4.5(b). As can be seen, the leakage-current density through the hybrid AlO_x/SAM dielectrics is smaller by approximately an order of magnitude than the current density through the bare-AlO_x dielectrics, regardless of the plasma parameters. For a plasma power of 200W and a duration of 60 s, the leakage-current density through the hybrid AlO_x/SAM dielectric drops below 10^{-6} A/cm². While this confirms the beneficial effect of the SAM in improving the insulating properties of the gate dielectric for low-power organic TFTs, the results in Fig.4.5(b) also clearly demonstrate the critical importance of providing an optimized AlO_x film, even when complementing it with a SAM.

4.3 Electrical properties of organic TFTs with plasmagrown AlO_x films as gate dielectrics

On the same substrates as the capacitors discussed above, bottom-gate, top-contact DNTT TFTs were also fabricated with either a bare-AlO_x or a hybrid AlO_x/SAM gate dielectric produced with the same fifteen combinations of plasma power and duration as discussed above. Fig.4.6 (bare-AlO_x gate dielectric) and Fig.4.7 (hybrid AlO_x/SAM gate dielectric) show the measured transfer characteristics and gate currents of TFTs fabricated with three of these plasma-parameter combinations. The complete set of results is shown in Appendix B.

One observation from Fig.4.6 and Fig.4.7 is that a plasma power of 10 W is insufficient to suppress the gate-leakage current to an acceptable level, regardless of the plasma duration and regardless of whether or not the AlO_x film is complemented with a SAM. The minimum plasma power required to obtain AlO_x films with sufficient thickness and



Figure 4.6: Transfer characteristics and gate currents of DNTT TFTs fabricated with a bare-AlO_x gate dielectric for three combinations of plasma power and plasma duration. (Appendix B shows the complete set of results)

sufficient quality is 50 W, which limits the gate current to 10^{-11} A over the range of gate-source voltages considered here (0 to -3 V). For a plasma power in the range of 100 to 200 W, both the bare-AlO_x and the hybrid AlO_x/SAM gate dielectric are able to provide gate currents not exceeding 10^{-11} A and on/off current ratios of 10^6 . The optimum plasma duration for this range of plasma power is dictated by whether or not the AlO_x is complemented by a SAM: For the bare-AlO_x gate dielectric, the optimum plasma duration is ≥ 300 s, whereas for the hybrid AlO_x/SAM gate dielectric, the optimum plasma duration is in the range of 30 to 60 s (for a plasma power ranging from 100 to 200 W).



Figure 4.7: Transfer characteristics and gate currents of DNTT TFTs fabricated with a hybrid AlO_x/SAM gate dielectric for three combinations of plasma power and plasma duration. (Appendix B shows the complete set of results)

In Fig.4.8(a), the effective charge-carrier mobility extracted from the measured transfer characteristics of the DNTT TFTs with a bare- AlO_x gate dielectric is plotted as a function of the plasma power and the plasma duration. As can be seen, the carrier mobility of these TFTs is rather small, between 0.1 and 0.6 $\rm cm^2/Vs$, which is possibly related to charge trapping and the formation of Fröhlich polarons, due to the fact that the organic semiconductor is in direct contact with the aluminum oxide [151, 212]. The dependence of the carrier mobility of the TFTs with a hybrid AlO_x/SAM gate dielectric on the plasma parameters is summarized in Fig.4.8(b). The beneficial contribution of the SAM, in providing a hydrophobic surface with a greatly reduced density of waterrelated trap sites and effective screening of the organic-semiconductor film from the highpermittivity oxide, leads to notably larger carrier mobilities of up to $2.3 \text{ cm}^2/\text{Vs}$. Closer inspection of Fig.4.8(b) shows that carrier mobilities of approximately $2 \text{ cm}^2/\text{Vs}$ are obtained along a track from the upper left to the lower right corner of the graph, i.e., from low-power/long-duration to high-power/short-duration combinations. For parameter combinations outside of this corridor, the carrier mobilities are notably smaller, as small as $0.2 \text{ cm}^2/\text{Vs}$. The reasons for this distinctive parameter-dependence pattern will be elucidated in the following section.



Figure 4.8: Effective charge-carrier mobilities extracted from the measured transfer characteristics of DNTT TFTs fabricated with either (a) a bare-AlO_x gate dielectric or (b) a hybrid AlO_x/SAM gate dielectric as a function of plasma power and plasma duration

4.4 Surface properties

The performance of field-effect transistors in general and of organic TFTs in particular is greatly dependent on the properties of the interface between the semiconductor and the gate dielectric [185]. In the case of bottom-gate TFTs, as considered here, this highlights the critical importance of the properties of the gate-dielectric surface. The surface energy and the surface roughness of the bare-AlO_x and the hybrid AlO_x/SAM dielectrics has been measured for each of the fifteen combinations of plasma power and plasma duration discussed above.



Figure 4.9: Surface energy of (a) bare-AlO_x dielectrics and (b) hybrid AlO_x/SAM dielectrics as a function of plasma power and plasma duration

The surface energies of both the bare-AlO_x and the hybrid AlO_x/SAM gate dielectrics show only very small variations and no systematic dependence on the plasma parameters, as shown in Fig.4.9. The surface energy of the bare-AlO_x dielectric varies between 61 and 71 mJ/m², and that of the hybrid AlO_x/SAM dielectric between 23 and 26 mJ/ m^2 , similar to previous reports [120, 155].

The surface roughness, on the other hand, shows a clear correlation with the plasma parameters. Prior to the plasma-oxidation process, the vacuum-deposited aluminum has a root-mean-square (RMS) surface roughness of 0.9 nm [197]. After the plasma-oxidation process, the RMS surface roughness of the bare AlO_x ranges from 0.34 to 0.91 nm, depending on the plasma parameters (shown in Fig.4.10(a)). The general trend seen in Fig.4.10(a) is that higher plasma power and longer plasma duration lead to smoother AlO_x films. Given the difference between the RMS surface roughness of the aluminum



Figure 4.10: Root-mean-square surface roughness of (a) bare-AlO_x dielectrics and (b) hybrid AlO_x/SAM dielectrics as a function of plasma power and plasma duration. (c) Height-height correlation functions (HHCF) of DNTT films deposited onto bare-AlO_x dielectrics for five combinations of plasma power and plasma duration. (d) Lateral correlation length ξ of the DNTT films as a function of the surface roughness of the bare-AlO_x dielectric.

prior to plasma oxidation (0.9 nm) and the RMS surface roughness of the plasmagrown AlO_x (0.34 to 0.91 nm), it appears that the plasma-oxidation process smoothens the surface, most prominently for sufficiently high plasma powers and sufficiently long durations. This effect was not observed in a similar study [197], in which an almost identical RMS surface roughness of 0.9 nm was measured for both the vacuum-deposited aluminum and the gate dielectric. However, that study did not explore the use of high plasma powers or long plasma durations, which might explain why no smoothening was observed.

The RMS surface roughness of the hybrid AlO_x/SAM gate dielectrics as a function of plasma power and duration is shown in Fig.4.10(b). Comparing Fig.4.10(a) and Fig.4.10(b) shows that the functionalization of the AlO_x surface with the SAM has no measurable effect on the roughness, i.e., the SAM covers the AlO_x surface in a conformal manner, as expected. The degree of the surface roughness of the gate dielectric affects the properties of the organic-semiconductor film deposited onto it [197]. DNTT films were deposited on the different AlO_x and AlO_x/SAM surfaces, and the morphology was recorded using AFM (shown in Fig.4.11). To evaluate the DNTT thin-film morphology in a quantitative manner, the height-height correlation function (HHCF) was calculated from the AFM data using the following equation [213]:

$$H_{\rm x}(\tau_{\rm x} = \frac{1}{N(M-m)} \sum_{l=1}^{N} \sum_{n=1}^{M-1} \left(z_{\rm n+m,l} - z_{\rm n,l} \right)^2 \tag{4.3}$$

(where N and M are the number of measured rows and columns, z is the height of a measurement point, Δx is the sampling interval along the x direction, m an integer (0 $\leq m \leq M$), and τx is the lateral distance ($\tau x = m \Delta x$).)

The HHCF has two distinct regimes, as seen in Fig.4.10(c): over short lateral distances, the heights are correlated and the HHCF increases linearly with distance. Over long distances, the heights are uncorrelated and the HHCF saturates at a value proportional to $2R_{RMS}^2$, where R_{RMS} is the root-mean-square surface roughness [214]. The lateral distance at which the crossover between the two regimes occurs is the lateral correlation length ξ . In Fig.4.10(d), the lateral correlation length determined for each of the DNTT films deposited onto bare AlO_x (Fig.4.10(a)) is plotted as a function of the RMS surface roughness of the AlO_x . As can be seen, the lateral correlation length of the DNTT films shows a monotonic dependence on the AlO_x surface roughness, increasing from 0.2 nm for the largest surface roughness to 1.5 nm for the smoothest surface. However, these trends are not reflected in the measured charge-carrier mobilities of the TFTs in which these bare-AlO_x films serve as the gate dielectric (Fig.4.8(a)), since the carrier mobility in these DNTT films is greatly suppressed, presumably by charge trapping and polaronic effects resulting from the close proximity of the DNTT and the aluminum oxide.



Figure 4.11: Morphology of DNTT films, imaged by AFM, deposited onto bare-AlO_x dielectrics and hybrid AlO_x/SAM dielectrics for five combinations of plasma power and plasma duration

The influence of the variation in surface roughness of the gate dielectric on the thinfilm morphology of the semiconductor can be seen in Fig.4.11, in which AFM images show the morphology of DNTT deposited onto bare-AlO_x and hybrid AlO_x/SAM gate dielectrics produced using different plasma parameters. When DNTT is deposited onto a smooth dielectric surface, the DNTT film exhibits a pronounced terrace-like structure, whereas deposition onto a rough dielectric surface results in a notably smaller terrace size. Considering the DNTT films on the AlO_x/SAM hybrid gate dielectric surface, the plasma-parameter combination of 10 W, 30 s, which yielded a surface roughness of 0.93 nm (Fig.4.10(b)), leads to an island-like formation in the DNTT morphology, unsuitable for high-performance organic TFTs. On the other hand, for the plasmaparameter combination of 200 W, 60 s, which yielded a surface roughness of 0.52 nm (Fig.4.10(b)), an optimum terrace-like structure in a complete, continuous DNTT film can be seen. With an increase in either the plasma power or the plasma duration beyond this combination, a degradation in the DNTT film morphology can be observed, with either cracks appearing in the film, or an island-like discontinuous film structure.

4.5 Correlation of surface properties of the dielectric with electrical properties of the TFTs

The extent to which the carrier mobility of the TFTs with the hybrid AlO_x/SAM gate dielectric correlates with its surface roughness can be seen in Fig.4.12. The largest RMS surface roughness (0.93 nm; obtained with low plasma power/short duration; indicated in grey in Fig.4.12(c)) leads to a disordered DNTT film with small grains that shows a



Figure 4.12: (a) Root-mean-square surface roughness of hybrid AlO_x/SAM dielectrics as a function of plasma power and plasma duration (b) Carrier mobilities extracted from the measured transfer characteristics of DNTT TFTs fabricated with a hybrid AlO_x/SAM gate dielectric as a function of plasma power and plasma duration (c) Carrier mobility plotted as a function of the RMS surface roughness (d), (e) AFM images of representative morphologies on the indicated surfaces with the indicated plasma parameters

very small carrier mobility (0.2 cm²/Vs), as expected. The largest carrier mobilities ($\geq 2 \text{ cm}^2/\text{Vs}$) are obtained only when the RMS surface roughness is below approximately 0.65 nm (indicated in green in Fig.4.12(c)), so that the DNTT morphology shows a pronounced terrace-like structure; this is also in line with expectations. On the other hand, the smallest RMS surface roughness (0.38 nm; obtained with medium power/long duration; indicated in blue in Fig.4.12(c)) does not lead to the largest carrier mobility, as one might have expected, but instead to a very small mobility (0.2 cm²/Vs). The reason for this anomaly is revealed by the AFM image of this dielectric (Fig.4.12(d)), in which a large density of small, tall features can be seen protruding from the surface. These features do not significantly contribute to the calculated RMS surface roughness, but they influence the DNTT morphology in an unfavorable manner, as seen in Fig.4.12(e). The specifics of these tall features are unknown, but the fact that they appear only for long plasma durations (≥ 300 s) suggests that they mark some form of mechanical

damage created on the AlO_x surface by prolonged plasma exposure. AFM images of DNTT films deposited onto hybrid AlO_x/SAM dielectrics for all fifteen combinations of plasma power and plasma duration are collected in Appendix B.

4.6 Experimental



Figure 4.13: Schematic cross sections and photographs of metal-insulator-metal capacitors and bottom-gate, top-contact TFTs in which the insulator or gate dielectric is either a film of plasma-grown AlO_x or a combination of plasma-grown AlO_x and an ntetradecylphosphonic acid (HC₁₄PA) SAM. Also shown are the chemical structure of ntetradecylphosphonic acid (HC₁₄PA) and of the organic semiconductor dinaphtho[2,3b:2',3'-f]thieno[3,2-b]thiophene (DNTT)

Fabrication of capacitors and TFTs on silicon substrates

Metal-insulator-metal capacitors and inverted staggered (bottom-gate, top-contact) TFTs were fabricated on silicon substrates coated with 100-nm-thick thermally grown silicon dioxide. For the bottom electrode of the capacitors and the gate electrode of the TFTs, aluminum with a thickness of 30 nm and a root-mean-square surface roughness of less than 1 nm (measured by AFM) was deposited by thermal evaporation in vacuum with a rate of about 20 Å/s. AlO_x films were produced by plasma oxidation in an Oxford Instruments ProLab100 Cobra system in pure oxygen with a partial pressure of 0.01 mbar using the capacitively-coupled-plasma mode with an excitation frequency of 13.56 MHz. Substrates with fifteen different combinations of plasma power (ranging from 10 to 300 W) and duration (ranging from 10 to 1800 s) were fabricated. After the plasma oxidation, each substrate was cleaved into two halves. One half was immersed into a 2-propanol solution of n-tetradecylphosphonic acid (HC₁₄PA; PCI Synthesis, Newburyport, MA, USA)

to form a self-assembled monolayer and hence a hybrid AlO_x/SAM gate dielectric. The other half remained without SAM (bare-AlO_x gate dielectric). Onto both halves of each substrate, the small-molecule semiconductor dinaphtho[2,3-b;2',3'-f]thieno[3,2-b]thiophene (DNTT; Sigma Aldrich) was deposited by thermal sublimation in vacuum with a deposition rate of 0.3 Å/s and with a nominal thickness of 25 nm. During the DNTT deposition, the substrate was held at a constant temperature of 80 °C. For the top electrode of the capacitors and the source/drain contacts of the TFTs, gold with a thickness of 30 nm was deposited by thermal evaporation in vacuum with a rate of 0.3 Å/s. The metals and the DNTT were patterned using polyimide shadow masks (CADiLAC Laser, Hilpoltstein, Germany). The capacitors have an area of 200 μ m \times 200 μ m. The TFTs have a channel length of 20 μ m and a channel width of 100 μ m. For each of the fifteen combinations of plasma power and plasma duration, capacitors and TFTs were fabricated on the same substrate to minimize the effects of unintentional parameter variations.

TEM characterization

The TEM specimen was prepared on a silicon substrate by repeating the deposition of 30-nm-thick aluminum and the plasma-induced oxidation of its surface five times, each time with a different combination of plasma power and plasma duration. Preparing all five AlO_x films on the same substrate, rather than on five separate substrates, was helpful in minimizing the time required for thinning the specimen in preparation for cross-sectional microscopy. The TEM specimen was fabricated by conventional focused ion beam (FIB) lift-out using a Thermo Fisher Scientific FEI Scios DualBeam instrument equipped with a gallium source. A platinum strip was deposited to protect the films from ion-beam damage. A lamella with a size of approximately 20 $\mu m \times 10 \mu m$ was released from the substrate and glued to a copper TEM lift-out grid. The lamella was then thinned to a thickness of less than 100 nm using an acceleration voltage of 30 kV and an ion current of initially 500 pA that was successively decreased to 100 pA. Afterwards, the lamella was cleaned using a low-voltage cleaning step with an acceleration voltage of 5 kV and an ion current of 48 pA to remove gallium-beam damage. The TEM image was recorded in bright-field (BF) mode using a Philips CM-200 FEG TEM operated with an acceleration voltage of 200 kV.

78

TEM Analysis



Figure 4.14: Analysis of the cross-sectional TEM image to extract the thicknesses of the AlO_x films produced using five different combinations of plasma power and plasma duration: (10 W, 30 s), (10 W, 300 s), (200 W, 10 s), (200 W, 60 s), (300 W, 30 s).

From the digital data representing the TEM image (Fig.4.14(a)), a depth profile (having a line width of 10 pixels) through the AlO_x film (starting and ending a few nanometers above and below the Al/AlO_x interfaces) representing the average pixel brightness (denoted as "b") was extracted using the image software Gwyddion (Fig.4.14(b)). In the depth profile, the materials Al and AlO_x are distinguished by larger and smaller values of the pixel brightness b, respectively. The second derivative of b with respect to the depth was calculated and plotted as a function of depth to identify the points of inflection (b" = 0) in the depth profile, and the distance between these points was measured as the thickness of the AlO_x film (Fig.4.14(c)). For each of the five AlO_x films, this process was repeated 10 times in various locations across the TEM image.

Electrical characterization

All electrical measurements were performed in ambient air at room temperature under yellow laboratory light. The capacitance measurements were performed using a Hameg HM8118 LCR meter by applying an alternating voltage with an amplitude of 0.2 V and a frequency of 1 kHz. The current– voltage measurements were performed using an Agilent 4156C Semiconductor Parameter Analyzer and the measurement software "SweepMe!" (https:// sweep-me.net).

Surface characterization

AFM images were recorded in air using a Bruker Dimension Icon Atomic Force Microscope in peak force tapping mode (for the DNTT films) or in tapping mode (for the gate dielectrics). Data processing was performed using the AFM analysis software Gwyddion. Static contact-angle measurements were performed using a Krüss contact angle measurement system. The contact angles of water and hexadecane on the AlO_x dielectrics were measured immediately after the plasma treatment of the aluminum films. The contact angles of water and hexadecane on the AlO_x/SAM hybrid dielectrics were measured immediately after the SAM treatment. The surface energies were calculated using the Owens–Wendt method [99].

4.7 Summary and Outlook

Ultrathin, high-capacitance hybrid gate dielectrics based on oxygen-plasma-grown aluminum oxide films in combination with an alkyl phosphonic acid SAM are fabricated for the realization of low-voltage organic TFTs. Depending on the plasma power and the duration of the plasma exposure, the thickness of the plasma-grown AlO_x films can be tuned to values between approximately 4.3 and 7.3 nm. The lower limit of the range of accessible thicknesses is dictated by the thickness of the native aluminum oxide film (approximately 3 nm) that forms spontaneously when the substrates are exposed to ambient air prior to the plasma-oxidation process, while the upper boundary is set by the inherently self-limiting oxidation kinetics of aluminum oxide close to room temperature. The relative permittivity of the oxygen-plasma-grown AlO_x films is approximately 8 ± 0.2 , and the capacitance of the bare-AlO_x films thus falls into the range from 1 to 1.6 μ F/cm2. By allowing a high-quality monolayer of an alkyl phosphonic acid with a medium alkyl-chain length to self-assemble on the freshly grown AlO_x surface, a hybrid AlO_x/SAM dielectric is obtained, the capacitance of which varies between 0.7 and 0.8 $\mu F/cm^2$. The leakage-current density through the hybrid AlO_x/SAM dielectrics was found to be smaller by about an order of magnitude than the current density through bare-AlO_x dielectrics. The importance of optimizing the underlying AlO_x film with suitable plasma parameters to achieve an overall reduced leakage-current density in the AlO_x/SAM hybrid dielectric is established. The effective charge-carrier mobility of DNTT TFTs with a bare-AlO_x gate dielectric is no greater than $0.6 \text{ cm}^2/\text{Vs}$, and shows little dependence on the plasma parameters. In TFTs with a hybrid AlO_x/SAM gate dielectric, carrier mobilities ranging from 1.8 to $2.3 \text{ cm}^2/\text{Vs}$ were obtained for a number of favorable combinations of plasma power and plasma duration that produce AlO_x films with low surface roughness and thus promote the formation of high-quality SAMs and well-ordered DNTT films on these dielectrics. The largest carrier mobility equaling 2.3 cm^2/Vs was obtained for a plasma-parameter combination of 200 W and 60 s.

An important finding of this chapter is the correlation between the surface properties of the gate dielectric and the electrical properties of the organic TFT. The plasma-assisted oxidation process can be used to form a high-quality gate oxide film, and more importantly, to tune the surface roughness of the AlO_x film. An optimum surface roughness obtained by a medium plasma power and a medium plasma duration promotes a favorable organic semiconductor morphology, which is best suited for high charge-carrier mobilities in organic TFTs. However, while the properties of the plasma-grown AlO_x films can be tuned over a certain range without negatively affecting the charge-transport properties in the organic-semiconductor films deposited onto them, this is only true as long as the plasma power and the plasma duration do not exceed values beyond which the quality of the plasma-grown oxide films suffers from surface damage. The nature and extent of this surface damage must be characterized and surface properties other than the surface roughness need to be investigated. This will enrich our understanding of the effect of extreme plasma process parameters on the AlO_x surface, as well as on the performance of organic TFTs.
Chapter 5

Investigating the thickness and composition of plasma-grown aluminum oxide films in hybrid dielectrics

The importance of the ultrathin aluminum oxide (AlO_x) film as the gate oxide in the hybrid gate dielectric, as well as the optimization of the plasma-oxidation process to fabricate the AlO_x film has been established in previous chapters. In Chapter 3, the importance of the AlO_x film towards suppressing the leakage-current density and providing a suitable surface for the SAM to adsorb on was discussed. A range of film deposition and growth techniques are used to fabricate AlO_x films for gate dielectrics in organic TFTs, such as atomic layer deposition (ALD) [112,175,215], plasma-enhanced chemical vapor deposition (PE-CVD) [216], sputtering, pulsed laser deposition (PLD) [217], anodization [115,218], along with solution-processing techniques such as spin-coating [219–221], bar-coating [222], etc. Compared to those techniques, the plasma-oxidation process produces a self-patterned, ultrathin and dense AlO_x film with a high-quality native interface with the underlying aluminum gate electrode. In addition, it is also compatible with alternative substrates such as plastic foils and paper, intended for use in organic TFT applications. However, the plasma process also has certain shortcomings compared to the other deposition and thin-film fabrication techniques. Due to the nature of the process, there is a limited degree of control over the thin-film properties during the plasma-oxidation process, although certain properties of the AlO_x film and their influence on the electrical properties of organic TFTs have already been examined in Chapter 4. For example, AlO_x films of varying thicknesses were fabricated by changing the plasma power and the duration of plasma exposure. The influence of the plasma process on the surface properties of the AlO_x films, particularly the surface roughness was also demonstrated. It has been established that the AlO_x film fabricated by the plasma-oxidation process forms a high-quality gate oxide in hybrid gate dielectrics for low-voltage organic TFTs, making it a worthwhile endeavor to further characterize this film. Knowledge of the film properties such as the thickness of the film and the film composition can lead to a better understanding of the role of the gate dielectric, and can potentially provide better control over the electrical performance in organic TFTs. The standard characterization techniques employed to investigate the thickness and chemical composition of metal oxide films are: X-ray Photoelectron Spectroscopy (XPS), Atomic Emission Spectroscopy (AES), X-Ray Diffraction (XRD), etc. However, due to the nature of the AlO_x film prepared by the plasma-oxidation process, in particular, its small thickness and its proximity to the underlying aluminum gate electrode, several of the above mentioned techniques are rendered impractical for characterization, and access to the thin-film properties of the AlO_x films is quite challenging. Nevertheless, the thin-film composition has been investigated using two different techniques : Electron Energy Loss Spectroscopy (EELS), in combination with Transmission Electron Microscopy (TEM) and Atom Probe Tomography (APT). The preliminary results from these two measurement techniques and the challenges faced during measurement are detailed in this chapter. Chapter 4 already discussed the effect of the plasma-process parameters on the electrical characteristics of organic TFTs. Certain combinations of the plasma power and the plasma duration were instrumental in forming AlO_x films with an optimum surface roughness ((R_{RMS}) \pm 0.5 nm), and hence organic TFTs with high charge-carrier mobilities. One of these combinations, i.e. a plasma power of 200 W and a plasma duration of **30** s has been selected to fabricate AlO_x films in this chapter and further characterize them in terms of the thickness of the film and its chemical composition. This plasma-parameter combination was also chosen to fabricate AlO_x films in the organic TFTs studied in the further chapters, and is thus most suitable to be considered

for further characterization here.

5.1 Thickness of plasma-grown AlO_x films

In order to operate organic thin-film transistors (TFTs) at low supply voltages, the overall thickness of the hybrid gate dielectric must be limited to a minimum value. The thickness of the hybrid gate dielectric is mainly controlled by the thickness of the oxide, since the thickness of the self-assembled monolayer (SAM) is limited to one molecular length of the particular molecule selected and does not have much scope for variation.



Figure 5.1: Thickness of plasma-grown AlO_x films as measured by transmission electron microscopy (TEM) in a multilayer Al/AlO_x structure. The AlO_x films were prepared by oxidizing the surface of aluminum films of two different nominal thicknesses, 30 nm and 100 nm

The plasma-oxidation process is extremely effective in forming an ultrathin AlO_x film, with the thickness of the film determined by the self-limiting nature of the process. In Chapter 4, the thickness of the AlO_x film was varied between approximately 4 nm and 7 nm using different combinations of the plasma-process parameters. In this section, AlO_x films have been fabricated with a plasma power of 200 W and a plasma duration of 30 s to measure the thickness of the films and examine the reproducibility of the plasma process. As shown in Fig.5.1, two different multilayer structures were fabricated by sequentially depositing aluminum films, and the surface of the aluminum was oxidized in an oxygen plasma. Aluminum films with nominal thicknesses of 30 nm and 100 nm were deposited to examine the influence of the thickness of the underlying aluminum film on the thickness of the subsequently formed AlO_x film. Chromium films with a nominal thickness of 100 nm were deposited before and after the Al/AlO_x multilayers to protect them from ion-milling damage while fabricating samples for transmission electron microscopy (TEM).

Fig.5.1(a) shows a schematic of the multilayer structure with nine aluminum thin films, each with a nominal thickness of 30 nm and the respective AlO_x films at the surface of each aluminum film. Fig.5.1(b) shows the TEM image of this Al/AlO_x multilayer structure, and the results are summarized in Fig.5.1(c), which were obtained from image analysis as explained in Chapter 4, Section 4.6. From a total of 40 positions, the average thickness of the AlO_x films measured is **6.5** \pm **0.5** nm. Similarly, Fig.5.1(d) shows the schematic of the multilayer structure with four aluminum films with a nominal thickness of 100 nm and the respective AlO_x films, and the TEM image in Fig.5.1(e). The average thickness of the AlO_x films, when measured in 20 positions, is **6.4** \pm **0.4** nm, as summarized in Fig.5.1(f).

Considering both multilayer structures, the thickness of the AlO_x films has a narrow deviation from the average value, which confirms the reproducible and reliable nature of the oxygen-plasma process to fabricate ultrathin AlO_x films with a particular thickness. By observing the effect of the plasma process on aluminum films of distinctly different thicknesses, it can be concluded that the thickness of the plasma-grown AlO_x films can be varied only by varying the plasma-process parameters, and the thickness of the underlying aluminum film has virtually no influence on it.

The gate-dielectric capacitance of a metal-insulator-metal structure with an AlO_x film prepared using the plasma-parameter combination 200 W, 30 s is approximately 1.1 μ F/cm², as shown in Fig.4.3(a). According to the relationship between the thickness of the gate-dielectric film and the gate-dielectric capacitance given by Eqn.2.9, and the relative permittivity of the plasma-oxidized AlO_x film determined in Chapter 4 (8 ± 0.2), the theoretical value of the thickness of the AlO_x film from the measured capacitance is calculated to be 6.4 nm, which is in good agreement with the measured thickness by TEM.

5.2 Composition of plasma-grown AlO_x films

As already mentioned in Chapter 3, several metal oxides are popular choices for gate dielectric materials in organic TFTs. The thin-film characterization of metal oxide films is important for determining their suitability as a gate dielectric and correlating the film properties of the metal oxide to the electrical properties of organic TFTs. One of the most important thin-film properties is the chemical composition, particularly for metal oxide gate dielectric films. A metal oxide with a perfect stoichiometric chemical composition is best suited as a gate dielectric material, while any disparity within the composition may affect its dielectric properties. For example, if a non-stoichiometric metal oxide film has a disproportionately large metal contribution, the film will display a stronger metallic behavior and be less effective as a gate dielectric in the transistor.

Among the various thin-film deposition and fabrication techniques, the group of chemical vapor deposition methods rely on a chemical reaction with precursors of the desired material to react with the surface of the substrate to fabricate a thin film in a controlled and self-limiting manner. Due to the chemical nature of the process, the resultant thin film has an almost stoichiometric composition, i.e. aluminum oxide films fabricated by ALD have a oxygen-aluminum ratio between 1.5 and 2 [223]. Similarly, in physical vapor deposition techniques such as sputtering and thermal evaporation, the thin film is deposited by condensation of a gas phase from a target material and the composition of the final thin film can be controlled by the composition of the target material. In such processes, a perfect stoichiometric Al_2O_3 film can be deposited to form the gate dielectric in transistors.

In contrast, the plasma-oxidation process does not afford a direct control of the thinfilm composition of the fabricated AlO_x film. The fabrication of the AlO_x film proceeds through oxygen ions and electrons in the oxygen plasma impinging on and reacting with the underlying aluminum surface. It has already been established in Chapter 3 that the AlO_x film prepared by the oxygen-plasma process serves as an excellent gate dielectric for low-voltage organic TFTs. The investigation of the composition of the AlO_x film is thus required to correlate it with the electrical performance of the organic TFTs, and possibly, further improve the gate-dielectric properties of the AlO_x films.

However, as it will be explained in the following sections, the characterization of the plasma-grown AlO_x films is not as straightforward as standard thin-film composition investigations. The plasma-oxidized AlO_x film has two relevant aspects which influence

the thin-film characterization, viz. the thickness of the film is relatively very small, approximately 6.5 nm, and the AlO_x film is in close proximity with the underlying aluminum film.

Certain photoelectron spectroscopy techniques are traditionally used to characterize thin films for elemental composition, such as X-ray Photoelectron Spectroscopy (XPS) or Time-of-Flight Secondary Ion Mass Spectrometry (ToF-SIMS). XPS is a surfacesensitive technique which uses X-rays to probe the surface of interest and the kinetic energy of the emitted electrons can be used to identify and quantify the elemental composition and electronic state of the atoms in the material. The depth of analysis of XPS is approximately 10 nm from the surface, which exceeds the typical thickness of the plasma-grown AlO_x films. Since the AlO_x films are always in close contact with the underlying aluminum films, the analysis depth of XPS extends into the aluminum, and a strong background signal makes it difficult to isolate the characterization of the AlO_x film. On the other hand, ToF-SIMS uses a finely focused ion beam to excite the surface material to emit secondary ions and ion clusters, whose exact mass and time-offlight is instrumental in identifying the elements or molecular fragments in the material. However, the depth of analysis in this technique is approximately 1 nm, which is much smaller than the typical thickness of the AlO_x film, and an ion milling technique will be required in conjunction to probe the entire depth of the AlO_x film. In addition, there are also problems with the exact quantification of elements, which will hinder the possibility of finding the exact composition of the AlO_x film using ToF-SIMS.

For these reasons, techniques with a higher spatial resolution, closer to an atomic scale, are required to characterize the plasma-grown AlO_x films. The two techniques chosen in this work are (1) Electron Energy Loss Spectroscopy (EELS), in combination with Transmission Electron Microscopy (TEM) and (2) Atom Probe Tomography (APT).

5.3 Qualitative measurement by TEM-EELS

Electron Energy Loss Spectroscopy (EELS) is a high-resolution analytical technique implemented along with High-Resolution TEM in which the transmitted beam of electrons through the material is analyzed for the energy loss of electrons that occurs due to inelastic scattering during transmission. The amount of energy loss, as measured by a spectrometer, can be interpreted and assigned to the different causes of the energy loss, such as inelastic interactions, phonon excitations, inter and intra-band transitions, plasmon excitations, inner shell ionizations, etc. to spatially map the elemental composition from the same specimen region, as observed by TEM. Fig.5.2 shows a schematic representation of the process, in conjunction with a TEM apparatus.



Figure 5.2: Schematic representation of the different electron beam interactions in a transmission electron microscope and the inelastically scattered electrons which are relevant for Electron Energy Loss Spectroscopy (EELS)

The data collected from EELS can be depicted in two different ways, an energy loss spectrum and TEM images with contrast which convey spectral information in a spatially resolved manner, also referred to as energy-filtered TEM images. These images are useful in mapping the elemental distribution in a sample and extracting other material properties. The energy-loss spectrum typically has three different regions:

- Zero-loss peak: This part corresponds to electrons which have not lost any energy, i.e. electrons which haven't been scattered and electrons which have been elastically scattered after interaction with the atomic nuclei.
- Low-loss region: This part corresponds to energy losses less than 50 eV and represents beam electrons which have interacted with the weakly bound electrons in the sample. This part of the spectrum typically gives information about the band structure and the dielectric properties of the material.
- High-loss or core-loss region: This part of the spectrum is reserved for higher energy losses, greater than 50 eV where the beam electrons have interacted with the tightly bound core electrons of the atoms in the sample, and the overall number

of inelastically scattered electrons is much lower. This region in the spectrum contains characteristic features called 'ionization edges', which have element-specific characteristics.

5.3.1 Experimental

A multilayer sample structure (similar to the one shown in Fig.5.1(a)) was chosen for analysis by EELS, i.e. a multilayer structure with aluminum films sequentially deposited and the surface oxidized in an oxygen plasma to form the respective AlO_x films. A silicon substrate with a thermally grown, 100 nm thick SiO_2 film, was chosen as the substrate on which the Al/AlO_x multilayer structure was fabricated.¹

5.3.2 Results

A typical EELS spectrum consists of the signal intensity plotted as a function of the energy loss of the transmitted electrons. As mentioned above, the region of the spectrum with an energy loss greater than 50 eV is called the 'high-loss region' and appears as a smoothly falling *background* with characteristic features superimposed on it, known as **ionization edges**. Ionization edges are formed when an inner shell electron in the sample material absorbs enough energy from a beam electron to be excited into a state above the Fermi level energy, and have element-specific energies. Fig.5.3 shows the ionization edges found in the EELS spectrum obtained for the sample and the three specific elements identified: silicon, aluminum and oxygen. The specific elements were identified based on their K-edge ionization energies, as they appear in the spectrum: silicon at 1839 eV, aluminum at 1560 eV and oxygen at 532 eV. Based on the intensity of the signal obtained, the element specific characteristics can be spatially mapped in the TEM image, to identify which element is located at which position in the sample. Inset images in each of the spectra show energy-filtered TEM images with silicon, aluminum and oxygen mapped, based on the respective ionization energies. These three images are composed into an RGB image with the following color code: blue for silicon, red for aluminum, and green for oxygen, as seen in Fig.5.3(d). Based on this image, the silicon substrate, the SiO_2 layer and the alternating layers of aluminum and

¹The preparation of the substrates using a Focused Ion Beam apparatus and the measurement using TEM-EELS were performed by Julia Deuschle, Yi Wang and Robin Lingstädt from the Max Planck Institute for Solid State Research.



89

Figure 5.3: High-loss region of the EELS spectrum with element specific ionization edges for (a) silicon, (b) aluminum and (c) oxygen with energy-filtered TEM images in the insets. (d) Composite RGB image depicting the spatially mapped elements in the TEM image

aluminum oxide can be clearly identified in a spatially mapped manner.

Apart from the high-energy-loss region of the spectrum, the low-loss region can also be analyzed for information, as shown in Fig.5.4. Fig.5.4(a) shows the low-loss region of the spectrum along with the zero-loss peak. The TEM image in the inset shows the position on the sample where the spectrum was collected, which was focused on one of the aluminum films. The zero-loss peak represents the beam electrons which haven't lost any energy and is always the tallest and most prominent feature of an EELS spectrum. The shape and size of the zero-loss peak is an indication of the energy resolution of the spectrum. In any measured spectrum, the zero-loss peak is spread over energies around 0 V and has a finite energy width due to the energy resolution of the spectrometer as well as the energy spread of the beam electrons, along with thermal diffusion scattering which occurs at an energy loss equal to approximately 10 meV. Because of these reasons, it is difficult to distinguish between electrons which have lost no energy from those electrons which have lost a small amount of energy. In Fig.5.4(a), the zero-loss peak has a spread of 1 eV around the 0 V mark, which is a fairly narrow energy width. This means that the probability of resolving finer details of the EELS spectrum is much higher, as compared to a zero-loss peak with a broader energy width.



Figure 5.4: (a) Low-loss region of the EELS spectrum with the zero-loss peak and the position of measurement in the sample, (b) composite image with the spatially mapped elements in the TEM image and element specific peaks for (c) aluminum and (d) oxygen with energy filtered TEM images in the respective insets

The features in the energy loss spectrum in the low-loss region result from the excitation of the valence or conduction band electrons. Similar to the high-energy ionization edges, the low-loss peaks in the spectrum are also element-specific and can be used to spatially map the elements present in the sample. Fig.5.4(c) shows the characteristic low-loss peak for the element aluminum obtained from one of the aluminum layers in the sample, at an energy loss equal to 30 eV. Similarly, Fig.5.4(d) shows the characteristic peak for oxygen at an energy loss of 23 eV, which was obtained from one of the AlO_x layers in the sample. From the entire spectrum, a color coded Red-Green image was constructed, as shown in Fig.5.4(b), which represents the alternating Al/AlO_x layers.

The low-loss peak for oxygen is slightly broader in comparison to the peak for aluminum, and the broad energy width of the oxygen peak restricts the energy resolution of the EELS spectrum. Fig.5.5(c) shows a characteristic X-ray spectrum obtained in conjunction with the EELS measurement. When core electrons in the sample material are ejected by beam electrons, the holes created are filled with energetically higher outershell electrons, emitting characteristic X-rays in the process. Fig.5.5(a) shows the region of the sample in which a line-scan was measured, with alternating Al/AlO_x multilayers and Fig.5.5(b) shows the spatial position of the line scan vs the X-ray energy detected. The characteristic energy for elemental aluminum can be observed for the entire line scan at an energy of 1.5 keV as aluminum is present in both the aluminum and the AlO_x films. In the AlO_x films, traces of oxygen can be observed at an X-ray energy of 0.5 keV.



Figure 5.5: X-ray spectrum obtained in conjunction with the EELS measurement. (a) Position of line scan in the sample (b) Spatial position of the line scan vs energy of X-rays detected and (c) characteristic X-ray spectrum showing presence of elemental aluminum, oxygen and carbon in the sample

Apart from detecting the presence of elemental aluminum and oxygen as expected in the X-ray spectrum, traces of elemental carbon are also detected at an energy of approximately 0.3 keV. This suggests the presence of contamination during the EELS measurement which limits the scope of the measurement. The carbon contamination limits the exposure time during the measurement, and therefore the necessary spectral quality and resolution is below optimum.



Figure 5.6: Multiple EEL spectra collected in the AlO_x film with (a) the positions of measurement and (b) the individual spectra averaged and smoothened using principle-component analysis

The effect of this contamination can be observed in Fig.5.6 in which individual spectra were collected from different positions located in one of the AlO_x layers. The positions were 1.5 nm apart and the spectra were averaged and smoothened using principle-component analysis. The objective behind collecting these spectra was to resolve the fine structure of the spectral peaks representing oxygen at 532 nm to investigate its electronic state in this part of the sample. The electronic state of oxygen would ideally convey information about the composition of AlO_x . However, as can be seen in the spectra, no finer details can be resolved due to insufficient exposure times and the electronic state of oxygen in the AlO_x film cannot be determined.

Since the electronic state of oxygen could not be resolved further from the EEL spectrum due to insufficient exposure, the oxidation state of aluminum in the AlO_x film was investigated, in an attempt to determine the composition of the plasma-oxidized aluminum oxide film.



Figure 5.7: (a) Position in the sample and (b) position in the AlO_x layers where the EELS measurement was conducted. (c) EELS spectrum measured and (d) comparison of measured spectrum with reference spectra for elemental aluminum (in green) and Al_2O_3 (in red)

Fig.5.7(a) shows the general region of the sample considered for the EELS measurement, and Fig.5.7(b) shows the 4 positions in the AlO_x films in which the EEL spectra were collected. The spectra were acquired and investigated in the range of aluminum $L_{2,3}$ ionization edges around an energy loss of 90 eV, as shown in Fig.5.7(c).

Although the sample was freshly prepared in the nanomill before mounting in the TEM, the presence of contamination limited the resolution of the spectra severely and the oxidation states of aluminum could not be resolved. However, an attempt was made to qualitatively assess the oxidation state of aluminum by comparing the acquired spectrum with reference spectra. Fig.5.7(d) shows three different spectra, the acquired one (in cyan), a reference spectrum for polycrystalline aluminum (in green), and a reference spectrum for Al_2O_3 (in red). By comparing these three spectra, it can be concluded that the acquired spectra bears more resemblance with the spectrum for Al_2O_3 than with the one for elemental aluminum. Thus, it can be induced that on a qualitative basis, the oxidation state of Al in the AlO_x is much closer to Al_2O_3 than to metallic aluminum.

5.3.3 Outlook

The analytical technique Electron Energy Loss Spectroscopy was employed in an attempt to identify the composition of the plasma-oxidized AlO_x films. Although the technique has a promising potential to analyze the chemical composition of thin films, on a resolution similar to that of TEM, it could not be implemented conclusively for plasma-oxidized AlO_x films. The main hurdle in this measurement and its analysis was the contamination of the sample of interest by carbon. Although extreme care was taken while preparing the TEM sample in a Focused Ion Beam (FIB) apparatus, and the measurement was carried out almost immediately after sample preparation, the level of contamination was significant enough to prevent long exposure times and limit the resolution required to resolve the electron energy loss spectra. One major reason for this is likely the high reactivity of aluminum, which might have been contaminated by hydrocarbons upon exposure to air during sample transfer from the FIB to the TEM. In order to overcome these problems of contamination, possible solutions include sample transfer in vacuum and cryogenic cooling during measurement. However, the implementation of these solutions will require more advanced instrumentation.



5.4 Quantitative measurement by Atom Probe Tomography

Figure 5.8: Schematic representation of the setup for an atom probe tomography measurement with the main components involved and a representation of the basic measurement principle

Atom Probe Tomography (APT) is a microscopy technique with single-particle detection capability, and extremely effective in investigating several phenomena in materials science [224–227]. A schematic representation of the basic components of the measurement setup and the operating principle is shown in Fig.5.8. Individual atoms are fieldevaporated from a specimen through layer-by-layer erosion and collected at a positionsensitive detector. Field evaporation is the process of removing atoms from the surface they are bonded to by overcoming local bonding forces. The original position of the surface atoms at the sample surface is later derived from the impact coordinates of the evaporated species onto the detector using a point-projection law. The chemical identity of the evaporated species is determined by time-of-flight mass spectrometry. After the measurement, a digital reconstruction of the original position and chemical identity of individual atoms is used to analyze various material properties on an atomic scale.

A high electric field is required for the field evaporation of atoms, to overcome the local bonding forces at the surface. This is achieved by applying a very high electric voltage between the specimen and a counter electrode. The specimen is shaped in the form of a very sharp needle. Typically, the tip radius of the specimen is approximately 50 nm and the high curvature of the tip creates the necessary large electric field. The fieldevaporated ions close to the tip surface have a diverging projection during flight which contributes to the extremely high magnification and spatial resolution of the instrument. For the chemical identification of the atoms, time-of-flight mass spectrometry requires a pulsed-field evaporation. This is achieved by either thermal pulsing (applying a pulsed laser) or by field pulsing (applying a pulsed voltage). Additionally, the sample tip is also cooled to cryogenic temperatures (20-150 K) because of which the evaporation process is mostly thermally activated.

Atom probe tomography is an extremely useful technique for the chemical analysis of alloys, studying solid-state reactions [228, 229], characterization of complex nanostructures [230–232] and even studying biomaterials and soft matter [233,234]. This technique has been chosen to study the properties of the plasma-oxidized AlO_x films with the expectation that its atomic resolution and three-dimensional volume analysis will prove useful in overcoming the shortcomings of the other characterization techniques and give insights into the Al/AlO_x layers for organic transistors.

5.4.1 Experimental

As mentioned above, the material of interest must be shaped into a very sharp needle geometry with a quasi-hemispherical end shape, in which a small radius of curvature is required to produce the mandatory high electric field at the surface. As a preliminary attempt, thin aluminum wires were electropolished and oxidized in the oxygen plasma to obtain the particular Al/AlO_x layers in a sharp tip-shaped geometry. However, this method was not successful in forming the desired tip geometry and the AlO_x film was exposed to air, causing contamination. Therefore, a more advanced lift-out technique was employed using a Focused Ion Beam (FIB) apparatus.

The various steps in this process as well as the corresponding SEM images are shown in Fig.5.9. The bulk specimen chosen for the lift-out technique was a multilayer Al/AlO_x structure, similar to the one shown in Fig.5.1(a). The first step in the process is to deposit a rectangular block of platinum on the surface of the bulk specimen to isolate the feature of interest from the sample.

As a next step, two rectangular regions with increasing depth towards the platinum

block are removed by ion milling with the gallium ion beam. By milling these parts away, the region of interest, hereafter referred to as the *lamella*, can be successfully isolated for further processing. Thereafter, the ion beam is used to remove material from two positions in the sample: one on the side of the platinum block, in which the incision is deep enough to reach the material of interest, and the other is positioned at the bottom of the isolated lamella. Essentially, the lamella is then connected to the rest of the bulk specimen only on one side, analogous to a cantilever beam. A micromanipulator is a device used in FIB systems which supports a very sharp needle or a gripping device with a high level of precision of movement. Once the lamella is separated from the specimen from two sides, a micromanipulator is attached to the free end using a temporary welding process by platinum deposition. After the micromanipulator is firmly attached, the lamella is milled from the other end, essentially freeing it from the bulk specimen. Using the micromanipulator movement, the lamella is pulled away from the bulk specimen, lifted out, and transferred to a supporting structure, which in this case, is a tungsten stub. The lamella is attached to the tungsten stub using a small amount of platinum deposition, and a part of the lamella is milled so that the remaining part can be utilized to make more sample tips.²

 $^{^{2}}$ The preparation of the sample tips using the Focused Ion Beam apparatus and expert technical assistance during the atom probe measurement were provided by Helena Solodenko from the University of Stuttgart.



Figure 5.9: Schematic representation and corresponding SEM images of the various stages of fabrication of tips, i.e needle-shaped samples from a bulk specimen for the atom probe measurement using a focused ion beam

The relative orientation of the lamella with respect to the tungsten stub is crucial for the geometry of the final sample tip, which will be measured in the atom probe instrument. In this case, the lamella was attached with its horizontal axis parallel to the tungsten stub. Due to this chosen orientation, the Al/AlO_x multilayers are aligned vertically on the final tip. At this stage, the sample of interest has been attached to the sample tip, but the tip must be further processed to produce a sharp needle-shaped geometry for measurement. This is achieved by milling the tip in concentric circles with reducing radii, as shown in Fig.5.10. At the end of this entire process, the sample is ready in a sharp tip with a radius of approximately 50 nm.



Figure 5.10: Successive stages of ion milling to sharpen the tip radius depicted through SEM images in the focused ion beam with a final tip radius of approximately 50 nm

5.4.2 Results

During the atom probe measurement, atoms at the surface of the tip are field evaporated and collected at the position-sensitive detector where their chemical identity is characterized by time-of-flight mass spectrometry. This data is recorded in the form of a mass spectrum, shown in Fig.5.11. The mass spectrum shows the species identified during the measurement, which in this case are atomic aluminum, chromium, oxygen and fragments of the aluminum oxide, AlO, Al₂O, and Al₂O₂ in different ionization states. It is important to note here that the formation of these ionized species is a result of the breaking of physical bonds in the specimen, and these ionized species are not necessarily representative of stoichiometric compounds. With the chemical identity of the species known, and using information from the position-sensitive detector, a three-dimensional image of the original tip geometry can be reconstructed to analyze the material properties further.



Figure 5.11: Mass spectrum depicting the different ionized fragments identified during the atom probe measurement

Fig.5.12(a) shows **Reconstructed View A**, a cylindrical portion of the sample tip. This reconstructed view corresponds to a cross-section of two aluminum layers and one AlO_x layer in between, with its estimated position in the bulk specimen depicted in Fig.5.12(b). A cylindrical profile along this reconstructed view is chosen to analyze the composition along this cylinder, shown in Fig.5.12(d).



Figure 5.12: Reconstructed View A (a) Cylindrical reconstructed view depicting $Al/AlO_x/Al$ layers (b) Estimated location of the cylinder in the bulk specimen (c) Numerical value of x in AlO_x along the cylindrical reconstruction as derived from (d) the composition profile as extracted from the reconstructed view

It can be seen from the reconstructed view as well as the composition profile that the oxide layer is situated between two aluminum layers. In the aluminum oxide region of the reconstructed view, fragments of Al, O and AlO, Al₂O, Al₂O₂ can be observed. Analyzing the respective contributions of aluminum and oxygen, the value of x, in AlO_x can be plotted along the composition profile, shown in Fig.5.12(c). As expected, in the aluminum portion of the reconstructed view, x is almost equal to 0, as in the aluminum layer, the content of oxygen is negligible. In the AlO_x layer, the value of x is calculated to equal approximately 0.8, i.e. there are 8 parts of oxygen to every 10 parts of aluminum in the AlO_x film.

Fig.5.13 shows a second **Reconstructed View B** from the same measurement on the same tip specimen. This particular reconstruction, which shows three layers, corresponds to the AlO_x layer which was fabricated last, below the top chromium layer in the

multilayer bulk specimen, with its location in the bulk specimen shown in Fig.5.13(b).



Reconstructed View B

Figure 5.13: Reconstructed View B (a) Cylindrical reconstructed view depicting $Al/AlO_x/Cr$ layers (b) Estimated location of the cylinder in the bulk specimen (c) Numerical value of x in AlO_x along the cylindrical reconstruction as derived from (d) the composition profile as extracted from the reconstructed view

The advantage of this particular reconstruction is that because of the asymmetrical nature of the multilayer structure, it is evident that the AlO_x layer measured was formed by oxidizing the aluminum layer in the reconstructed view. Fig.5.13(d) shows the relative compositions of the fragments measured along a cylindrical profile in the reconstructed view. The three layers identified are aluminum, aluminum oxide and chromium, as evident by the almost unity atomic fractions of aluminum and chromium respectively. Similar to the first reconstructed view, the relative contribution of oxygen to aluminum was analyzed to determine the value of x along the composition profile. In this reconstructed view, the value of x is similar, approximately 0.65, i.e 6.5 parts of oxygen for every 10 parts of aluminum. In both reconstructions, although it was possible to reconstruct the aluminum and chromium layers with greater accuracy, the composition of AlO_x does not correlate with the expected value of 1.5, and seems much lower than that. Particularly, the oxygen content in both cases is significantly lower than the expected stoichiometric value, i.e. 15 parts of oxygen to 10 parts of aluminum, and the detected composition of the AlO_x films suggests a more metallic character.

5.4.3 Challenges

There were several challenges faced during the atom probe measurement which compromise the reliability of the results obtained. There were multiple instances of the sample tip breaking during the measurement when transitioning from the aluminum layer to the aluminum oxide layer. This is possibly because of different measurement conditions required for the two different materials. It was not possible to adjust the combination of cryogenic cooling and local spatial heating (caused by the pulsed laser) for the aluminum and aluminum oxide films simultaneously. Moreover, due to the small thickness of the AlO_x film, the sample tip did not have sufficient buffer time to adjust the measurement settings, and brittle fracture of the oxide layer caused the tip to break. In the rare occasion where the temperature of the tip was kept relatively high to prevent the brittle fracture of the tip, the measurement detailed above has been carried out. As can be seen in the mass spectrum in Fig.5.11, the signal-to-noise ratio is severely compromised, and the mass peak for aluminum has a strong tailing feature. Secondly, there is a strong suspicion that a majority of the oxygen in the AlO_x film might have escaped the system as O_2 gas, without being ionized, and hence would not feature in the mass spectrum. This could be one of the primary reasons why the measured composition value appears to be oxygen-deficient and does not reflect an accurate composition for the plasma-oxidized AlO_x film. A better control of the laser power through a systematic variation and the applied voltage is required to ensure that oxygen content in the AlO_x film is accurately measured.

5.5 Summary and Outlook

Ultrathin plasma-oxidized films of aluminum oxide are excellent gate dielectric films in hybrid gate dielectrics for organic thin-film transistors. However, due to the nature of the plasma process, the film properties of the AlO_x films need further characterization

for better control of the film properties. The most important properties from the gate dielectric perspective are the film thickness and the film composition, and an effort has been made to examine the properties of the AlO_x films. For a plasma-process-parameter combination of 200 W and 30 s, the thickness of the AlO_x films was measured to be 6.5 nm using transmission electron microscopy. The thickness of the films was found to be relatively reproducible and consistent and was independent of the thickness of the underlying aluminum films.

An attempt was made to measure the composition of the AlO_x films by two different characterization techniques: Electron Energy Loss Spectroscopy (EELS) (in conjunction with TEM) and Atom Probe Tomography. Through EELS, the respective elements aluminum and oxygen were spatially mapped to reveal successive Al/AlO_x films as fabricated. By comparing the oxidation states of aluminum in the AlO_x film, it was qualitatively concluded that the composition of the plasma-oxidized AlO_x films is closer to Al_2O_3 . The quantitative measurement of the AlO_x was unfortunately not successful, due to a significant amount of carbon contamination which limited the exposure time and hence the resolution of the EEL spectra.

Through atom probe tomography, the different Al/AlO_x layers were measured with atomic resolution and reconstructed on an atomic scale. Although this technique shows great potential for determining the composition of the AlO_x films, further tuning of the measurement conditions would be required to find a more accurate measurement of the composition. A calibration measurement with standardized sapphire samples (composition of Al_2O_3) and a variation of the laser power to measure its effect on the amount of gaseous oxygen ionized will improve the ability of this technique to accurately measure the composition of the AlO_x films.

A successful measurement of the composition of the plasma-oxidized AlO_x films would enable a better understanding of the properties of the AlO_x films as gate dielectrics and also opportunities for improving the electrical properties of the organic transistors further by improving the material properties of the AlO_x gate dielectric films.

Chapter 6

Influence of self-assembled monolayer thickness in hybrid gate dielectrics: Alkyl and fluoroalkyl phosphonic acids

The second component of the hybrid gate dielectrics investigated in this thesis is a molecular self-assembled monolayer (SAM) formed by phosphonic acid molecules, assembled from solution on the plasma-grown aluminum oxide (AlO_x) surface in organic thin-film transistors (TFTs). The role of the SAM in the hybrid gate dielectric as well as in the overall functioning of organic TFTs has been well established in numerous previous studies and summarized in Chapter 3. The primary function of SAMs is to enable low-voltage operation of organic TFTs by reducing the gate-dielectric thickness, while simultaneously maintaining a low leakage current density [35]. In addition, the possibility of employing different SAMs to modify the threshold voltage [157, 235, 236], surface properties, organic-semiconductor morphology and overall organic-TFT characteristics has also been widely explored.

The gate dielectric-semiconductor interface is extremely important for the electrical characteristics of the organic TFT, since the conducting channel is situated in close proximity to this interface. A well-established strategy to obtain a suitable organicsemiconductor morphology in bottom-gate transistor configurations, is to modify the surface of the gate dielectric onto which the organic semiconductor is deposited. Several different types of SAMs, as well as polymer gate dielectrics [237], have been used for this purpose. However, a *suitable* morphology of the organic semiconductor is challenging to define narrowly and may depend on the semiconductor/gate dielectric combination in question. In general, a suitable semiconductor morphology promotes a larger area of percolation for the charge carriers and a lower density of charge-carrier traps. Often, the semiconductor morphology is characterized by the grain size of a polycrystalline organic-semiconductor layer at its complete thickness of several monolayers. A straightforward understanding of the semiconductor morphology prefers a larger grain size so that a lower density of grain boundaries implies a lower density of charge-carrier traps. This correlation between the grain size of the semiconductor layer and the electrical performance of the organic TFT has been established in several reports [118]. Nevertheless, this does not completely explain the influence of the semiconductor morphology as there are also several contradictory reports, in which a larger grain size has resulted in a decreased effective charge-carrier mobility [169]. To gain a better understanding of the semiconductor morphology, the growth mode of the semiconductor must also be considered in addition to the grain size. The growth mode of the organic semiconductor influences the percolation path for the charge carriers, especially in the first few layers close to the gate dielectric, i.e. in the conducting channel and gives a deeper insight into the influence of the organic-semiconductor morphology on the electrical characteristics of organic TFTs.

An important parameter regarding the surface properties of the gate dielectric, especially in bottom-gate organic TFTs, is the surface energy. Chapter 4 discussed the importance of the surface roughness and the manner in which the gate oxide and the plasma process to fabricate the gate oxide, are instrumental in tuning the surface roughness of the gate dielectric. It was also seen in Chapter 4 that the plasma process has no effect on the surface energy of the gate dielectric, and the surface energy of the bare AlO_x surface was changed only upon formation of the SAM. Several previous reports have investigated the effect of changing the surface energy of the gate dielectric on the wettability of the gate-dielectric surface and thereby the growth mode and the morphology of the semiconductor layer, and subsequently the performance of the organic TFT [118, 238]. Among SAMs, different strategies have been used to modify the surface energy, such as employing different functional groups [155, 175], inserting polar atoms in the alkyl chain [239], using hybrid multilayers [240, 241], etc. With the choice of AlO_x as the inorganic gate oxide component of the hybrid gate dielectric, the optimum choice for the SAM is a molecule with a phosphonic acid head group to assemble on the AlO_x surface [35, 242].



Figure 6.1: Alkyl and fluoroalkyl phosphonic acid molecules of different chain lengths selected to form SAMs in hybrid gate dielectrics

This chapter explores two different types of phosphonic acid molecules, viz. alkyl phosphonic acids (HC_xPA) and fluoroalkyl phosphonic acids (FC_xPA), where x is the number of carbon atoms in the aliphatic chain of the phosphonic acid molecule. The chemical structures of the phosphonic acid molecules of different chain lengths are given in Fig.6.1. By employing molecules with a different number of carbon atoms in the aliphatic backbone of the molecule, the thickness of the SAM is systematically changed and the influence of the changing thickness of the gate dielectric on the electrical and surface properties of hybrid gate dielectrics in organic TFTs is investigated. Furthermore, the nucleation and growth of the organic semiconductor DNTT on four different surfaces was recorded to observe changes in the growth mode of the semiconductor. Along with the growth mode, the semiconductor morphology for two different semiconductors (DNTT and DPh-BTBT) of different thicknesses on different SAMs has also been investigated. This chapter aims to understand the effect of the alkyl chain length of phosphonic acid molecules on the formation and properties of the SAM, as reflected in its surface properties, and a resulting influence on the growth and morphology of organic semiconductors and the electrical properties of organic TFTs.

6.1 Background

The influence of the chain length of phosphonic acid molecules used in SAMs for hybrid gate dielectrics towards low-voltage organic TFTs has been explored in several other literature accounts, as summarized in Table 6.1. Acton and co-workers used alkyl phosphonic acid molecules with chain lengths from six to eighteen carbon atoms in combination with a gate oxide consisting of SiO_2 with a thickness of 2.5 nm and a high-k oxide HfO_2 with a thickness of 8 nm [120]. By employing ATR-FTIR (Attenuated Total Reflectance-Fourier Transform Infrared Spectroscopy) and studying the morphology of the organic semiconductor pentacene deposited onto the different SAM surfaces, they concluded that the ordering in the phosphonic acid SAMs increases steadily with an increase in chain length. The largest grain size for pentacene was observed for the medium-chain-length molecules from HC_8PA to $HC_{14}PA$, and was correlated to the largest charge-carrier mobilities. Jedaa and co-workers investigated alkyl phosphonic acids in combination with plasma-grown AlO_x oxide films, fabricated with two different plasma processes, to study the effect of the underlying AlO_x film on the resulting SAM formation [191]. They concluded that for longer chain lengths of the phosphonic acid molecule, the SAM compensates for a lower quality of the AlO_x thin film, and that the electrical properties of the organic TFTs at higher electrical fields are dominated by the SAM and not by the oxide. Fukuda and co-workers also investigated the different alkyl phosphonic acids in combination with a plasma-grown AlO_x thin film, and performed bias-stress measurements along with studying the semiconductor morphology of pentacene deposited onto the different SAMs [154]. The highest charge-carrier mobility was found for the TFT with the $HC_{14}PA$ SAM, and was attributed to its low surface roughness (as determined by X-ray reflectivity measurements), resulting in the largest grain size of pentacene on the HC₁₄PA SAM. Hill and co-workers used SiO₂ with a thickness of 100 nm as the gate oxide, which means that the TFTs had to be operated with high voltages, but characterized the SAMs using water contact-angle measurements and analyzed the subthreshold performance of the organic TFTs to study the behavior of different SAMs [243]. This study found an optimum organic-TFT performance for phosphonic acid molecules with eight and ten carbon atoms with a minimum trap density at the gate dielectric and thereby an improved subthreshold swing and threshold voltage. Hannah and co-workers used a novel vapor-deposition technique to form the phosphonic acid SAMs, instead of the usual solution-processing method, and also characterized the

Reference	Molecules investigated (HC _x PA)	Substrate	Gate oxide	Organic Semicon- ductor	Mobility optimum
[120] Acton et al.	$\begin{array}{c} x = 6, 8, 10, \\ 12, \ 14, \ 16, \\ 18 \end{array}$	Silicon	$\begin{array}{c} 2.5 \text{ nm } \operatorname{SiO}_2 \\ + 8 \text{ nm } \operatorname{HfO}_2 \\ \text{(sol-gel)} \end{array}$	Pentacene	$\begin{array}{c} \mathrm{HC_8PA-}\\ \mathrm{HC_{14}PA} \end{array}$
[191] Jedaa et al.	$ \begin{array}{c} x = 2, 6, 10, \\ 1, 18 \end{array} $	Glass	AlO _x , mild air and oxy- gen plasma	Pentacene	HC ₁₄ PA
[154] Fukuda et al.	$\begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$	Silicon	AlO_x , oxy- gen plasma	Pentacene	$HC_{14}PA$
[243] Hill et al.	$ \begin{array}{l} x = 6, 8, 10, \\ 12, \ 14, \ 16, \\ 18 \end{array} $	Silicon	100 nm SiO_2	Pentacene	HC ₈ PA - HC ₁₀ PA -
[244] Hannah et al.	x = 8, 10, 12, 14, 16, 18 (vacuum- processed)	Glass	$AlO_x, UV/$ ozone	DNTT	HC ₁₈ PA
This work	$\begin{array}{c} HC_xPA, & x \\ = \ 6, \ 8, \ 10, \\ 12, \ 14, \ 16, \\ 18 \end{array}$	flexible PEN	AlO _x , oxy- gen plasma	DNTT	HC ₁₄ PA
This work	$\begin{array}{c} FC_{y}PA, y = \\ 6, 8, 10, 12, \\ 14 \end{array}$	flexible PEN	AlO _x , oxy- gen plasma	DPh-BTBT	FC ₁₀ PA

 Table 6.1: Summary of literature reports investigating the influence of chain length in phosphonic acid molecules in hybrid gate dielectrics

SAMs formed using FTIR measurements and DFT (Density Functional Theory) calculations to analyze the packing density and ordering within the monolayers [244]. An increased molecular order within the SAM was observed with an increase in the chain length of the molecule, thereby improving the performance of the organic TFTs. Most of these studies found that an optimum charge-carrier mobility was obtained by using a medium chain-length phosphonic acid SAM (between ten and fourteen carbon atoms) and attributed it to the medium chain-length SAMs forming the most densely packed and well-ordered monolayers. This was then verified by several characterization techniques, either by spectroscopic measurements of the SAMs, or by observing the resulting organic-semiconductor morphology on these SAM surfaces.

This chapter explores the influence of the chain length of two different phosphonic acid molecules, in combination with two high performance organic semiconductors, DNTT and DPh-BTBT, which are more air-stable than pentacene. Organic TFTs have been fabricated on flexible PEN (polyethylene naphthalate) foils, in comparison to the rigid glass or silicon wafer substrates used in previous studies. For the first component of the hybrid gate dielectric, a consistent combination of 200 W and 30 s was selected to fabricate the plasma-grown AlO_x films with a thickness of approximately 6.5 nm. The choice of the particular phosphonic acid molecule in combination with the particular organic semiconductor is explained in Appendix D. The fluoroalkyl phosphonic acid SAMs were used in combination with the organic semiconductor DPh-BTBT because this combination yielded a turn-on voltage much closer to 0 V, which is beneficial from a technological point of view. Seven different alkyl phosphonic acid molecules (from six to eighteen carbon atoms) and five different fluoroalkyl phosphonic acid molecules (six to fourteen carbon atoms) were used to form SAMs. The fluoroalkyl phosphonic acid molecules were chosen in which the entire alkyl chain (except two carbon atoms) has been fluorinated. Partially fluorinated phosphonic acids with longer chains (e.g. eighteen carbon atoms) were not included, since these are not commercially available. The electrical properties of the organic TFTs with the different SAMs and the morphology of the semiconductors on the respective SAMs is investigated, and correlated to the surface properties of the different SAMs.

6.2 Dielectric properties of hybrid gate dielectrics with self-assembled monolayers

It has already been established in Chapter 4 that the overall gate-dielectric capacitance of the hybrid dielectric is dominated by the capacitance of the SAM due to the influence of its low dielectric constant. Any variation in the gate-dielectric capacitance of oxide dielectrics obtained by modifying the aluminum oxide layer is overpowered after introducing the SAM and results in an almost constant capacitance of the hybrid gate dielectrics. This means that for hybrid gate dielectrics with the same molecule forming the SAM, very little variation in the gate-dielectric capacitance is possible due to the large contribution of the SAM to the overall gate-dielectric capacitance. Hence, the possibility of using molecules with different chain lengths to form SAMs of different thicknesses has been explored to modify the overall gate-dielectric capacitance of hybrid gate dielectrics.



Figure 6.2: Measured gate-dielectric capacitance of (a) AlO_x/HC_xPA SAM and (b) AlO_x/FC_xPA SAM hybrid gate dielectrics as a function of the phosphonic acid molecule chain length (number of carbon atoms)

Fig.6.2 shows the variation of the measured gate-dielectric capacitance of metal-insulatormetal capacitors with the different chain-length phosphonic acid molecules, with Fig.6.2(a) showing the variation for alkyl phosphonic acid molecules from six carbon atoms to eighteen carbon atoms in the alkyl chain and Fig.6.2(b) for fluoroalkyl phosphonic acid molecules with six to fourteen carbon atoms in the aliphatic backbone. For the alkyl phosphonic acid SAMs, the gate-dielectric capacitance of hybrid AlO_x/SAM gate dielectrics can be varied between 0.6 and 1 μ F/cm² and for the fluoroalkyl phosphonic acids, the gate-dielectric capacitance varies between 0.6 and 1.1 μ F/cm². It can be observed that a longer alkyl chain in both phosphonic acid molecules, which signifies an increase in the thickness of the SAM (t_{SAM}), results in a steady decrease in the gate-dielectric capacitance C_{diel} . This is in accordance with the theoretical equation:

$$\frac{1}{C_{\text{diel}}} = \frac{t_{\text{ox}}}{\varepsilon_0 \varepsilon_{\text{ox}}} + \frac{t_{\text{SAM}}}{\varepsilon_0 \varepsilon_{\text{SAM}}}$$
(6.1)

where t_{ox} is the thickness of the AlO_x layer, ε_0 is the absolute permittivity, ε_{ox} is the permittivity of AlO_x, and ε_{SAM} is the relative permittivity of the SAM.

A similar effect of the chain length of the phosphonic acid molecules on the gate-dielectric capacitance was also observed by Acton et al. [120] and Hannah et al. [244], and the

gate-dielectric capacitance values were used to calculate the thickness of the SAMs for the different chain-length phosphonic acid molecules. The thickness of the SAM for alkyl phosphonic acid molecules from HC₆PA to HC₁₈PA was calculated to be between 0.8 nm and 2.4 nm. However, it must be noted that the thickness of the SAM does not correspond to the chain length of the molecule as the phosphonic acid molecules usually assemble with a tilt angle of 20°-30° with respect to the gate-dielectric surface. It can also be observed in Fig.6.2 that for the longest chain-length molecules, the gate-dielectric capacitance does not decrease monotonically, i.e. the gate-dielectric capacitance corresponding to SAMs with HC₁₆PA and HC₁₈PA is approximately 0.6 μ F/cm² and similarly for the fluoroalkyl FC₁₂PA and FC₁₄PA molecules. This indicates the possibility that even with an increase in the chain length of the molecule, the thickness of the SAM does not change proportionally, especially for the longer chain length molecules. This aspect will be revisited again while discussing the morphology and ordering of the SAMs in Section 6.4.2.

Thus, by modifying the chain length of the phosphonic acid molecules, the capacitance of the hybrid gate dielectric can be varied between 0.6 and 1 μ F/cm², which was not possible by varying the plasma parameters of the AlO_x fabrication process. In general, a large gate-dielectric capacitance is preferred for low operating voltages in the organic TFT, thus making smaller chain length phosphonic acid molecules more desirable. However, the effect of the different molecules on the electrical characteristics of organic TFTs and the surface properties of the gate dielectric must be considered before selecting an optimum range of chain lengths of the phosphonic acid molecules.

6.3 Electrical properties of organic TFTs with self-assembled monolayer hybrid gate dielectrics

The importance of the SAM in hybrid gate dielectrics in organic TFTs is the central theme of this chapter. Hence, the different phosphonic acid molecules were employed in hybrid gate dielectrics in organic TFTs to observe the influence of the chain length of the molecules on TFT performance. Throughout this section, the effect of AlO_x/HC_xPA SAM hybrid gate dielectrics on DNTT TFTs and AlO_x/FC_xPA SAM hybrid gate dielectrics on DNTT TFTs and AlO_x/FC_xPA SAM hybrid gate dielectrics on DNTT TFTs will be considered in parallel.



Figure 6.3: Transfer characteristics of (a) DNTT TFTs with AlO_x/HC_xPA hybrid gate dielectrics and (b) DPh-BTBT TFTs with AlO_x/FC_xPA hybrid gate dielectrics for all the different phosphonic acid molecules forming the SAMs. The TFTs have a channel length of 20 μ m and a channel width of 100 μ m

The measured transfer characteristics of organic TFTs (the drain current I_D as a function of the gate-source voltage V_{GS}) with the phosphonic acid molecules of different chain lengths superimposed can be seen in Fig.6.3(a) for DNTT TFTs with alkyl phosphonic acid SAMs and in Fig.6.3(b) for DPh-BTBT TFTs with fluoroalkyl phosphonic acid SAMs. The different phosphonic acid molecules considered are all suitable to form hybrid gate dielectrics in functional organic TFTs. Among the alkyl phosphonic acid SAMs, there is a slight variation for the shorter chain length molecules (HC₆PA and HC₁₀PA), but mostly all molecules lead to similar transfer characteristics of the TFTs with the alkyl and fluoroalkyl phosphonic acid SAMs. To observe a more detailed effect of the chain length, individual TFT-performance characteristics obtained from the transfer curves as a function of the chain length of the phosphonic acid molecules are compared below.



Figure 6.4: Threshold voltage of (a) DNTT TFTs with AlO_x/HC_xPA hybrid gate dielectrics and (b) DPh-BTBT TFTs with AlO_x/FC_xPA hybrid gate dielectrics as a function of the phosphonic acid molecule chain length (number of carbon atoms)

The threshold voltage of organic TFTs is defined as the minimum gate-source voltage at which charge carriers are accumulated in the conducting channel and gives an insight into the density of charge-carrier traps in the gate dielectric and at the dielectricsemiconductor interface. The threshold voltage is plotted as a function of the chain length of the respective phosphonic acid SAMs in the DNTT and DPh-BTBT TFTs in Fig.6.4. For the DNTT TFTs with the alkyl phosphonic acid SAMs, the threshold voltage varies between approximately -1 V and -1.5 V, while for the DPh-BTBT TFTs with the fluoroalkyl phosphonic acid SAMs, the threshold voltage is more or less constant at approximately -1 V. Although two different phosphonic acid molecules were used for the SAMs, the threshold voltage for both types of TFTs is almost consistent, because the threshold voltage range is unique for each semiconductor-gate dielectric combination. The influence of the different functional groups on the threshold-voltage shift has been reported previously [155, 157, 175] and is also detailed in Appendix D by employing different SAMs with the same semiconductor, i.e. DNTT to fabricate organic TFTs. A slight variation in the threshold voltage depending on the chain length can be observed

for the alkyl phosphonic acid molecules, with an increase in the chain length leading to more negative threshold voltages (higher in magnitude). However, this effect cannot be observed for the fluoroalkyl phosphonic acid molecules, which suggests that the chain length of the molecules has little to no effect on the density of defects and charge-carrier traps in the fluoroalkyl phosphonic acid SAMs.



Figure 6.5: Gate-leakage current of (a) DNTT TFTs with hybrid AlO_x/HC_xPA hybrid gate dielectrics and (b) DPh-BTBT TFTs with AlO_x/FC_xPA hybrid gate dielectrics as a function of the phosphonic acid molecule chain length (number of carbon atoms)

The gate current is the leakage current through the gate dielectric in an organic TFT, and the primary advantage of hybrid gate dielectrics is to maintain a low leakage current due to the SAM. The influence of the chain length of the phosphonic acid molecules on the gate current in the respective TFTs can be seen in Fig.6.5. For both the alkyl and the fluoroalkyl phosphonic acids, the gate current varies between approximately 10^{-11} A and 10^{-12} A, with a steady decrease in the gate current with an increase in the chain length of the phosphonic acid molecule. This relationship is consistent with the typical correlation between the gate-dielectric thickness and the gate current in a transistor, i.e. a larger gate-dielectric thickness leads to a smaller gate leakage current. As smaller leakage currents are desirable for better transistor characteristics, the longer chain length molecules are more suitable for SAMs in hybrid gate dielectrics towards low-power organic TFTs.



Figure 6.6: Charge-carrier mobility of (a) DNTT TFTs with AlO_x/HC_xPA hybrid gate dielectrics and (b) DPh-BTBT TFTs with AlO_x/FC_xPA hybrid gate dielectrics as a function of the phosphonic acid molecule chain length (number of carbon atoms)

The charge-carrier mobility of an organic TFT is one of the most important TFT characteristics and is plotted as a function of the chain length of the respective phosphonic acid SAMs in Fig.6.6. It can be observed that the range of charge-carrier mobilities of the DNTT TFTs is much higher $(0.8 \text{ cm}^2/\text{Vs} - 2 \text{ cm}^2/\text{Vs})$ than those of the DPh-BTBT TFTs $(0.2 \text{ cm}^2/\text{Vs} - 0.4 \text{ cm}^2/\text{Vs})$, most likely due to a higher intrinsic mobility for the semiconductor DNTT. Nevertheless, a trend in the charge-carrier mobility with the chain length of the phosphonic acid molecule can be seen for both cases. As the chain length of the phosphonic acid molecule initially increases, the charge-carrier mobility also increases. Considering the range of molecules studied in this case, the carrier mobility reaches a maximum for the medium chain length molecules: $2 \text{ cm}^2/\text{Vs}$ for the DNTT TFTs with $HC_{12}PA$ and $HC_{14}PA$ SAMs and 0.4 cm²/Vs for the DPh-BTBT TFT with the $FC_{10}PA$ SAM. Thereafter, as the chain length of the phosphonic acid molecule increases, the carrier mobility decreases for both sets of TFTs. To explain the above trend seen for the charge-carrier mobility, the growth and morphology of the respective organic semiconductors will be considered along with the surface properties of the SAM in the following sections.


Figure 6.7: Subthreshold swing of (a) DNTT TFTs with AlO_x/HC_xPA hybrid gate dielectrics and (b) DPh-BTBT TFTs with AlO_x/FC_xPA hybrid gate dielectrics as a function of the phosphonic acid molecule chain length (number of carbon atoms)

The subthreshold swing, as defined in Chapter 2, characterizes the subthreshold region of TFT operation and is largely defined by the density of electrically active impurities at the gate dielectric-semiconductor interface. Fig.6.7 shows the variation of the subthreshold swing as a function of the chain length of the respective phosphonic acid molecules in DNTT and DPh-BTBT TFTs. A smaller subthreshold swing indicates a smaller density of trap states in the gate dielectric as well as at the dielectric-semiconductor interface. A smaller subthreshold swing for the alkyl phosphonic acid SAMs can be observed for the medium-chain length $HC_{10}PA-HC_{14}PA$ molecules (89 mV/dec - 92 mV/dec). Similarly, the smallest subthreshold swing for the fluoroalkyl phosphonic acid SAMs is obtained for the medium chain length $FC_{10}PA$ molecule (125 mV/dec).

According to Hill et al., the chain length of the molecule forming the SAM influences the density of trap states at the dielectric-semiconductor interface [243]. If the chain length is too short, an interaction with the underlying gate oxide may lead to deep electron trap states. On the other hand, long chain length molecules produce surface disorder, leading to shallow trap states at the surface. In both cases, the increased density of trap states leads to a higher subthreshold swing. This is a likely explanation for medium-chain-length phosphonic acid molecules forming SAMs with a lower subthreshold swing, as observed in Fig.6.7.



Figure 6.8: Current on-off ratio of (a) DNTT TFTs with AlO_x/HC_xPA hybrid gate dielectrics and (b) DPh-BTBT TFTs with AlO_x/FC_xPA hybrid gate dielectrics as a function of the phosphonic acid molecule chain length (number of carbon atoms)

The current on-off ratio is defined as the ratio between the drain current (I_D) in the onstate of transistor operation (measured at a gate-source voltage of -3 V) and the drain current in the off-state (measured at a gate-source voltage of 0 V) and is plotted in Fig.6.8 as a function of the respective phosphonic acid molecule chain length. With the exception of the HC₁₂PA SAM for the alkyl phosphonic acid molecules, the general trend of the current on-off ratio with the varying chain length of the phosphonic acid molecules is similar to that observed for the carrier mobility and the subthreshold swing, i.e. a preferred large current on-off ratio is obtained for the medium-chain-length molecules. The largest values are obtained for the HC₁₄PA SAM in the alkyl phosphonic acids (1.6x10⁷) and for the FC₁₀PA SAM in the fluoroalkyl phosphonic acids (3x10⁶).

The different TFT characteristics suggest that the medium-chain-length molecules of the available range of alkyl and fluoroalkyl phosphonic acid molecules are an optimum choice for SAMs in hybrid dielectrics from a technological point of view. The next sections examine the properties of the SAMs as a function of the chain length of the phosphonic acid molecule, and the organic-semiconductor layer properties to correlate them with the TFT characteristics observed above.

6.4 Surface properties

6.4.1 Surface roughness and Surface energy

As was observed in Chapter 4, the surface properties of the gate dielectric are influential in modifying the organic-semiconductor morphology and thereby the electrical properties of the organic TFTs. In Chapter 4, the different combinations of the plasma-process parameters were instrumental in modifying the surface roughness of the AlO_x surface and thereby the surface of the conformal SAM. The surface roughness of the different phosphonic acid SAMs is plotted in Fig.6.9(a) for the AlO_x/HC_xPA hybrid gate dielectrics, i.e. the alkyl phosphonic acid SAMs and in Fig.6.9(b) for the AlO_x/FC_xPA hybrid gate dielectrics, i.e. the fluoroalkyl phosphonic acid SAMs.



Figure 6.9: Surface roughness of (a) AlO_x/HC_xPA and (b) AlO_x/FC_xPA surfaces as a function of the phosphonic acid molecule chain length (number of carbon atoms) as measured by atomic force microscopy

It was observed in Chapter 4, that the surface roughness of an AlO_x surface prepared using the plasma parameter combination 200 W, 30 s was approximately 0.57 nm(Fig.4.10(a)). It was also observed that due to the conformal nature of the $HC_{14}PA$ SAM, there was no change in the surface roughness for the respective AlO_x and AlO_x/SAM surfaces. Since the underlying AlO_x films were fabricated by the identical plasmaparameter combination (200 W, 30 s), the surface roughness of all the AlO_x films before forming the different SAMs can be assumed to be constant. Fig.6.9 shows that for all the different SAMs, whether the alkyl phosphonic acid or fluoroalkyl phosphonic acid SAMs, the surface roughness has almost no variation with the chain length and is constant at approximately 0.56 nm, similar to the bare AlO_x surface. This suggests that there is no variation in the surface roughness of the SAM surface with the chain length of the molecule and that all phosphonic acid molecules discussed here, form a conformal layer on the underlying AlO_x surface.



Figure 6.10: Surface energy of (a) AlO_x/HC_xPA and (b) AlO_x/FC_xPA surfaces as a function of the phosphonic acid molecule chain length (number of carbon atoms) as determined by contact-angle measurements and calculated using the Owens-Wendt method

It was also observed in Chapter 4 that the plasma conditions have virtually no effect on the surface energy of the AlO_x or the AlO_x/SAM surfaces. The surface energy as a function of the chain length of the different phosphonic acid molecule SAMs in plotted in Fig.6.10(a) for the the AlO_x/HC_xPA SAM surfaces and in Fig.6.10(b) for the AlO_x/FC_xPA SAM surfaces. The first important observation is that the surface energy for the fluoroalkyl phosphonic acids is between 13 mJ/m² and 18 mJ/m², which is on an average lower than that for the alkyl phosphonic acids, which is between 21 mJ/m² and 24 mJ/m². This is an effect of the strong electronegative fluorine atom substitutions and has also been observed in other literature reports investigating fluoroalkyl phosphonic acid SAMs [155].

The surface energy of the medium-chain-length $HC_{14}PA$ SAM is 21.5 mJ/m² and of the long chain $HC_{18}PA$ SAM is approximately 22.8 mJ/m², showing that the surface energy has a systematic variation with the chain length of the alkyl phosphonic acid SAMs. For

the fluoroalkyl phosphonic acid SAMs, a larger degree of variation is observed with the chain length of the molecules and the lowest surface energy of 13 mJ/m^2 is observed for the FC₁₀PA SAM. Similar to other properties discussed above, the surface energy which was unaffected by the properties of the AlO_x layer, can be varied by the chain length and the type of the phosphonic acid molecule. The possibility of the surface energy affecting the growth and morphology of the organic-semiconductor layer will be discussed in the next section.

6.4.2 Morphology of SAMs as determined by the chain length of the individual molecules

Apart from the surface roughness and the surface energy of the SAMs, another important aspect of SAMs is the packing density and the molecular ordering of the individual molecules within the SAM. Several studies have characterized different types of SAMs using techniques such as contact-angle measurements, spectroscopic ellipsometry, infrared spectroscopy and X-ray photoelectron spectroscopy. Based on these in-depth studies, a basic idea of the morphology of SAMs dependent on the chain length of the individual molecules is presented below.



Figure 6.11: Schematic representation of the molecular ordering and packing density of SAMs dependent on the chain length of the individual molecules

After the almost instantaneous adsorption of the head group on the substrate, the aliphatic chains of the individual molecules start to align themselves by Van der Waals bonding forces among them to form the SAM. Based on the extensive studies of different SAMs, it can be deduced that the extent of the bonding forces among the aliphatic chains depends on the chain length of the molecule. A schematic representation of the proposed morphology of the SAMs can be seen in Fig.6.11. For short chain length molecules (chain length shorter than 12 atoms), the interaction between the chains is smaller and the molecular chains are largely disordered. As the chain length increases, the bonding forces of the medium chain length molecules also increase and the SAM becomes more well-ordered and the packing density within the SAM increases. An increased interaction among the individual aliphatic chains is also expected to lower the surface energy of the SAM, as is also observed in Fig.6.10, in which the medium-chainlength molecules form SAMs with the lowest surface energy. For a further increase in the chain length of the molecule, it has been reported that the aliphatic chains are not stable enough to maintain the ordered nature of the SAM. Typically, gauche defects and surface disorder are observed at the surface of the SAMs with long-chain-length molecules. In such a case, the thickness of the SAM may not linearly increase with the chain length of the molecules, as seen with the trend in the gate-dielectric capacitance (Fig.6.2) of hybrid gate dielectrics. An increase in the disorder within the SAM, either for short-chain-length or long-chain-length molecules is also expected to increase the surface energy of the SAM, as observed in Fig.6.10 for the fluoroalkyl phosphonic acid SAMs.

The packing density and molecular ordering of the SAM is reported to influence the growth mode of the organic-semiconductor layer, besides the influence of other surface properties of the gate dielectric. A spectroscopic investigation of the SAMs to quantitatively measure the molecular ordering is beyond the scope of this thesis, but the general understanding of the morphology of the SAMs as explained above can be corroborated further by investigating the growth and morphology of organic semiconductors on different SAM surfaces.

6.5 Growth and morphology of organic semiconductors on self-assembled monolayers

The growth mode of the organic semiconductor on the gate-dielectric surface is crucial for the electrical properties of the organic TFT, and depends on a variety of factors such as the surface properties of the gate dielectric, and deposition parameters such as the deposition rate, the substrate temperature during deposition, etc. The organic semiconductor molecules are deposited by thermal sublimation in a vacuum chamber. The organic-semiconductor material is placed in a container, which is resistively heated in order to sublimate the molecules for deposition. The substrate on to which the molecules are to be deposited is placed at a certain distance away from the material source, and is initially covered with a metal plate, hereafter referred to as the shutter. Once a suitable and stable rate of deposition is obtained, the metal shutter is moved away from the substrate to expose it to the sublimated organic-semiconductor molecules. The thickness of the film is controlled by a quartz microbalance and once the desired thickness is deposited, the metal shutter is moved to cover the substrate and stop the deposition process.

The organic-semiconductor morphology is usually observed by microscopy techniques, in which each observation of a microscopy image may take up to a few minutes. Due to the different time scales of observation with the microscopy techniques, and the growth process of the organic semiconductor, as well as the absence of a combined equipment for deposition and imaging, it is practically impossible to monitor the growth process of the organic semiconductor in situ. Therefore, the regular deposition process needed to be altered in order to make it possible to observe the growth mode of the organic semiconductor on different gate-dielectric surfaces. This was achieved by altering the deposition process and the movement of the shutter during the semiconductor deposition. Instead of a semiconductor film of a uniform thickness, the organic semiconductor was deposited with an increasing thickness along the length of the substrate. By imaging the organic semiconductor with consecutively increasing thicknesses, i.e. at the different stages of growth and compare the growth mode of the organic semiconductor on different gate dielectric surfaces.

6.5.1 Growth of the organic semiconductor DNTT on different surfaces

The growth mode of the small-molecule organic semiconductor DNTT was observed by modifying the deposition setup of the thermal sublimation process. Usually, a metal shutter covers the substrate and is moved completely when a suitable deposition rate is obtained, to expose the substrate for semiconductor deposition. In the modified setup, a motor was attached to the shutter to gradually move it during the semiconductor deposition, to deposit the organic semiconductor with a thickness gradient along the length of the substrate. The details of this experimental setup are given in Section 6.6. All other factors, including the rate of deposition and the substrate temperature during deposition were kept constant. Four different terminating surfaces were chosen as the substrate to deposit the semiconductor DNTT on: (a) bare AlO_x (Fig.6.12), (b) $AlO_x/HC_{14}PA$ SAM (Fig.6.13), (b) $AlO_x/HC_{18}PA$ SAM (Fig.6.14), and (d) $AlO_x/FC_{14}PA$ SAM (Fig.6.15). The growth mode of the semiconductor was observed by scanning electron microscopy (SEM) by imaging the sample at different positions located equidistant from each other. The different stages of growth are marked as 1-12, corresponding to different thicknesses of the semiconductor DNTT along the thickness gradient, and together show the progression of growth as the thickness of the semiconductor increases. Due to a difference in the atomic weight of the materials forming the dielectric and the semiconductor, it is clear that the darker regions represent the semiconductor and the lighter background is the underlying gate dielectric surface.



Figure 6.12: Progression of growth of the organic semiconductor DNTT on a bare-AlO_x surface observed by depositing a thickness gradient of DNTT and imaged by scanning electron microscopy at equidistant locations. The different stages (1-12) denote the different thicknesses of DNTT from a nominal thickness of approximately 1 nm to 7 nm deposited along the thickness gradient

The surface energy of the bare AlO_x surface is approximately 65 mJ/², the largest of

all four surfaces considered, and the AlO_x surface is extremely hydrophobic. On this bare AlO_x surface, the DNTT molecules nucleate as relatively small islands, which continue to grow and connect with each other. It can be observed in Fig.6.12, in stages 8 and 9, that before the first monolayer is formed as a complete layer, more islands are formed on top as the second monolayer. Towards the end of the observed growth process, the second monolayer islands can be seen on top of the disconnected first monolayer. This type of growth can be characterized as the island type, or 2D form of growth. The two alkyl phosphonic acid molecules considered in this section, HC₁₄PA and HC₁₈PA, form SAMs with a very small difference in their surface energies (21.5 $\rm mJ/m^2$ and 22.8 mJ/m^2 respectively). However, as can be seen in Fig.6.13 and Fig.6.14, the semiconductor DNTT has significantly different growth patterns on the two surfaces. On the medium chain length HC₁₄PA SAM, the semiconductor DNTT has a typical island-layer growth (Stranski-Krastanov growth mode) where the islands which have nucleated in the initial stages, grow and coalesce to form a complete monolayer (as seen in Stages 5 and 6). Subsequently, the second monolaver starts growing as islands on the first complete monolayer (Stages 7 and 8). The isolated islands seen in Stages 6, 7 and 8 on the exposed gate dielectric surface are a result of spontaneous morphological changes occurring in ultrathin semiconductor films. This phenomenon is explained in detail in Chapter 7. It is important to note that these structures have formed as a result of the morphological changes, and are not part of the growth process. The evidence of the second monolayer of DNTT molecules forming as islands on top of the first complete monolayer film can be seen in growth stages 7 and 8, confirming that the growth mode is an island-layer (Stranski-Krastanov) type of growth.



2.5 µm

Figure 6.13: Progression of growth of the organic semiconductor DNTT on an $AlO_x/HC_{14}PA$ surface observed by depositing a thickness gradient of DNTT and imaged by scanning electron microscopy at equidistant locations. The different stages (1-12) denote the different thicknesses of DNTT from a nominal thickness of approximately 1 nm to 7 nm deposited along the thickness gradient

AIO_x / HC₁₈PA SAM / DNTT



2.5 µm

Figure 6.14: Progression of growth of the organic semiconductor DNTT on an $AlO_x/HC_{18}PA$ surface observed by depositing a thickness gradient of DNTT and imaged by scanning electron microscopy at equidistant locations. The different stages (1-12) denote the different thicknesses of DNTT from a nominal thickness of approximately 1 nm to 7 nm deposited along the thickness gradient

It can also be observed that the initial nucleated islands on the surface of the $HC_{14}PA$ SAM, as seen in Stage 1, are the largest in size as compared to the other surfaces, signifying a smaller nucleation density. In contrast, the growth pattern of the semiconductor DNTT on the surface of the $HC_{18}PA$ SAM is notably different. The initial nucleation of islands is relatively irregular and has a higher nucleation density. The growth can still be characterized as island-layer type of growth as the islands are seen to grow and proceed towards forming a complete monolayer, without a second monolayer starting to nucleate. However, it must be noted that for the same thickness of the semiconductor deposited as on the $HC_{14}PA$ SAM, the first monolayer of DNTT on the $HC_{18}PA$ SAM is still incomplete and shows a significant exposure of the underlying gate dielectric.



Figure 6.15: Progression of growth of the organic semiconductor DNTT on an $AlO_x/FC_{14}PA$ surface observed by depositing a thickness gradient of DNTT and imaged by scanning electron microscopy at equidistant locations. The different stages (1-12) denote the different thicknesses of DNTT from a nominal thickness of approximately 1 nm to 7 nm deposited along the thickness gradient

The $AlO_x/FC_{14}PA$ SAM has the lowest surface energy of the four surfaces considered in this section (15 mJ/m²). When the semiconductor DNTT is deposited onto the $AlO_x/FC_{14}PA$ SAM surface, as can be seen in Fig.6.15, it has an extremely high nucleation density with a large number of nuclei of extremely small size forming in the first stage. As the semiconductor deposition progresses, it appears that the islands grow in size towards coalescing and forming a complete film, but a complete monolayer is not

Substrate	$\begin{array}{c} Surface \ energy \\ (mJ/m^2) \end{array}$	Charge-carrier mobility (cm^2/Vs)	DNTT growth mode
AlO_x	65	0.4	island
$AlO_x/HC_{14}PA$	21.6	2	island-layer
$AlO_x/HC_{18}PA$	22.8	1.2	island-layer
$AlO_x/FC_{14}PA$	15	0.6	unclear

Table 6.2: Summary of surface energy of different gate dielectrics, the charge-carrier mobility of DNTT TFTs, and the growth mode of the semiconductor DNTT on those gate dielectric surfaces as observed by scanning electron microscopy

formed. Due to the small size and large density of the nuclei, it is unclear whether a second monolayer starts nucleating before the first monolayer is complete, and hence the growth mode cannot be clearly defined. Nevertheless, it is clear that for the same nominal amount of the organic semiconductor deposited, the molecule DNTT does not form a complete film on the fluorinated $FC_{14}PA$ SAM surface, and the degree of percolation within the initial monolayer is significantly lower, compared to the other surfaces studied.

Table 6.2 tabulates the four different gate dielectrics considered above, their surface energy, the charge-carrier mobility of DNTT TFTs with the particular gate dielectric and a nominal DNTT thickness of 25 nm and the growth mode of DNTT on the surface as observed above. The charge-carrier mobility was extracted from the transfer characteristics (given in Appendix D) of DNTT TFTs, which were fabricated with all other components and conditions identical except the gate dielectric, for comparison. It is evident from this table and from observing the growth pattern of DNTT on the four different surfaces that there is no systematic correlation between the surface energy of the gate dielectric and the charge-carrier mobility of the organic TFT, in case of the phosphonic acid SAM-DNTT combination. A drastic change in the surface energy from a hydrophobic surface to a hydrophilic surface is likely to influence the growth of the semiconductor due to a change in the wettability of the surface, as is observed with the island growth of DNTT on the bare- AlO_x surface. This effect has also been reported previously in other studies. However, in an attempt to overcome the hurdle of a high surface energy, an uncontrolled reduction in surface energy does not seem to be the solution. As can be seen on the $FC_{14}PA$ SAM surface, which has the lowest surface energy, the semiconductor DNTT does not show a favorable growth pattern, and the DNTT TFT also has a lower charge-carrier mobility. Instead, there exists a complex

relationship between the surface properties of the gate dielectric, the growth mode of the organic semiconductor and thereby the electrical properties of the organic TFT. It must also be noted that each combination of an organic semiconductor and the substrate is unique, and these correlations can very rarely be generalized.

Apart from the surface energy (and the surface roughness), other aspects of SAMs have been reported to influence the growth and morphology of the organic semiconductor, such as the chemical nature, packing density, molecular ordering and defects within the SAM. For example, Virkar et al. showed that the organic semiconductor pentacene showed a change in the growth mode and subsequent quality depending on the density of the underlying OTS (octadecylsilane) SAM. OTS SAMs with a higher packing density and molecular order led to pentacene TFTs with a higher charge-carrier mobility. This was, quite contrary to common belief, attributed to a 2D-island-type growth in pentacene films which resulted in a smaller grain size, but a more complete film coverage in the first monolayer. Based on energetic calculations at the interface, they concluded that a higher ordering in the underlying SAM led to better-quality pentacene films, due to a more suitable growth mode at the gate dielectric-semiconductor interface. For DNTT films observed above on the two alkyl phosphonic acid SAMs, with almost identical surface energies, a similar reasoning can be expected. Surface disorder in the $HC_{18}PA$ SAM may lead to an unfavorable DNTT growth on the surface and a subsequent smaller charge-carrier mobility in the transistor.

Virkar et. al also performed extensive calculations of the interlayer interaction energies and the difference in chemical potentials of the vapor and condensed phases to show that the condition for 2D growth is mathematically satisfied [169]. In this process, along with experimental factors such as the rate of deposition and the substrate temperature which influence the semiconductor growth, other quantities were used from literature, such as the enthalpy of sublimation for pentacene, and the interlayer molecular-substrate energy, determined elsewhere. To establish the exact mechanisms at play for the combination of the semiconductor DNTT and the phosphonic acid SAMs, similar thermodynamic energies will need to be experimentally determined, which is beyond the scope of this current work.

Nevertheless, the growth mode of the semiconductor DNTT on the different surfaces can be interpreted to draw several important conclusions. Firstly, correlations between the charge-carrier mobilities and characteristics such as island/island-layer growth or the grain size of the complete polycrystalline film are incomplete. An organic-semiconductor film might show an island type of growth mode, or a large density of nuclei, and therefore a large density of grain boundaries and yet form a high-quality film, yielding a large carrier mobility. Therefore, not just the density of grain boundaries, but also the nature of those grain boundaries are important. From the SEM images of the growth mode of DNTT and the charge-carrier mobility values in Table 6.2, the percolation path in the first monolayer of the organic semiconductor, i.e. the formation of a complete monolayer appears to be of utmost importance to the charge-transport properties of the organic TFT. The highest charge-carrier mobility is obtained for the DNTT TFT with the HC₁₄PA SAM in the hybrid gate dielectric, and DNTT forms a completely closed monolayer film on the gate-dielectric surface, as observed in Fig.6.13. The energetic reasons for having a highly dense percolation path and thereby a complete monolayer seem to be multifold, and need to be carefully investigated for individual gate dielectricsemiconductor combinations.

6.5.2 Morphology of organic semiconductors DNTT and DPh-BTBT on phosphonic acid SAMs of different chain lengths

Along with the growth of the semiconductor on different surfaces, it is equally important to assess the morphology of the organic-semiconductor layer as it is deposited for transistor operation.



Figure 6.16: Morphology of the organic semiconductor DNTT with a nominal thickness of 2.5 nm, 5 nm and 25 nm on three alkyl phosphonic acid SAMs as imaged by atomic force microscopy

As each gate dielectric-organic semiconductor combination must be considered individually, this section investigates the morphology of the organic semiconductors used to study the electrical properties of organic TFTs in this chapter, i.e. DNTT with different chain-length alkyl phosphonic acid SAMs and DPh-BTBT with different chain-length fluoroalkyl phosphonic acid SAMs. Different thicknesses of the organic semiconductor were deposited onto the respective SAM surfaces, closer to the dielectric-semiconductor interface and at its full thickness, and imaged using atomic force microscopy (AFM). Fig.6.16 shows the AFM micrographs of nominally 2.5 nm, 5 nm and 25 nm-thick DNTT films on three different alkyl phosphonic acid SAMs: the shortest-chain-length HC_6PA , the medium-chain-length $HC_{14}PA$ and the longest-chain-length $HC_{18}PA$. Considering the nominally 2.5 nm-thick DNTT films, an initial growth pattern of the semiconductor DNTT can be observed on the three surfaces, as was observed in the previous section. The semiconductor forms a complete, uniform film on the medium chain length $HC_{14}PA$ SAM, while a relatively more disconnected monolayer film with taller islands on the other two SAMs. It can also be observed that the island size is relatively large on the HC_6PA SAM as compared to that of the $HC_{18}PA$ SAM. For the nominally 5 nm-thick films of DNTT, a second-monolayer morphology can be observed, and the largest grain size for DNTT is seen on the $HC_{14}PA$ SAM. A continuation of this distinction can also be seen in the nominally 25 nm thick DNTT films as they are deposited for transistor operation. Although all films show a dendritic terrace-like structure, the largest grain size and thereby the relatively smallest density of grain boundaries for the semiconductor DNTT can be observed on the medium-chain-length $HC_{14}PA$ SAM.



Figure 6.17: Morphology of the organic semiconductor DPh-BTBT with a nominal thickness of 5 nm and 25 nm on three fluoroalkyl phosphonic acid SAMs as imaged by atomic force microscopy

Similarly, for the DPh-BTBT TFTs with the different fluoroalkyl phosphonic acid SAMs, Fig.6.17 shows the morphology of nominally 5 nm and 25 nm-thick films of DPh-BTBT on three SAMs with the respective fluoroalkyl phosphonic acid molecules: FC_6PA , $FC_{10}PA$ and $FC_{14}PA$. Considering the nominally 5 nm-thick DPh-BTBT films, the formation of the second monolayer can be seen during the semiconductor growth process. It can be observed that the grain size of the DPh-BTBT layer is relatively the smallest on the long chain length $FC_{14}PA$ SAM. The AFM images of the nominally 25 nm-thick DPh-BTBT films, as they are deposited for TFT operation, give a much clearer picture of the film quality. The DPh-BTBT film on the short-chain-length FC_6PA SAM shows an almost amorphous film structure with no distinct grains or terrace-like structures. There are also minor discontinuities in the film which can be observed. For the medium-chain-length $FC_{10}PA$ SAM, a favorable semiconductor morphology can be observed, with a dendritic grain structure and relatively large-sized terraces. In contrast, the DPh-BTBT film on the long-chain $FC_{14}PA$ SAM shows a relatively smaller grain size with a larger density of grain boundaries in the film.

The growth of the semiconductor DNTT and the film morphology of the semiconductors DNTT and DPh-BTBT on different phosphonic acid SAMs provide useful insights into the correlation between the chain length of the phosphonic acid molecules and electrical characteristics of the organic TFTs. Although the correlation between the grain size of an organic-semiconductor film and the charge-carrier mobility of an organic TFT cannot be generalized, it appears that in the combination of materials studied in this work, a larger grain size in the films of DNTT and DPh-BTBT is beneficial for larger charge-carrier mobilities. In both cases, it was observed that for the long-chain-length molecules ($HC_{18}PA$ and $FC_{14}PA$), the grain size of the semiconductor films was relatively very small. This effect may be a result of the surface disorder in long-chain phosphonic acid molecules. The highest charge-carrier mobility in DNTT TFTs was found for the medium-chain-length $HC_{14}PA$ SAM and in DPh-BTBT TFTs for the medium-chainlength $FC_{10}PA$ SAM, which in both cases had the largest grain size and the smallest density of grain boundaries. A number of tall features can be observed in the DNTT films on the $HC_{14}PA$ SAM as well as the DPh-BTBT films on the $FC_{10}PA$ SAM, which are characteristic of high-mobility semiconductor films, although the exact origin and nature of these features will need further investigation. It can be concluded for this particular case, a large grain size and more importantly, the formation of a complete film in the initial monolayer is vital for high-quality organic TFTs. Although specific reasons behind these layer properties will need more investigation, the optimum choice of phosphonic acid molecules for these organic TFTs from a technological point of view has been identified.

6.6 Experimental

Fabrication of capacitors and TFTs

Using phosphonic acid molecules of different chain lengths shown in Fig.6.18, SAMs of different thicknesses were investigated. For each type of phosphonic acid molecule, capacitors and organic TFTs with hybrid gate dielectrics consisting of AlO_x and the phosphonic acid SAM were fabricated.



Figure 6.18: Schematic cross-section of the organic TFTs and capacitor structures fabricated in this thesis chapter along with the chemical structures of the organic semiconductors DNTT and DPh-BTBT and the different alkyl and fluoroalkyl phosphonic acid molecules used to form SAMs in the hybrid gate dielectrics

All devices were fabricated on flexible poly(ethylene)2,6-naphthalate (PEN) plastic substrates (PEN provided kindly by William A. Macdonald from Du Pont Teijin Films, Wilton, UK). The PEN substrates were annealed at 80°C beforehand in order to avoid any shrinking effects or changes in geometry during the fabrication process. Metalinsulator-metal capacitors and inverted staggered (bottom-gate, top-contact) TFTs were fabricated on the flexible PEN substrates. As a first step, 20 nm thick gold contact pads were deposited to make electrical contact with the gate electrode later possible. For the bottom electrode of the capacitors and the gate electrode of the TFTs, aluminum films with a nominal thickness of 30 nm were deposited by thermal evaporation in vacuum with a rate of about 2 nm/s. AlO_x films were produced by plasma oxidation using a capacitively coupled plasma mode with a partial oxygen pressure of 0.01 mbar and an oxygen flow rate of 30 sccm. The plasma power selected was 200 W and the duration of plasma exposure was limited to 30 s. After formation of the AlO_x component of the gate dielectric, the substrates were immersed into a solution of a particular phosphonic acid molecule in 2-propanol to allow the molecules to assemble into a monolayer, i.e. the self-assembled monolayer (SAM). Each substrate was kept immersed in solution for 3 hours, after which it was rinsed with isopropanol to remove any non-chemisorbed molecules, dried with dry nitrogen and subsequently exposed to 80°C for 10 minutes to stabilize the monolayer.

Alkyl phosphonic acid molecules of seven different chain lengths, from six carbon atoms to eighteen carbon atoms in the aliphatic chain were used, the chemical names are given below: (Alkyl phosphonic acid molecules were procured from PCI Synthesis, Newburyport, MA, USA).

- HC₆PA : hexylphosphonic acid
- HC₈PA : octylphosphonic acid
- HC₁₀PA : decylphosphonic acid
- HC₁₂PA : dodecylphosphonic acid
- HC₁₄PA : tetradecylphosphonic acid
- HC₁₆PA : hexadecylphosphonic acid
- HC₁₈PA : octadecylphosphonic acid

The alkyl phosphonic acid molecules were used to fabricate organic TFTs in combination with the small-molecule organic semiconductor dinaphtho[2,3-b:2',3'-f]thieno[3,2b]thiophene (DNTT; Sigma Aldrich).

Fluoroalkyl phosphonic acid molecules of five different chain lengths, from six carbon atoms to fourteen carbon atoms in the aliphatic chain were used, the chemical names are given below: (Fluoroalkyl phosphonic acid molecules were procured from Specific Polymers).

• FC₆PA : 1H,1H,2H,2H-perfluorohexylphosphonic acid

- FC₈PA : 1H,1H,2H,2H-perfluorooctylphosphonic acid
- FC₁₀PA : 1H,1H,2H,2H-perfluorodecylphosphonic acid
- FC₁₂PA : 1H,1H,2H,2H-perfluorododecylphosphonic acid
- FC₁₄PA : 1H,1H,2H,2H-perfluorotetradecylphosphonic acid

The fluoroalkyl phosphonic acid molecules were used to fabricate organic TFTs in combination with the small-molecule organic semiconductor 2,7-diphenyl[1]-benzothieno[3,2b][1]benzothiophene (DPh-BTBT; Sigma Aldrich).

Once the SAM had been formed and the fabrication of the hybrid gate dielectric was complete, the particular small-molecule organic semiconductor was deposited by thermal sublimation in vacuum in a Leybold University vacuum evaporator system at a base pressure of 10^{-5} mbar, with a deposition rate of 0.03 nm/s and with a nominal thickness of 25 nm. During the DNTT deposition, the substrate was held at a constant temperature of 60 °C, while the substrates were not heated during the DPh-BTBT deposition and kept at room temperature (25 $^{\circ}$ C). For the top electrode of the capacitors and the source/drain contacts of the TFTs, gold with a thickness of 30 nm was deposited by thermal evaporation in vacuum with a rate of 0.03 nm/s. The metals and the organic semiconductors were patterned using polyimide shadow masks (CADiLAC Laser, Hilpoltstein, Germany). The capacitors have an area of 200 $\mu m \times 200 \mu m$. The TFTs have a channel length of 20 μ m and a channel width of 100 μ m. For each of the phosphonic acid molecules, capacitors and TFTs were fabricated on the same substrate to minimize the effects of unintentional parameter variations. Substrates were stored in ambient air at room temperature under yellow laboratory light before electrical measurement.

Experimental setup for observing the growth mode of the organic semiconductor on different gate dielectric surfaces

Silicon substrates with a 100 nm-thick thermally grown silicon dioxide layer with dimensions 9 cm x 2 cm were used to investigate the growth mode of the organic semiconductor DNTT on different surfaces. The experimental setup is shown schematically in Fig.6.19. The aluminum, aluminum oxide, SAMs and DNTT films were deposited as described above.



Figure 6.19: Schematic representation of the experimental setup for observing the growth process of the organic semiconductor on different gate dielectric surfaces

Four different substrates were used to deposit the DNTT on: (a) bare AlO_x , (b) $AlO_x/HC_{14}PA$ (c) $AlO_x/HC_{18}PA$ and (d) $AlO_x/FC_{14}PA$. The deposition setup was modified with a motor attached to the shutter covering the substrate during deposition inside the vacuum chamber. With the assistance of the motor, the shutter was gradually moved at a rate of 0.75 mm/s to gradually expose the substrate along its length during the semiconductor deposition. The semiconductor DNTT was deposited by thermal sublimation at a rate of 0.06 nm/s. The substrate was imaged using a Zeiss Merlin scanning electron microscope (SEM).

Electrical characterization

All electrical measurements were performed in ambient air at room temperature under yellow laboratory light. The capacitance measurements were performed by applying an alternating voltage with an amplitude of 0.2 V and a frequency of 1 kHz. The current–voltage measurements were performed using an Agilent 4156C Semiconductor Parameter Analyzer.

Surface characterization

Static contact-angle measurements were performed using a Krüss contact angle measurement system. The contact angles of water and hexadecane on the AlO_x/SAM hybrid dielectrics were measured immediately after the SAM treatment. The surface energies were calculated using the Owens–Wendt method [99]. AFM images were recorded in air using a Bruker Dimension Icon Atomic Force Microscope in peak force tapping mode. Data processing was performed using the AFM analysis software Gwyddion.

6.7 Summary and Outlook

The aliphatic chain length of molecules used to form SAMs is an important aspect influencing the properties of hybrid gate dielectrics and organic TFTs. Two different phosphonic acid molecules, i.e. molecules with an unsubstituted alkyl chain and molecules with a fluorine-substituted fluoroalkyl chain were investigated with different chain lengths (number of carbon atoms in the aliphatic backbone). With a change in the chain length of the phosphonic acid molecules, the thickness of the SAM can be modified, and the capacitance of the gate dielectric can be varied, which is not possible by varying the thickness of the underlying AlO_x film. Organic TFTs were fabricated with the different chain-length phosphonic acid SAMs, with two different organic semiconductors. The performance characteristics of the TFTs were studied as a function of the chain length and the medium-chain-length molecules for both types of phosphonic acids showed an optimum TFT performance in terms of the charge-carrier mobility, subthreshold swing and the current on-off ratio. The surface properties of the different SAMs were investigated in terms of the surface roughness and the surface energy. The chain length of the molecule had no measurable effect on the surface roughness of the SAMs, unlike the effect observed by changing the plasma parameters of the AlO_x fabrication process. The surface energy of the SAMs changed drastically by changing the type of the molecule from alkyl phosphonic acid molecules to fluoroalkyl phosphonic acid molecules, with a further slight variation with the chain length of the fluoroalkyl phosphonic acids. The morphology of the organic semiconductors DNTT and DPh-BTBT on the different phosphonic acid SAMs revealed that the medium-chain-length phosphonic acid molecules ($HC_{14}PA$ and $FC_{10}PA$) led to the largest grain size in the polycrystalline organic semiconductor films and can be correlated with the largest charge-carrier mobilities in the respective organic TFTs. The growth of the organic semiconductor DNTT on different gate-dielectric surfaces revealed no systematic correlation with the surface energy of the gate-dielectric surface. However, it can be concluded that a well-connected initial monolayer and a dense percolation path for charge carriers during the growth of the semiconductor is essential for high-performance organic TFTs. A further study of the properties of the SAMs with respect to their packing density and molecular order with spectroscopic techniques would improve the understanding of the different growth modes of organic semiconductors and give better control over the electrical properties of organic TFTs.

Chapter 7

Stability of ultrathin organic-semiconductor films and fabrication of ultrathin organic transistors

The previous chapters have all focused on the different components of the hybrid gate dielectric and the influence of the gate dielectric on the electrical properties of organic thin-film transistors (TFTs). Chapter 6 also discussed the effect of the gate-dielectric surface on the growth and morphology of the organic-semiconductor film and in-turn the electrical performance of the organic TFT. In this chapter, the organic-semiconductor layer, specifically the part in close proximity with the gate dielectric will be studied in detail. Apart from the many technological applications, the organic TFT is also a favorable system to explore new organic semiconductors and corresponding device physics. For typical organic TFTs with vacuum-processed organic semiconductors, as have been discussed in this thesis, the semiconductor film is usually several monolayers (MLs) thick. The influence of the semiconductor-layer thickness on different TFT characteristics, especially the charge-carrier mobility, has been extensively investigated by both theoretical simulations as well as experimental studies [245–249]. It has also been demonstrated how a variety of factors like the choice of the semiconductor molecule,

TFT architecture (top-contact or bottom-contact) and semiconductor-deposition parameters influence the growth of the organic-semiconductor layer and the development of the semiconductor-gate dielectric interface. Studies with TFTs based on thiophene oligomers reveal that the gate-field-induced conducting channel in a thin-film transistor, and hence majority of the charge transport, is positioned in the first one to two monolayers of the organic semiconductor thin film closest to the gate dielectric [83,84,250]. For studies with pentacene TFTs, the charge-carrier mobility saturated after six monolayers of pentacene had been deposited [251].

Although great progress has been made in the development of organic TFTs in terms of their performance and stability, there is further scope in terms of exploring the dynamic relationship between the organic-semiconductor-layer thickness and transistor performance to gain further insight into the structure-property relationship at the semiconductor-gate dielectric interface. Fabricating stable and reliable organic TFTs with ultrathin (less than 5 nm thick) semiconductor layers could lead to further development towards understanding intrinsic charge transport, exploring new and unique applications and low-cost device fabrication [252]. Since the conductive channel in bottom-gate architectures would be located at the surface in ultrathin-film transistors, these devices would make it possible to investigate charge transport and the morphology of the semiconductor in the channel using surface-sensitive techniques as opposed to the channel being buried under additional layers for transistors with several monolayer-thick semiconductor layers. Ultrathin devices with a range of semiconductors have also been used to study aspects of transistor performance such as contact resistance, thresholdvoltage shifts, and trap densities and their correlation to the semiconductor thickness. Ultrathin TFTs are also suitable for the fabrication of highly sensitive sensors and could be developed into new applications [53]. There is also a great economic advantage of lowering the overall cost of fabrication by minimizing the material consumption by depositing the semiconductor as an ultrathin layer, especially those fabricated by vacuum deposition techniques which entail a considerable loss of excess material.

In order to fabricate ultrathin organic TFTs for both technological and investigative purposes, it is necessary to have devices with a suitably long stability and reliable, reproducible performance characteristics. One key hurdle in the past concerning the stability of organic thin-film transistors was the environmental stability of small-molecule organic semiconductors, such as pentacene. Pentacene undergoes rapid oxidation when exposed to air, thereby losing its π -conjugation and ability to transport charge, and as a result, a rapid decrease in the charge-carrier mobility of pentacene TFTs is observed upon exposure to air [253-255]. To overcome this, molecules with a higher ionization potential, and hence higher air-stability, were synthesized [101]. These included molecules such as di(phenylvinyl)anthracene (DPVAnt) [81] and dinaphtho[2,3-b:2',3'-f]thieno[3,2b]thiophene (DNTT) [72,256]. DNTT is a particularly promising small-molecule organic semiconductor for the fabrication of organic-TFT-based active-matrix displays [257,258], sensor arrays [259, 260], and integrated circuits [261–264]. As demonstrated in previous chapters of this thesis, DNTT TFTs are p-channel transistors that typically have chargecarrier mobilities up to $3 \text{ cm}^2/\text{Vs}$. In terms of stability against oxidation, chemical decomposition, elevated temperature and formation of defect states, DNTT is one of the most stable organic semiconductors [101,176,189,265,266]. It was also shown in Chapter 6, that on suitable gate-dielectric surfaces, as the $AlO_x/HC_{14}PA$ hybrid gate dielectric, DNTT films typically grow in a Stranski–Krastanov (island-layer) mode, much like films based on other small-molecule organic semiconductors with similar chemical structures. The molecules adopt an upright-standing molecular orientation both in the complete monolayer(s) that initially cover(s) the substrate surface as well as in the islands that are formed beyond a critical film thickness to minimize mechanical strain. This molecular orientation was verified using X-ray diffraction measurements, and the results are given in Appendix A. This upright-standing orientation is generally the preferred molecular arrangement for the purpose of fabricating organic TFTs, as it promotes π -orbital overlap and efficient charge transport in the lateral direction.

However, Breuer et al. recently observed that DNTT films with a nominal thickness of one to two molecular monolayers spontaneously undergo a rapid structural reconfiguration at room temperature within minutes after deposition that proceeds by lateral and vertical mass transport and leads to a rearrangement of the initially closed (or at least connected) single-monolayer-thick DNTT films into disconnected islands with a thickness of several layers [267]. In analogy to the behavior of thin liquid films on solid surfaces [268], this process is also referred to as post-deposition dewetting [269,270] or rapid roughening [271], and it has been observed in ultrathin films based on a variety of organic materials, such as polystyrene [270], diindenoperylene [271] and pentacene [272, 273]. When a molecular monolayer of, for example, DNTT is deposited onto a solid surface, this monolayer will form during a non-equilibrium phase of the film-growth process and will thus be characterized by significant mechanical strain that results from the competition between adsorption and intermolecular forces [274]. If the molecule–substrate interactions are weaker than the intermolecular interactions, thermally activated molecular diffusion will cause this initially closed (or at least connected) molecular monolayer to break up, and a combination of lateral and vertical mass transport will lead to the formation of tall, disconnected islands. This can be considered similar to the Ostwald ripening process often observed during annealing processes, where large, stable droplets grow at the expense of smaller, less stable droplets [275], in order to reduce the contribution of interfacial energy. While temporal morphological changes have also been observed for thick DNTT films (with a nominal thickness of 40 nm) [267], these occur on much larger time scales of weeks or months, due to the substantially larger mass transport involved in such thick films, and will not be considered in this particular thesis.

The focus of this chapter is on the spontaneously occurring structural reconfiguration of ultrathin DNTT films and its effect on the electrical characteristics of organic TFTs based on such films. Furthermore, the extent to which this structural reconfiguration can be prevented by cryogenic cooling or in-situ encapsulation has also been explored. For this purpose, organic TFTs based on ultrathin DNTT films have been fabricated, and unlike in the previous chapters, the TFT architecture of choice is inverted coplanar, i.e. bottom-gate, bottom-contact (Schematic shown in Fig.2.4). This investigation is motivated by the need for a better understanding of the dynamics of the morphological changes that occur in ultrathin films of vacuum-deposited small-molecule semiconductors and of the effects that these morphological changes have on the long-term stability of ultrathin devices, such as highly sensitive chemical, physical or biological sensors.

An important prerequisite for this investigation is the ability to unambiguously assign the changes observed in the electrical properties of the ultrathin semiconductor films to the mass transport and resulting changes in morphology, as opposed to any other processes, such as any chemical reactions, e.g. the oxidation or decomposition of the molecules during exposure to air and humidity. This makes DNTT the perfect choice for this study, as DNTT is one of the most stable organic semiconductors with respect to long-term exposure to ambient air and elevated temperatures. Owing to its excellent chemical stability, the changes in the measured electrical properties of the ultrathin DNTT films can be unequivocally assigned to the morphological changes observed.

7.1 Ultrathin films of the organic semiconductor DNTT

7.1.1 Determining the minimum nominal DNTT thickness required for functional TFTs

Before observing the morphological changes in ultrathin films of the organic semiconductor DNTT, it is necessary to determine the smallest nominal DNTT thickness for which functional TFTs can be obtained. Four different DNTT films with nominal thicknesses of 1, 1.5, 2 and 2.5 nm were deposited on to an $AlO_x/HC_{14}PA$ SAM hybrid gate-dielectric surface by sublimation in vacuum. Immediately after the DNTT deposition, the substrates were removed from the vacuum system, and the morphology of the DNTT films was examined by atomic force microscopy (AFM). Fig.7.1 shows the AFM images of nominally 1, 1.5, 2 and 2.5 nm thick DNTT films, each obtained within 15 minutes of deposition. The AFM images indicate that a nominal thickness of 1 nm results in disconnected islands, and although the islands appear to be progressively more connected for nominal thicknesses of 1.5 and 2 nm, a closed film is clearly identified only for a nominal thickness of 2.5 nm.



Figure 7.1: Thin-film morphology of DNTT films with nominal thicknesses of 1, 1.5, 2 and 2.5 nm deposited onto a hybrid $AlO_x/HC_{14}PA$ SAM gate dielectric, observed by AFM within 15 minutes after completion of the DNTT deposition

Based on these results, the minimum nominal thickness for DNTT ultrathin films was chosen to be 2.5 nm. Based on the fact that the DNTT molecules have a length of approximately 1.6 nm [72] and stand approximately upright on an electrically insulating surface (as confirmed by the height profile shown in Appendix A), a nominal thickness of 2.5 nm corresponds to less than two complete molecular monolayers. As the nominal DNTT thickness is increased from 1.0 to 2.5 nm, the RMS surface roughness decreases from 1.86 nm to 0.69 nm (as measured by AFM), indicating a more planar and uniform film morphology with increasing nominal thickness.

7.1.2 Temporal changes in the morphology of ultrathin DNTT films

Having determined the minimum nominal DNTT thickness required for functional TFTs to be 2.5 nm, the time-dependent changes in the morphology of these films were monitored. For this purpose, AFM images were recorded on a nominally 2.5 nm-thick DNTT film in a quasi-continuous manner by repeatedly scanning the same area on the surface of the DNTT film over a period of 12 hours. This made it possible to observe the process of the spontaneous structural reconfiguration essentially in real time.



Figure 7.2: Spontaneously occurring morphological changes in a DNTT film with a nominal thickness of 2.5 nm deposited onto a hybrid $AlO_x/HC_{14}PA$ SAM gate dielectric, observed by AFM over a period of 12 hours

Fig.7.2 shows six snapshots obtained in intervals of three hours, with the first image taken within 15 minutes after completion of the DNTT deposition. To rule out that the morphological changes seen in these AFM images were caused by the force of the AFM tip, an identical substrate was prepared on which AFM images were taken only at 15 minutes after completion of the DNTT deposition and again at 12 hours after deposition, and these images indicate the same time-dependent changes in the morphology of the DNTT films. The AFM images confirm the spontaneous rearrangement of the DNTT films, as reported by Breuer et al [267]. The maximum height of the DNTT film evolves from 3.18 nm (about 2 molecular monolayers), 15 minutes after deposition to 10.9 nm (about 6 to 7 molecular monolayers), 12 hours after deposition. The root-mean-square surface roughness increases from 0.69 to 2.61 nm (as measured by AFM) in the same duration. This increase in the maximum height and RMS surface roughness is accompanied by the appearance of tall, disconnected islands and by a substantial reduction of the surface coverage, i.e. by the appearance of larger and larger areas of the gate-dielectric surface not covered with DNTT molecules.



Figure 7.3: Evolution of the morphology of nominally 2.5-nm-thick DNTT films deposited onto (a) bare AlO_x (without SAM); (b) AlO_x/HC_6PA SAM (c) $AlO_x/HC_{18}PA$ SAM. The AFM images were recorded within 15 minutes after the completion of the DNTT deposition and again after 12 hours

The measurements in Fig.7.2 were performed on a DNTT film deposited onto an $AlO_x/HC_{14}PA$ SAM gate-dielectric surface, as this was determined to be the optimum choice of SAM

for TFT performance and the growth of the semiconductor in Chapter 6. Since morphological changes in ultrathin DNTT films were previously also observed on other insulating surfaces, it was necessary to observe ultrathin DNTT films on other gatedielectric surfaces relevant to organic TFTs. DNTT films with a nominal thickness of 2.5 nm were deposited onto bare AlO_x , and two hybrid gate-dielectric surfaces with the shortest and longest available alkyl phosphonic acid SAMs: AlO_x/HC_6PA SAM and $AlO_x/HC_{18}PA$ SAM, and the morphological changes over time were observed in a similar way. The results are summarized in Fig.7.3. Although there are some differences in the initial morphology of the ultrathin DNTT films on these surfaces, as is already detailed in Chapter 6, the morphological changes in the ultrathin DNTT films are quite similar, suggesting that they are not significantly affected by the surface energy of the substrate. As it has been suggested in previous reports which have studied similar morphological changes in ultrathin organic-semiconductor films, the reorganization of the molecules occurs due to the relatively stronger molecule-molecule interactions as compared to the substrate-molecule interactions. Therefore, in order for the substrate to influence the morphological changes towards stabilizing the organic-semiconductor monolayer, the substrate-molecule interactions would have to be stronger than those observed for the gate-dielectric surfaces considered in Fig.7.3. However, it must also be noted that with stronger substrate-molecule forces, there is a high probability of the organic-semiconductor molecules adopting a horizontal orientation, parallel to the substrate, which is not conducive to charge transport in organic TFTs.

7.1.3 Influence of the changes in the DNTT thin-film morphology on TFT characteristics

To investigate how the spontaneous morphological changes in the DNTT films described above affect the performance of TFTs in which such films serve as the semiconductor, organic TFTs were fabricated in the inverted coplanar (bottom-gate, bottom-contact) architecture. This TFT configuration was particularly chosen so that the deposition of the organic semiconductor is the last step in the TFT fabrication process, and that the TFTs can be electrically measured on the same time scale as the morphological changes occurring in the semiconductor films.



Figure 7.4: Influence of the nominal thickness of vacuum-deposited DNTT films on the evolution of the electrical characteristics of TFTs based on these films (a) Measured transfer curves of TFTs with an ultrathin DNTT film (nominal thickness of 2.5 nm) measured repeatedly over a period of 10 days after TFT fabrication (b) Transfer curves of TFTs with a thick DNTT film (nominal thickness of 25 nm) measured over a period of 20 days after fabrication. (c) Evolution of the effective charge-carrier mobility extracted from the measured transfer curves

The organic TFTs were fabricated with two different thicknesses of the DNTT semiconductor films, nominally 2.5 nm and 25 nm-thick films to compare the effect of the morphological changes in the ultrathin films with the relatively stable thick films on the electrical characteristics of organic TFTs, with the results summarized in Fig.7.4. The initial TFT characteristics for both TFTs are quite similar, with effective chargecarrier mobilities and threshold voltages of $0.4 \text{ cm}^2/\text{Vs}$ and -1.5 V for a nominal DNTT thickness of 2.5 nm and $1 \text{ cm}^2/\text{Vs}$ and -1.7 V for a nominal DNTT thickness of 25 nm. This confirms that it is indeed possible to fabricate functional TFTs based on vacuumdeposited DNTT films with a nominal thickness of less than two molecular monolavers. However, the evolution of the effective charge-carrier mobility, as shown in Fig.7.4(c), is dramatically different for the two nominal DNTT thicknesses. While the effective charge-carrier mobility of the TFTs with the thick DNTT film decreases by less than 20% over a period of 20 days, the effective carrier mobility of the TFTs with the ultrathin DNTT film decreases far more rapidly and far more dramatically, from initially 0.4 to $0.0007 \text{ cm}^2/\text{Vs}$ (a 99.9% decrease) within the first five days. The rapid decrease of the effective charge-carrier mobility of the TFTs with the ultrathin DNTT film seen in Fig.7.4(c) is consistent with the rapid reduction of the surface coverage of the semiconductor and the degree of percolation through the ultrathin DNTT film during the spontaneous rearrangement of the initially closed (or at least connected) ultrathin film into disconnected islands seen in Fig.7.2. Although the morphological changes commence immediately after the completion of the DNTT deposition, their cumulative impact on the surface coverage and the degree of percolation appears to be most pronounced during the period between about one hour and about six hours after the DNTT deposition, which explains the observation that the drop in the effective charge-carrier mobility is most prominent during this period. The results confirm the severe impact of the spontaneous morphological changes of ultrathin DNTT films reported by Breuer et al. [267] on the effective charge-transport characteristics of organic TFTs based on the ultrathin DNTT films.

The thick DNTT films, with a nominal thickness of 25 nm, are far more stable than the ultrathin DNTT films, both in terms of the film morphology as observed by AFM, as seen in Fig.7.5, and in terms of the electrical TFT characteristics, as seen in Fig.7.4(b). This was already reported by Breuer et al. [267] and indicates that the spontaneous disaggregation of the first molecular monolayer into disconnected islands seen in Fig.7.2 is effectively prevented when this first molecular monolayer is covered with a sufficient amount of additional material.



Figure 7.5: Morphology of a nominally 25-nm-thick thick DNTT film deposited onto a hybrid $AlO_x/HC_{14}PA$ SAM surface. The AFM images were recorded within 15 minutes after the completion of the DNTT deposition and again after 12 hours

7.2 Effect of cryogenic cooling of ultrathin DNTT films

To confirm that the spontaneously occurring mass transport within the ultrathin DNTT films that leads to the disaggregation of the initially closed monolayer is indeed thermally activated, a substrate with an ultrathin DNTT film was cooled to a temperature of 193 K in dry nitrogen, immediately after completion of the DNTT deposition and stored at this temperature for a duration of 18 days¹. Prior to cooling the substrate, the morphology of the film was recorded immediately after deposition, as shown in Fig.7.6. After cryogenic cooling for 18 days, the morphology of this ultrathin film was recorded immediately after allowing the substrate to warm up to room temperature, shown in Fig.7.6(b). An identical substrate with an identical ultrathin DNTT film served as a reference on which an AFM image was recorded immediately after the DNTT deposition and which was kept in ambient air at room temperature for a period of two weeks, after which another AFM image was taken, shown in Fig.7.6(c).



Figure 7.6: Effect of cryogenic cooling on the spontaneously occurring morphological changes in ultrathin DNTT films, observed by AFM at room temperature (a) DNTT morphology immediately after completion of the DNTT deposition (b) DNTT morphology after the substrate had been stored at a temperature of 193 K for 18 days (c) DNTT morphology on a reference substrate stored at room temperature for two weeks

Immediately after the DNTT deposition, the DNTT ultrathin film indicates the initial closed-monolayer morphology (with a second monolayer seen as yet-disconnected islands). Fig.7.6(b) confirms that this closed-monolayer morphology is perfectly preserved during storage at a cryogenic temperature for 18 days. For comparison, the reference substrate as seen in Fig.7.6(c), which was stored at room temperature in ambient air,

¹The cryogenic cooling and AFM imaging thereafter were performed by Tobias Breuer and Darius Günder at the Philipps-Universität Marburg.

shows the break-up of the initially closed DNTT monolayer into tall, disconnected islands. This effect and similar morphological changes were also observed on the substrate that had been stored at a cryogenic temperature within less than a day after it had been returned to room temperature. These observations strongly support the hypothesis that these morphological changes in the ultrathin DNTT films are indeed thermally activated.

7.3 In-situ encapsulation of ultrathin DNTT films

As discussed previously, the morphological changes in the ultrathin DNTT films proceed by lateral and vertical mass transport that causes a rearrangement of the initially closed (or at least connected) single-monolayer-thick films into disconnected islands. While the results in Fig.7.6 confirm that this mass transport is indeed thermally activated, the results in Fig.7.4(b) and Fig.7.5 show that it can be prevented by covering the ultrathin films with additional material (in the case of Fig.7.5 with additional DNTT). The following sections explore the extent to which this can also be accomplished using encapsulation materials other than DNTT. The choice of encapsulation materials is dictated by the requirement to deposit the encapsulation immediately (within one to two minutes) following the completion of the deposition of the ultrathin DNTT film in the same vacuum-deposition system and without breaking the vacuum. The requirement of encapsulating the ultrathin DNTT films without breaking vacuum is based on the timedelay and subsequent morphological changes in the films during changing the vacuum system, and not related to any influence of vacuum or air exposure on the morphological changes themselves. This ruled out virtually all popular organic-TFT-encapsulation materials, such as Al_2O_3 [276], Cytop [277] and parylene [278], as none of these can be deposited in the vacuum-deposition system in which the DNTT films are deposited. The three materials selected for the in-situ encapsulation of ultrathin DNTT films are polytetrafluoroethylene (PTFE), C₆₀ and titanyl phthalocyanine (TiOPc).

7.3.1 Encapsulation using vacuum-deposited polytetrafluoroethylene (PTFE)

Polytetrafluoroethylene (PTFE) is a synthetic fluoropolymer available as a powder that can be deposited by thermal sublimation in vacuum. PTFE is an excellent electrical insulator and has been used previously to encapsulate pentacene TFTs [279]. PTFE films with a nominal thickness of either 2.5 nm or 10 nm were deposited directly onto ultrathin DNTT films having a nominal thickness of 2.5 nm. The PTFE deposition was started within one to two minutes after completion of the DNTT deposition in the same vacuum-deposition system without breaking vacuum, by using one sublimation source for the DNTT deposition and another for the PTFE deposition.



Figure 7.7: Effect of in-situ encapsulation of ultrathin, vacuum-deposited DNTT films (having a nominal thickness of 2.5 nm) with vacuum-deposited PTFE on the evolution of the TFT characteristics. Transfer curves of TFTs encapsulated with a nominally (a) 2.5 nm-thick and (b) 10 nm-thick PTFE film, measured repeatedly over a period of 5 days after TFT fabrication (c) Evolution of the effective charge-carrier mobility extracted from the measured transfer curves

The results of the TFT measurements are summarized in Fig.7.7. As can be seen, the initial TFT characteristics are not significantly affected by the PTFE encapsulation, but the degradation of the charge-carrier mobility is notably decelerated by the PTFE encapsulation, especially for the larger PTFE thickness (10 nm). However, even with the nominally 10 nm-thick PTFE encapsulation, the charge-carrier mobility of the TFTs still decreases by approximately one order of magnitude (from 0.8 to 0.08 cm²/Vs) within two months. Analysis of thin PTFE films deposited onto surfaces other than DNTT suggests that PTFE films also undergo morphological changes, and assuming these changes also occur in PTFE films deposited onto DNTT, this might explain the limited effectiveness of the PTFE encapsulation observed here. Nevertheless, as it is apparent that increasing the PTFE thickness from 2.5 to 10 nm increases the encapsulation effectiveness, PTFE

encapsulation with a nominal thickness of more than 10 nm was also attempted. However, this turned out to be quite challenging and prohibitively time-consuming in the deposition system, due to the unfavorable combination of the high vapor pressure of the PTFE powder and the limited through-put of the vacuum pump.

7.3.2 Encapsulation using vacuum-deposited C_{60}

As an alternative to PTFE, C_{60} was also tested for the encapsulation of ultrathin DNTT films. The vapor pressure of C_{60} is substantially smaller than that of PTFE, so that thick C_{60} films can be easily deposited by sublimation in vacuum. Other potential benefits of C_{60} for the purpose of encapsulating ultrathin DNTT films are the superior thermal and mechanical stability of C_{60} films [280] and the fact that C_{60} does not form undesirable Diels–Alder adducts with DNTT [281]. A substantial drawback of C_{60} for this purpose is the fact that it is an electron-transporting semiconductor and thus expected to adversely affect the current–voltage characteristics of the DNTT TFTs, especially their off-state behavior, due to the uncontrollable flow of electrons from the drain contact through the C_{60} film to the source contact. C_{60} films with a nominal thickness of either 2.5 nm or 25 nm were directly deposited onto ultrathin DNTT films having a nominal thickness of 2.5 nm.



Figure 7.8: Effect of in-situ encapsulation of ultrathin, vacuum-deposited DNTT films (having a nominal thickness of 2.5 nm) with vacuum-deposited C_{60} on the evolution of the TFT characteristics. Transfer curves of TFTs encapsulated with a nominally (a) 2.5 nm-thick and (b) 25 nm-thick C_{60} film, measured repeatedly over a period of 5 days after TFT fabrication (c) Evolution of the effective charge-carrier mobility extracted from the measured transfer curves
The results are summarized in Fig.7.8. As can be seen in Fig.7.8(b), the deposition of a nominally 25-nm thick film of C_{60} indeed causes ambipolar behavior, which is highly undesirable, as it increases the off-state leakage current of the TFTs and thus the static power consumption and the noise margins of logic circuits based on ambipolar TFTs. However, the degradation of the hole mobility in the ultrathin DNTT film is notably decelerated by the C_{60} encapsulation, which suggests that the morphology of the ultrathin DNTT film is indeed greatly stabilized by the C_{60} encapsulation. A workable compromise between the beneficial effect of stabilizing the morphology of the ultrathin DNTT films and the drawback of increasing the leakage current is the use of a C_{60} encapsulation with a smaller thickness, as shown in Fig.7.8(a). With a nominally 2.5 nm-thick C₆₀ encapsulation, the off-state drain current of the ultrathin-DNTT TFTs is approximately 10⁻¹⁰ A, which is similar to that of ultrathin-DNTT TFTs without encapsulation (Fig.7.4(a)) or with an electrically insulating encapsulation (Fig.7.7(a)). However, the suppression of the degradation of the hole mobility in the ultrathin DNTT film is not quite as pronounced as in the case of the nominally 25 nm-thick C_{60} encapsulation (Fig. 7.8(c)). This is mainly due to the fact that the deposition of nominally 2.5 nm-thick C_{60} does not lead to the formation of a closed C_{60} film on the surface of the ultrathin DNTT film, but rather to edge decoration (similar to previous reports for C_{60} deposited on pentacene [282]).

7.3.3 Encapsulation using vacuum-deposited titanyl phthalocyanine (TiOPc)

TiOPc is a hole-transporting small-molecule semiconductor [283]. TiOPc molecules usually adopt a flat-lying orientation when deposited onto metals [284], but an uprightstanding molecular orientation when deposited onto electrically-insulating surfaces or organic-semiconductor films [285]. A potential benefit of TiOPc for the purpose of encapsulating ultrathin DNTT films is its excellent chemical robustness and thermal stability. A drawback of TiOPc for this purpose is that the possibility of hole transport in the TiOPc film in parallel to the ultrathin DNTT film cannot be ruled out. This makes it impossible to monitor the electrical behavior of the DNTT film exclusively. TiOPc films with a nominal thickness of either 2.5 or 25 nm were deposited directly onto ultrathin DNTT films. The results are summarized in Fig.7.9.

The first observation is that the current-voltage characteristics of the ultrathin-DNTT

TFTs with the TiOPc encapsulation (Fig.7.9(a) and (b)) are exactly the same as those of the ultrathin-DNTT TFTs without encapsulation (Fig.7.5(a)).



Figure 7.9: Effect of in-situ encapsulation of ultrathin, vacuum-deposited DNTT films (having a nominal thickness of 2.5 nm) with vacuum-deposited TiOPc on the evolution of the TFT characteristics. Transfer curves of TFTs encapsulated with a nominally (a) 2.5 nm-thick and (b) 25 nm-thick TiOPc film, measured repeatedly over a period of 5 days after TFT fabrication (c) Evolution of the effective charge-carrier mobility extracted from the measured transfer curves

This suggests that the TiOPc film does not participate in the charge transport, which is in contradiction to the results reported previously for TFTs based on a heterostructure of ultrathin dihexyl-sexithiophene and TiOPc, but this may be due to the fact that the intrinsic carrier mobility in DNTT is significantly larger than that in both TiOPc and dihexyl-sexithiophene. The second observation is that the TiOPc encapsulation slows down the degradation of the charge-carrier mobility of the TFTs, similar to encapsulation with PTFE and C₆₀. However, in stark contrast to PTFE (Fig.7.7) and C₆₀ (Fig.7.8), this effect is virtually independent of the thickness of the TiOPc encapsulation. Nevertheless, the charge-carrier mobility of the TiOPc-encapsulated TFTs still decreases by approximately one order of magnitude within two weeks.

7.3.4 Summary of encapsulation results

The results from the encapsulation experiments are summarized in Fig.7.10. The transfer curves of the TFTs with and without the various encapsulation films recorded within

30 minutes after the last deposition are summarized in Fig.7.10(a). The TFTs with the thick DNTT films (nominal thickness 25 nm) initially have an effective carrier mobility of 1 cm²/Vs (Fig.7.10(b)), which is similar to previous reports of bottom-gate, bottom contact DNTT TFTs [257]. The effective mobility of these TFTs degrades from 1 to 0.8 cm²/Vs within 5 days and to 0.5 cm²/Vs within 23 days (as also seen in Fig.7.5(c)).



Figure 7.10: (a) Transfer curves of TFTs based on vacuum-deposited DNTT films with nominal thicknesses of 2.5 and 25 nm with and without encapsulation, recorded within 30 minutes after the completion of the last deposition. (b) Absolute value of the effective carrier mobility (c) Effective carrier mobility normalized to the mobility measured immediately after TFT fabrication

This degradation in effective mobility is notably more rapid than the mobility degradation in top-contact DNTT TFTs, but for the experiments described here, it was necessary to fabricate the TFTs in the bottom-contact architecture in order to minimize the delay between the completion of the DNTT deposition and the first measurement of the current–voltage characteristics. The long-term stability of bottom-contact DNTT TFTs has yet to be systematically investigated. The reasons for the relatively rapid degradation of the effective mobility observed here is not known, but it may be related to a degradation of the interface between the source/drain contacts and the vacuum-deposited DNTT film.

The TFTs with the ultrathin DNTT films (nominal thickness 2.5 nm) initially have effective carrier mobilities between 0.3 and 0.4 $\rm cm^2/Vs$, regardless of whether they were encapsulated or not and, if so, regardless of the encapsulation material (PTFE, C₆₀, TiOPc) and the nominal thickness of the encapsulation film (Fig.7.10(b)). The only exception is the TFT in which the ultrathin DNTT film was encapsulated with a nominally 10 nm-thick PTFE film, which has a somewhat larger initial mobility $(0.7 \text{ cm}^2/\text{Vs})$. The reason for this discrepancy is unknown, but it may be the result of unavoidable substrate-to-substrate variations in the fabrication process. The results confirm that it is indeed possible to fabricate functional TFTs based on vacuum-deposited DNTT films with a nominal thickness of less than two molecular monolayers. Encapsulation of these films has a pronounced effect on their stability: while the effective mobility of unencapsulated ultrathin DNTT films decreases by two orders of magnitude within a day, the mobility of encapsulated films decreases by less than one order of magnitude within five days (Fig.7.10(c)). The mechanism by which the encapsulation of the ultrathin DNTT films decelerates the degradation of the carrier mobility in these films is not by protecting the DNTT molecules from ambient air (given the outstanding chemical stability of DNTT during long-term exposure to air and humidity), but by stabilizing the fragile morphology of the ultrathin DNTT films by burying it under a thick film of additional mass. Although it is not possible to inspect the morphology of the aged ultrathin DNTT films buried underneath the thick encapsulation films by AFM measurements, it can be hypothesized that the deceleration of the mobility degradation observed here is a direct result of preventing mass transport within the ultrathin DNTT films by the encapsulation film. This hypothesis is supported by the observation that for two of the three encapsulation materials, thicker encapsulation films provide better stability (Fig. 7.10(c)). The interesting question for how long the encapsulation can delay the break-up of the initially closed DNTT monolayer is beyond the scope of the present study.

The choice of the encapsulation material is dictated by a number of considerations. Since the spontaneous structural reconfiguration of the ultrathin DNTT films commences immediately after the DNTT deposition, it is necessary to start the deposition of the encapsulation film immediately after the completion of the DNTT deposition. This limits the choice of encapsulation materials to such materials that can be deposited in the same high-vacuum system in which the ultrathin DNTT films are deposited, without breaking vacuum and without time-consuming substrate transfers. If the purpose of the experiments is an investigation of the charge transport in ultrathin DNTT films, the choice of encapsulation materials should be limited to materials that do not participate in the charge transport. Of the three materials that have been tested here, this obviously eliminates C_{60} (and possibly also TiOPc, although Fig.7.9 seems to indicate that the TiOPc films do not participate in charge transport). The observation that the stabilizing effect of the encapsulation increases with increasing thickness of the encapsulation film calls for materials that can be easily deposited as thick films. Unfortunately, this essentially eliminates PTFE, due to its unfavorably high vapor pressure.

Another aspect to be considered in the choice of the encapsulation material is the homogeneity and long-term morphological stability of the encapsulation film, which would likely benefit from choosing materials with a higher molecular weight and thus smaller diffusivity on the DNTT surface, and possibly also from performing the deposition of the encapsulation film at cryogenic temperatures to promote homogeneous coverage of the encapsulating material.

7.4 Experimental

7.4.1 Film characterization

Samples for film characterization were prepared on silicon substrates. Silicon was chosen as the substrate to provide the smallest possible surface roughness. Aluminum with a thickness of 30 nm and a root-mean-square surface roughness of less than 1 nm [197] was deposited by thermal evaporation in vacuum at a rate of 2 nm/s. The aluminum surface was exposed to a capacitively coupled oxygen plasma with a partial oxygen pressure of 0.01 mbar and an oxygen flow rate of 30 sccm. The plasma power selected was 200 W and the duration of plasma exposure was limited to 30 s to increase the thickness of the native aluminum oxide (AlO_x) to approximately 6 nm. On some substrates, ultrathin DNTT (Sigma-Aldrich) was deposited directly onto bare AlO_x. On other substrates, a monolayer of HC₁₄PA, HC₆PA, or HC₁₈PA (PCI Synthesis, Newbury- port, MA, USA) was allowed to self-assemble on the AlO_x surface by immersing the substrates into a 1 mM-solution of the phosphonic acid in 2-proponal for a duration of 12 hours, after which the substrates were rinsed with 2-proponal, dried with nitrogen and heated to a temperature of 80 °C for 10 minutes to stabilize the monolayer. DNTT was deposited by thermal sublimation in vacuum at a rate of 2 nm/min (without heating the substrate to avoid the unnecessary acceleration of the morphological changes). AFM images were acquired in tapping mode (in ambient air at room temperature) using a Bruker Nanoscope III MultiModes or an Agilent SPM 5500 and cantilevers with resonance frequencies between 200 and 500 kHz and a typical tip radius of 8 nm.

7.4.2 TFT fabrication and characterization

TFTs were fabricated in the inverted coplanar (bottom-gate, bottom-contact) architecture on silicon substrates coated with 100 nm-thick thermally grown silicon dioxide. This device architecture was chosen to minimize the delay between the completion of the semiconductor deposition and the first measurement of the current-voltage characteristics. Aluminum gate electrodes with a thickness of 30 nm were deposited by thermal evaporation in vacuum at a rate of 2 nm/s. The gate dielectric consists of a thin, oxygen-plasma-grown film of AlO_x and a solution-processed HC₁₄PA SAM and has a capacitance of about 0.7 μ F/cm². For the source and drain contacts, titanium with a thickness of 0.2 nm (to improve adhesion) and gold with a thickness of 30 nm were sequentially deposited by thermal evaporation in vacuum. To minimize the contact resistance, the surface of the Au source/drain contacts was functionalized with a monolayer of pentafluorobenzenethiol (PFBT). DNTT was deposited by thermal sublimation in vacuum (base pressure 10^{-6} mbar; substrate temperature 20 °C) at a rate of 2 nm/min. PTFE (Sigma-Aldrich, free- flowing powder, particle size < 12 mm), C₆₀ (Sigma-Aldrich) or TiOPc (Sigma-Aldrich) encapsulation films were deposited by thermal sublimation in vacuum within one to two minutes after completion of the DNTT deposition. Deposition rate and nominal thickness of all vacuum-deposited films were measured using a quartz-crystal microbalance positioned in close proximity to the substrate holder. Gate electrodes, source/drain contacts, organic semiconductor and encapsulation films were all patterned using polyimide shadow masks (CADiLAC Laser, Hilpoltstein, Germany). The TFTs have a channel length of 20 μ m and a channel width of 100 μ m. The electrical measurements were performed in ambient air at room temperature using a manual probe station and an Agilent 4156C Semiconductor Parameter Analyzer. The first measurement was performed within 15 minutes after the completion

of the last vacuum deposition, and the substrates were stored in ambient air at room temperature between measurements.

7.5 Summary and Outlook

The thickness of the organic-semiconductor layer is an important parameter towards understanding the operation of organic TFTs and their further development. Ultrathin films of organic semiconductors are particularly significant, as the conductive channel in organic TFTs is limited to the first few monolayers of the semiconductor, in close proximity to the gate dielectric. The spontaneous occurring morphological changes in the thin-film morphology of vacuum-deposited ultrathin DNTT films have been observed by AFM. The influence of this spontaneous structural reconfiguration on the electrical characteristics of TFTs based on such films has been analyzed. The results confirm that the observed transition from an initially closed (or at least connected) monolayer to tall, isolated islands and the associated loss of percolation through the films cause a rapid and substantial degradation of the effective charge-carrier mobility of the TFTs. Of the available gate-dielectric surfaces studied in this thesis, including bare-AlO_x and several different phosphonic acid SAMs, a similar progression of the morphological changes can be seen in ultrathin DNTT films. Whether the structural reconfiguration of ultrathin DNTT films has to be prevented by changing the type of the substrate, gate-dielectric materials with functional groups similar to the chemical structure of the organic semiconductor will have to be considered. For example, phosphonic acid molecules containing benzyl functional groups will potentially increase the substrate-molecule interactions and possibly stabilize the ultrathin DNTT films.

In this work, the morphological changes observed in ultrathin DNTT films were controlled using cryogenic cooling and in-situ encapsulation. By showing that the morphological changes are prevented by keeping the substrates at cryogenic temperatures, it has been confirmed that this structural reorganization is indeed thermally activated. The extent to which encapsulation of ultrathin DNTT films is able to stabilize the morphology has been tested using three different materials: PTFE, C_{60} and TiOPc. Although the morphological changes were not stopped entirely, it has been shown that in-situ encapsulation can slow the rate of degradation quite significantly. The ability to fabricate TFTs based on vacuum-deposited ultrathin organic–semiconductor films with sufficient long-term stability is potentially of interest for interface and charge-transport investigations. Future investigations could lead to the further improvement of the stability and performance of ultrathin organic TFTs. These include investigating other materials as potential encapsulation materials which meet the requirements of encapsulation in ultrathin organic TFTs and which provide a higher stability to the ultrathin semiconductor films. The ultrathin organic TFTs in this work were fabricated in a bottom-gate, bottom-contact (inverted coplanar) configuration, in order to encapsulate the ultrathin DNTT films, while still maintaining contact with the source and drain contacts. In the currently available device fabrication techniques, it is not possible to fabricate ultrathin organic TFTs in the top-contact configuration with simultaneous contact with the electrodes as well as encapsulation. Since the inherent stability of top-contact organic TFTs is higher than that of bottom-contact organic TFTs, top-contact TFTs with a new fabrication technique and design in which the source-drain contacts are deposited sequentially at the end and are still in contact with the ultrathin semiconductor films with in-situ encapsulation would further improve the stability of ultrathin organic TFTs. These efforts would be advantageous towards the development of highly sensitive chemical or biological sensors due to the benefits of ultrathin organic TFTs.

Chapter 8

Conclusion

Organic semiconductors are a widely researched class of materials due to their unique properties such as low-temperature processing and targeted synthesis to develop electronic devices such as organic light-emitting diodes and organic thin-film transistors (TFTs). Organic TFTs in particular, are being developed for flexible electronics applications, such as rollable displays, conformable sensors and electronic skin. Among the many electrical characteristics, an important requirement for organic TFTs is lowvoltage operation, for the realization of portable or wearable applications. The operating voltage of the TFT is determined by the gate-dielectric capacitance, and ultrathin, hybrid gate dielectrics are an excellent choice for high-capacitance gate dielectrics. Hybrid gate dielectrics are extremely advantageous in simultaneously achieving a low operating voltage as well a a low leakage current in organic TFTs. This thesis investigates hybrid gate dielectrics with two components, an ultrathin aluminum oxide (AlO_x) film in combination with an organic self-assembled monolayer (SAM). In this thesis, the different components of the hybrid gate dielectric, the organic semiconductor layer as well as the gate dielectric-semiconductor interface have been investigated and their material properties have been correlated to the electrical properties of organic TFTs.

The first component of the hybrid gate dielectric, the AlO_x film, is fabricated by exposing the surface of the underlying aluminum gate electrode to a capacitively-coupled oxygen plasma in a self-limiting growth process. The plasma-oxidation process has several benefits for the fabrication of organic TFTs, including but not limited to producing a self-patterned, ultrathin and dense AlO_x film with a high-quality interface with the underlying aluminum gate electrode. In the absence of the plasma-oxidation process,

the native aluminum oxide on the underlying aluminum gate is an insufficient and inadequate gate oxide and leads to organic TFTs with suboptimal electrical characteristics. The AlO_x film has several critical roles in the hybrid gate dielectric and the overall electrical performance of organic TFTs. By virtue of the plasma process, the AlO_x film provides a suitable surface for the SAM to adsorb onto and thus ensures a low leakage current through the gate dielectric.

The thin-film properties of the plasma-oxidized AlO_x film, namely the thickness and the chemical composition were investigated. The thickness of the film was measured using transmission electron microscopy and was varied between 4 nm and 7 nm, depending on the plasma power and the duration of plasma exposure during the fabrication process. In particular, for a plasma power of 200 W and a plasma duration of 30 s, the thickness of the AlO_x film was determined to be about 6.5 nm. By measuring the thickness and the gate-dielectric capacitance of different AlO_x films fabricated using different process parameters, the relative permittivity of the plasma-oxidized AlO_x was determined to be approximately 8. These properties are crucial to the dielectric properties of the AlO_x films, and a process to fabricate AlO_x films with a particular thickness makes it possible to deterministically control the gate-dielectric capacitance, and in turn the operating voltage of the TFT. In this work, the gate-dielectric capacitance of AlO_x gate dielectrics was varied between 1 and 1.6 $\mu F/cm^2$ by varying the plasma power and plasma duration. The fabrication of AlO_x films thinner than 4 nm or thicker than 7 nm, by further varying the plasma-process parameters or other aspects of the plasma process, might potentially grant a better and more precise control of the gate-dielectric capacitance and the operating voltage.

Along with the thickness of the AlO_x film, the thin-film composition was also investigated. Using electron energy loss spectroscopy, the presence of elemental aluminum and oxygen was spatially mapped in an Al/AlO_x layer configuration. With the help of atom probe tomography, a three-dimensional reconstruction of the Al/AlO_x films was obtained with an atomic-scale resolution. Based on these measurements, the thin-film composition can be qualitatively assessed and resembles stoichiometric Al_2O_3 films. However, further efforts with the measurement parameters of these techniques, or alternative analytical ones, are required to quantitatively determine the chemical composition of the plasma-oxidized AlO_x films. Fabricating perfectly stoichiometric Al_2O_3 films with the plasma-oxidation process will ideally improve the electrical characteristics of organic TFTs. Apart from the thickness of the AlO_x films, the oxygen-plasma process was instrumental in tailoring several other material and thin-film properties, which influence the electrical properties of organic TFTs. The most important impact of the plasma power and the plasma duration was observed on the surface roughness of the AlO_x films. An optimum surface roughness, of approximately 0.52 nm, was obtained for a combination of medium plasma powers (100-200 W) and medium plasma durations (30-60 s), which facilitated an optimum organic-semiconductor morphology and high charge-carrier mobilities in the organic TFTs. The exact mechanism behind modifying the surface roughness of the metal oxide film using an oxygen plasma will need further investigation. It was also observed that long plasma exposure times (≥ 1800 s) lead to thicker AlO_x films with a damaged surface unsuitable for organic semiconductors to be deposited onto, despite a lower surface roughness. Further characterization of these films beyond the parameter of surface roughness would improve the understanding of the influence of the plasmaoxidation process on the surface properties of AlO_x films.

Self-assembled monolavers (SAMs) are molecular assemblies of individual molecules which adsorb on a suitable surface and align themselves into a monolayer film through Van der Waals interactions. SAMs have been a popular choice in hybrid gate dielectrics and provide a variety of advantages for the properties of organic TFTs. Phosphonic acid molecules of two different aliphatic chains, an unsubstituted alkyl chain and a functionalized fluoroalkyl chain, were assembled from solution to form the SAM on the plasma-oxidized AlO_x films. Several benefits of employing SAMs in the hybrid gate dielectric were observed throughout this work. A leakage current density of the order of 10^{-11} A was achieved for organic TFTs operating at a supply voltage as low as 3 V. Compared to organic TFTs with bare- AlO_x gate dielectrics, the TFTs with hybrid AlO_x/SAM gate dielectrics had higher charge-carrier mobilities, due to the contribution of the SAM in screening polaronic charges and providing a suitable surface for the growth of the organic semiconductor. The organic semiconductor DNTT was used to fabricate TFTs with alkyl as well as fluoroalkyl phosphonic acid SAMs and a threshold-voltage shift was observed in the organic TFTs with the two different SAMs. The fluoroalkyl phosphonic acid SAMs were used in combination with the organic semiconductor DPh-BTBT for fabricating ultralow-voltage organic TFTs.

The most important aspect of the SAMs studied in this work is the thickness of the SAM as a function of the chain length of the phosphonic acid molecule. By employing molecules of different chain lengths, electrical characteristics of the organic TFTs such as the gate-dielectric capacitance, the charge-carrier mobility, the subthreshold swing and the current on-off ratio were modified. In particular, the chain length of the phosphonic acid molecules was instrumental in modifying the gate-dielectric capacitance of hybrid gate dielectrics, which is not possible by varying the thickness of the AlO_x film. An optimum electrical performance in terms of the TFT characteristics was obtained for the medium-chain-length phosphonic acids of the available range of molecules, i.e. $HC_{10}PA-HC_{14}PA$ and $FC_{10}PA$ SAMs.

With the formation of the SAM on the AlO_x surface, the surface energy of the hybrid gate dielectric can be modified, and the hydrophilic AlO_x surface can be modified into a preferable, hydrophobic surface. In addition, the surface energy of the hybrid gate dielectric can be further varied as a function of the type of the phosphonic acid molecule as well as its chain length. Apart from the surface energy, the properties of the SAM as a function of the chain length affect the growth and morphology of the organic semiconductor deposited onto the surface. Although no direct correlation was made with the growth pattern of the semiconductor DNTT and the surface energy of the phosphonic acid SAMs, it was identified that an optimum growth mode is obtained for DNTT deposited on the HC₁₄PA SAM, i.e. an island-layer type of growth with maximum semiconductor coverage obtained in the initial monolayers during the growth process. A preferred growth mode of the semiconductor as well as a large density of grain boundaries in the thin-film morphology were correlated with a large charge-carrier mobility in the organic TFTs.

The influence of the surface properties at the gate dielectric-semiconductor interface, on the film properties of the organic semiconductor are thus established. Depending on the surface roughness of the AlO_x film, the surface energy of the SAM and the type of the phosphonic acid molecule forming the SAM, an optimum growth mode and morphology of the organic semiconductor was determined and correlated with high charger-carrier mobilities in the organic TFTs. A further investigation of the film properties of the SAM and the hybrid gate dielectric, beyond the surface roughness and the surface energy, is required. Particularly, the packing density and molecular ordering of individual molecules within the SAM will promote the deposition of high-quality organic semiconductor films. Extending the observations made for the organic semiconductor DNTT in terms of the optimum SAM to other organic semiconductors will also benefit the development of organic TFTs.

The part of the semiconductor layer in close proximity to the gate dielectric is extremely

crucial because the conducting channel in the transistor is located in this region. Ultrathin films of the organic semiconductor, of the order of one or two molecular monolayers, were investigated. It was observed that ultrathin films undergo rapid structural reorganization and morphological changes in which an initially closed or connected monolayer develops into individual, disconnected islands over time. Functional organic TFTs with ultrathin organic semiconductor films were fabricated, and a drastic decrease in the charge-carrier mobility with time was observed and correlated with the morphological changes observed. The ultrathin semiconductor films were partially stabilized by cryogenic cooling and in-situ encapsulation using different encapsulation materials, which showed that encapsulation is a promising concept and investigating other encapsulation materials along with exploring new device configurations will further improve the stability of ultrathin organic TFTs and their development for a variety of sensors and studying device physics.

Due to the construction of organic TFTs with different materials deposited in the form of thin films and a variety of material combinations, fabrication methods and process parameters, it is extremely important to establish a relationship between the material properties of the different components and the electrical properties of the organic TFTs. The gate dielectric, the organic semiconductor and the interface between the two has been studied in detail in this thesis. The essential relationship between the thin-film and surface properties of the gate dielectric, morphology of the organic semiconductor and the electrical properties of the organic TFT was established. Extending the understanding of this relationship to other materials and processes will advance the development of organic thin-film transistors and promote the realization of unique electronic applications.

Appendix A

Additional characterization of DNTT films



Figure A.1: AFM image depicting the morphology of a nominally 2.5-nm-thick DNTT film deposited onto a hybrid $AlO_x/HC_{14}PA$ SAM gate dielectric, and the height profile indicating a characteristic step height of 1.6 nm, confirming that the DNTT molecules stand approximately upright on the substrate surface



Figure A.2: Specular X-ray diffractogram of a nominally 10-nm-thick DNTT film deposited onto a hybrid $AlO_x/HC_{14}PA$ SAM gate dielectric together with a calculated powder spectrum. The inset schematically depicts the molecular arrangement



Figure A.3: Output and transfer curves and charge-carrier mobility as a function of gate-source voltage of (a) a TFT with a nominally 2.5-nm thick DNTT film; (b) a TFT with a nominally 25-nm thick DNTT film. The transfer curves were recorded within 15 minutes after the completion of the DNTT deposition

Appendix B

Optimizing plasma oxidation for aluminum oxide films

In Chapter 4, the influence of the plasma power and duration of plasma during the fabrication of AlO_x films on the electrical performance of organic TFTs was observed. The transfer characteristics, semiconductor morphology and other properties were show-cased for certain combinations of the plasma-process parameters. In this Appendix, the transfer characteristics, output characteristics and the morphology of DNTT films for all 15 combinations tested are given.



Figure B.1: Transfer characteristics and gate currents of DNTT TFTs fabricated using either a bare-AlO_x gate dielectric (red curves) or a hybrid AlO_x/HC₁₄PA SAM gate dielectric (blue curves) for all fifteen combinations of the plasma power and plasma duration



Figure B.2: Output characteristics of DNTT TFTs fabricated using a hybrid $AlO_x/HC_{14}PA$ SAM gate dielectric for all fifteen combinations of plasma power and plasma duration



Figure B.3: AFM images of DNTT films deposited onto a hybrid $AlO_x/HC_{14}PA$ SAM gate dielectric for all fifteen combinations of plasma power and plasma duration and effective charge-carrier mobilities extracted from the measured transfer characteristics of DNTT TFTs fabricated using a hybrid $AlO_x/HC_{14}PA$ SAM gate dielectric as a function of plasma power and plasma duration. The high charge-carrier mobilities can be correlated with an optimum semiconductor morphology, i.e. a closed, connected film with a distinct dendritic structure

Appendix C

Optimizing the substrate temperature during semiconductor deposition for different substrates

The organic semiconductor DPh-BTBT was deposited by thermal sublimation in vacuum with the provision of heating the substrate during the semiconductor deposition process. Conventionally, substrates are heated to an optimum temperature to facilitate the diffusion of semiconductor molecules on the surface and form polycrystalline thin films for maximum carrier mobility and an enhanced electrical performance. However, increasing the substrate temperature (T_{sub}) beyond the optimum may result in island-mode growth or desorption of semiconductor molecules from the surface. A striking difference is observed between DPh-BTBT TFTs fabricated on silicon and on PEN substrates in the way the morphology of the DPh-BTBT layer and the TFT performance are affected by the substrate temperature during the semiconductor deposition. As seen in Fig.C.1(a-c), the TFTs fabricated on silicon have carrier mobilities of 0.56 cm²/Vs when the substrate is held at a temperature of 25 °C during the DPh-BTBT deposition and 1.1 cm²/Vs when the DPh-BTBT is deposited at a substrate temperature of 100 °C. This is the expected behavior for organic semiconductors deposited by thermal sub-limation where a higher substrate temperature promotes larger grain size and enhanced

crystallinity in the thin film. In contrast, for DPh-BTBT TFTs fabricated on PEN, Fig.C.1(d-f) shows that heating the substrate during the DPh-BTBT deposition to a temperature of 100 °C yields a carrier mobility that is 3 orders of magnitude lower than that obtained without substrate heating.



Figure C.1: Transfer curves of DPh-BTBT TFTs fabricated on silicon substrates with the substrate held at temperatures of (a) 25°C and (b) 100°C during the semiconductor deposition and (c) carrier mobilities extracted from the transfer curves. Transfer curves of DPh-BTBT TFTs fabricated on flexible PEN substrates with the substrate held at temperatures (d) 25°C and (e) 100°C during the semiconductor deposition and (f) carrier mobilities extracted from the transfer curves

The thin-film morphologies as seen in the scanning electron microscopy (SEM) images show that when the PEN substrate is heated during the semiconductor deposition, the DPh-BTBT molecules form isolated islands (Fig.C.2(c), inset) instead of a continuous film formed when the PEN substrate is held at 25 °C(Fig.C.2(d), inset). Since a percolation path for the molecules is harder to form through isolated islands, the carrier mobility is significantly lower. For silicon substrates held at 100 °C(Fig.C.2(a), inset) and 25 °C(Fig.C.2(b), inset), the thin-film morphologies indicate thin films with almost complete coverage in both cases, thereby establishing that the substrate temperature has little influence on the thin-film morphology of DPh-BTBT films on silicon substrates. Out-of-plane XRD measurements (Fig.C.2) performed on the DPh-BTBT films on PEN and silicon substrates indicate the possible orientation of the DPh-BTBT molecules, establishing the influence of the substrate temperature on the thin-film morphology for the two different substrates. On the PEN substrate for the films deposited at a substrate temperature of 25°C, the DPh-BTBT molecules are mostly in an upright-standing orientation, with a dominant diffraction peak at $2\theta = 4.65^{\circ}$ that corresponds to a (001) interlayer spacing of 19 Å. The appearance of the peaks corresponding to (002) and (003) further indicates a good layer-by-layer structure of the film. On the other hand, the DPh-BTBT films deposited on PEN at a substrate temperature of 100°C are characterized by a mix of upright-standing and lying-down orientation of the molecules, with peaks appearing at $2\theta = 4.65^{\circ}$, 23.8°, and 27.1° that correspond to the (001), (020), and (120) orientations, respectively. The intensity of the (001) signal is much smaller than that measured on the film deposited at a substrate temperature of 100°C. Only the semiconductor molecules that are standing upright contribute to the lateral carrier transport, which explains why the carrier mobility is significantly larger when the semiconductor deposition is carried out at a substrate temperature of 25°C, rather than 100°C. In contrast, it is observed that the XRD spectra of the DPh-BTBT films on the silicon substrates are very similar for the two substrate temperatures, with a single dominant peak at $2\theta = 4.65^{\circ}$, corresponding to molecules in an upright-standing configuration. Since this orientation is favorable to lateral carrier transport, this also explains the carrier mobilities of 0.56 and 1.1 cm²/Vs for DPh-BTBT TFTs on silicon substrates for substrate temperatures of 25°C and 100°C, respectively, and the fact that the transistor characteristics depend only weakly on the substrate temperature during the semiconductor deposition.

A similar trend in the DPh-BTBT morphology can be seen in Fig.C.3, in which the underlying surface was a bare-AlO_x gate dielectric. On the silicon substrate, at a substrate temperature of 100°C, DPh-BTBT molecules exhibit a closed, connected thin-film morphology with XRD data indicating molecules in a favorable upright-standing configuration. On the PEN substrate, at an elevated temperature of 100 °C, the DPh-BTBT thin-film morphology is suboptimal with many holes and cracks seen in the SEM image. However, the XRD data indicates that the molecules are in a favorable orientation at both substrate temperatures.



Figure C.2: Out-of-plane XRD spectra and corresponding scanning electron microscopy images of DPh-BTBT thin films deposited onto hybrid $AlO_x/FC_{10}PA$ SAM dielectrics on silicon substrates held at temperatures of (a) 100°C (red line) and (b) 25°C (black line) and on PEN substrates held at temperatures of (c) 100°C (red line) and (d) 25°C (black line) during the semiconductor deposition



Figure C.3: Out-of-plane XRD spectra and corresponding scanning electron microscopy images (insets) of DPh-BTBT thin-films deposited onto bare AlO_x dielectrics on silicon substrates held at temperatures of 100°C (red line, (a)) and 25°C (black line, (b)) and on PEN substrates held at a temperature of 100°C (red line, (c)) and 25°C (black line, (d)) during the semiconductor deposition.

Appendix D

Optimizing combinations of the organic semiconductor/phosphonic acid SAM



Figure D.1: Transfer characteristics of DNTT TFTs with different gate dielectrics (a) bare-AlO_x (b) AlO_x/HC₁₄PA SAM (c) AlO_x/HC₁₈PA SAM and (d) AlO_x/FC₁₄PA SAM



Figure D.2: (a) Carrier mobilities and (b) turn-on voltages of DPh-BTBT and DNTT TFTs with hybrid AlO_x/SAM gate dielectrics based on fluoroalkylphosphonic acids with chain lengths of 6, 8, 10, 12, and 14 carbon atoms fabricated on silicon substrates. (c) Transfer curves of five DPh-BTBT TFTs with a hybrid $AlO_x/FC_{10}PA$ SAM gate dielectric fabricated on a silicon substrate, showing a turn-on voltage of exactly 0 V. The substrate was held at a temperature of 100°C during the DPh-BTBT deposition and at a temperature of 60°C for the DNTT deposition



Figure D.3: Out-of-plane XRD spectra measured on DPh-BTBT thin films deposited onto three different hybrid AlO_x/SAM dielectrics on PEN substrates



Figure D.4: Transfer curves of five DPh-BTBT TFTs each with a hybrid AlO_x/SAM gate dielectric based on fluoroalkyl phosphonic acids with five different chain lengths, along with a bare AlO_x gate dielectric (without SAM) fabricated on silicon substrates with substrate temperature at 100°C during semiconductor deposition

Bibliography

- [1] J. E. Lilienfeld, Method and apparatus for controlling electric currents, Patent number US1745175, (1930).
- [2] O. Heil, Improvements in or relating to electrical amplifiers and other control arrangements and devices, Patent number GB439457, (1935).
- [3] J. Bardeen and W. H. Brattain, The transistor, a semi-conductor triode, *Phys. Rev.*, 74, (1948), 230.
- [4] W. Shockley, The theory of p-n junctions in semiconductors and p-n junction transistors, *Bell Syst. Tech. J.*, 28, (1949), 435.
- [5] M. M. Atalla, Junction formation by thermal oxidation of semiconductive material, Patent number US2953486, (1960).
- [6] D. Kahng, Electric field controlled semiconductor device, Patent number US3102230, (1963).
- [7] M. M. Atalla, Semiconductor devices having dielectric coatings, Patent number US3206670, (1965).
- [8] P. K. Weimer, The TFT-A new thin-film transistor, Proc. IRE, 50, (1962), 1462.
- [9] T. P. Brody and H. E. Kunig, A high-gain InAs thin-film transistor, Appl. Phys. Lett., 9, (1966), 259.
- [10] T. P. Brody, J. A. Asars, and G. D. Dixon, A 6x6 inch 20 lines-per-inch liquid crystal display panel, *IEEE Trans. Electron Devices*, 20, (1973), 995.
- [11] P. G. leComber, W. E. Spear, and A. Ghaith, Amorphous-silicon field-effect device and possible application, *Electronics Letters*, 15, (1979), 179.

- [12] P. D. C. King and T. D. Veal, Conductivity in transparent oxide semiconductors, J. Phys. Condens. Matter, 23, (2011), 334214.
- [13] E. Fortunato, P. Barquinha, and R. Martins, Oxide semiconductor thin-film transistors: A review of recent advances, Adv. Mater., 24, (2012), 2945.
- [14] J. S. Park, W. J. Maeng, H. S. Kim, and J. S. Park, Review of recent developments in amorphous oxide semiconductor thin-film transistor devices, *Thin Solid Films*, 520, (2012), 1679.
- [15] D. Jariwala, V. K. Sangwan, L. J. Lauhon, T. J. Marks, and M. C. Hersam, Emerging device applications for semiconducting two-dimensional transition metal dichalcogenides, ACS Nano, 8, (2014), 1102.
- [16] Q. H. Wang, K. Kalantar-Zadeh, A. Kis, J. N. Coleman, and M. S. Strano, Electronics and optoelectronics of two-dimensional transition metal dichalcogenides, *Nature Nanotechnol.*, 7, (2012), 699.
- [17] X. Duan, C. Wang, A. Pan, R. Yu, and X. Duan, Two-dimensional transition metal dichalcogenides as atomically thin semiconductors: Opportunities and challenges, *Chem. Soc. Rev.*, 44, (2015), 8859.
- [18] A. P. Graham, G. S. Duesberg, W. Hoenlein, F. Kreupl, M. Liebau, R. Martin, B. Rajasekharan, W. Pamler, R. Seidel, W. Steinhoegl, and E. Unger, How do carbon nanotubes fit into the semiconductor roadmap?, *Appl. Phys. A*, 80, (2005), 1141.
- [19] V. N. Popov, Carbon nanotubes: properties and application, *Mater. Sci. Eng. R*, 43, (2004), 61.
- [20] X. Z. Bo, C. Y. Lee, M. S. Strano, M. Goldfinger, C. Nuckolls, and G. B. Blanchet, Carbon nanotubes-semiconductor networks for organic electronics: The pickup stick transistor, *Appl. Phys. Lett.*, 86, (2005), 182102.
- [21] W. Lu and C. M. Lieber, Semiconductor nanowires, J. Phys. D: Appl. Phys., 39, (2006), R387.
- [22] W. Lu, P. Xie, and C. M. Lieber, Nanowire transistor performance limits and applications, *IEEE Trans. Electron Devices*, 55, (2008), 2859.

- [23] O. Hayden, R. Agarwal, and W. Lu, Semiconductor nanowire devices, *Nano Today*, 3, (2008), 12.
- [24] C. Jia, Z. Lin, Y. Huang, and X. Duan, Nanowire electronics: From nanoscale to macroscale, *Chem. Rev.*, **119**, (2019), 9074.
- [25] S. Ju, A. Facchetti, Y. Xuan, J. Liu, F. Ishikawa, P. Ye, C. Zhou, T. J. Marks, and D. B. Janes, Fabrication of fully transparent nanowire transistors for transparent and flexible electronics, *Nature Nanotechnol.*, 2, (2007), 378.
- [26] S. Ju, J. Li, J. Liu, P. C. Chen, Y. G. Ha, F. Ishikawa, H. Chang, C. Zhou, A. Facchetti, D. B. Janes, and T. J. Marks, Transparent active matrix organic light-emitting diode displays driven by nanowire transistor circuitry, *Nano Lett.*, 8, (2008), 997.
- [27] F. F. Runge, Uber einige Producte der Steinkohlen destillation, Annu. Rev. Phys. Chem., 31, (1834), 308.
- [28] H. Letheby, On the production of a blue substance by the electrolysis of sulphate of aniline, J. Chem. Soc., 15, (1862), 161.
- [29] S. C. Rasmussen, The early history of polyaniline: discovery and origins, Substantia, 1, (2017), 99.
- [30] R. McNeill, R. Siudak, J. H. Wardlaw, and D. E. Weiss, Electronic conduction in polymers. I. The chemical structure of polypyrrole, *Aust. J. Chem.*, 16, (1963), 1056.
- [31] H. Shirakawa, E. J. Louis, A. G. MacDiarmid, C. K. Chiang, and A. J. Heeger, Synthesis of electrically conducting organic polymers: Halogen derivatives of polyacetylene, (CH)x, J. Chem. Soc. Chem. Commun., 16, (1977), 578.
- [32] M. Noda, N. Kobayashi, M. Katsuhara, A. Yumoto, S. Ushikura, R. Yasuda, N. Hirai, G. Yukawa, I. Yagi, K. Nomoto, and T. Urabe, An OTFT-driven rollable OLED display, J. Soc. Inf. Disp., 19, (2011), 316.
- [33] T. Sekitani, T. Yokota, U. Zschieschang, H. Klauk, S. Bauer, K. Takeuchi, M. Takamiya, T. Sakurai, and T. Someya, Organic nonvolatile memory transistors for flexible sensor arrays, *Science*, **326**, (2009), 1516.

- [34] T. Sekitani, U. Zschieschang, H. Klauk, and T. Someya, Flexible organic transistors and circuits with extreme bending stability, *Nature Materials*, 9, (2010), 1015.
- [35] H. Klauk, U. Zschieschang, J. Pflaum, and M. Halik, Ultralow-power organic complementary circuits, *Nature*, 445, (2007), 745.
- [36] W. Xiong, Y. Guo, U. Zschieschang, H. Klauk, and B. Murmann, A 3-V, 6-bit C-2C digital-to-analog converter using complementary organic thin-film transistors on glass, *IEEE J. Solid-State Circuits*, 45, (2010), 1380.
- [37] U. Zschieschang, T. Yamamoto, K. Takimiya, H. Kuwabara, M. Ikeda, T. Sekitani, T. Someya, and H. Klauk, Organic electronics on banknotes, *Adv. Mater.*, 23, (2011), 654.
- [38] U. Zschieschang and H. Klauk, Low-voltage organic transistors with steep subthreshold slope fabricated on commercially available paper, Org. Electronics, 25, (2015), 340.
- [39] U. Kraft, T. Zaki, F. Letzkus, J. N. Burghartz, E. Weber, B. Murmann, and H. Klauk, Low-voltage, high-frequency organic transistors and unipolar and complementary ring oscillators on paper, Adv. Electron. Mater., 5, (2019), 1800453.
- [40] U. Zschieschang and H. Klauk, Organic transistors on paper: A brief review, J. Mater. Chem. C, 7, (2019), 5522.
- [41] J. B. Lee and V. Subramanian, Weave patterned organic transistors on fiber for e-textiles, *IEEE Trans. Electron Devices*, **52**, (2005), 269.
- [42] A. Bonfiglio, D. De Rossi, T. Kirstein, I. R. Locher, F. Mameli, R. Paradiso, and G. Vozzi, Organic field effect transistors for textile applications, *IEEE Trans. Inf. Technol. Biomed.*, 9, (2005), 319.
- [43] A. N. Sokolov, B. C.-K. Tee, C. J. Bettinger, J. B.-H. Tok, and Z. Bao, Chemical and engineering approaches to enable organic field-effect transistors for electronic skin applications, Acc. Chem. Res., 45, (2012), 361.
- [44] S. Lee, Y. Inoue, D. Kim, A. Reuveny, K. Kuribara, T. Yokota, J. Reeder, M. Sekino, T. Sekitani, Y. Abe, and T. Someya, A strain-absorbing design for
tissue-machine interfaces using a tunable adhesive gel, *Nat. Commun.*, **5**, (2014), 5898.

- [45] N. Thejo Kalyani and S. J. Dhoble, Organic light emitting diodes: Energy saving lighting technology - A review, *Renew. Sustain. Energy Rev.*, 16, (2012), 2696.
- [46] E. G. Jeong, J. H. Kwon, K. S. Kang, S. Y. Jeong, and K. C. Choi, A review of highly reliable flexible encapsulation technologies towards rollable and foldable OLEDs, J. Inf. Disp., 21, (2020), 19.
- [47] S. M. Lee, J. H. Kwon, S. Kwon, and K. C. Choi, A review of flexible OLEDs toward highly durable unusual displays, *IEEE Trans. Electron Devices*, 64, (2017), 1922.
- [48] D. Wöhrle and D. Meissner, Organic solar cells, Adv. Mater., 3, (1991), 129.
- [49] T. Ameri, P. Khoram, J. Min, and C. J. Brabec, Organic ternary solar cells: A review, Adv. Mater., 25, (2013), 4245.
- [50] Q. An, F. Zhang, J. Zhang, W. Tang, Z. Deng, and B. Hu, Versatile ternary organic solar cells: A critical review, *Energy Environ. Sci.*, 9, (2016), 281.
- [51] I. Yagi, N. Hirai, Y. Miyamoto, M. Noda, A. Imaoka, N. Yoneya, K. Nomoto, J. Kasahara, A. Yumoto, and T. Urabe, A flexible full-color AMOLED display driven by OTFTs, J. Soc. Inf. Disp., 16, (2008), 15.
- [52] T. Someya, Y. Kato, T. Sekitani, S. Iba, Y. Noguchi, Y. Murase, H. Kawaguchi, and T. Sakurai, Conformable, flexible, large-area networks of pressure and thermal sensors with organic transistor active matrixes, *Proc. Natl. Acad. Sci.*, **102**, (2005), 12321.
- [53] T. Someya, A. Dodabalapur, J. Huang, K. C. See, and H. E. Katz, Chemical and physical sensing by organic field-effect transistors and related devices, *Adv. Mater.*, 22, (2010), 3799.
- [54] J. A. Rogers, Z. Bao, K. Baldwin, A. Dodabalapur, B. Crone, V. R. Raju, V. Kuck, H. Katz, K. Amundson, J. Ewing, and P. Drzaic, Paper-like electronic displays: Large-area rubber-stamped plastic sheets of electronics and microencapsulated electrophoretic inks, *Proc. Natl. Acad. Sci.*, **98**, (2001), 4835.

- [55] T. Someya, T. Sekitani, S. Iba, Y. Kato, H. Kawaguchi, and T. Sakurai, A largearea, flexible pressure sensor matrix with organic field-effect transistors for artificial skin applications, *Proc. Natl. Acad. Sci.*, **101**, (2004), 9966.
- [56] W. Lee, D. Kim, J. Rivnay, N. Matsuhisa, T. Lonjaret, T. Yokota, H. Yawo, M. Sekino, G. G. Malliaras, and T. Someya, Integration of organic electrochemical and field-effect transistors for ultraflexible, high temporal resolution electrophysiology arrays, Adv. Mater., 28, (2016), 9722.
- [57] H. Fuketa, K. Yoshioka, Y. Shinozuka, K. Ishida, T. Yokota, N. Matsuhisa, Y. Inoue, M. Sekino, T. Sekitani, M. Takamiya, T. Someya, and T. Sakurai, 1 μmthickness ultra-flexible and high electrode-density surface electromyogram measurement sheet with 2 V organic transistors for prosthetic hand control, *IEEE Trans. Biomed. Ckt. Sys.*, 8, (2014), 824.
- [58] T. Someya, Z. Bao, and G. G. Malliaras, The rise of plastic bioelectronics, *Nature*, 540, (2016), 379.
- [59] A. Kekulé, Ueber einige Condensationsproducte des Aldehyds, Justus Liebigs Ann. Chem., 162, (1872), 77.
- [60] C. W. Scherr, Free-electron network model for conjugated systems. II. Numerical Calculations, J. Chem. Phys., 21, (1953), 1582.
- [61] A. Troisi and G. Orlandi, Band structure of the four pentacene polymorphs and effect on the hole mobility at low temperature, J. Phys. Chem. B, 109, (2005), 1849.
- [62] Y. C. Cheng, R. J. Silbey, D. A. da Silva Filho, J. P. Calbert, J. Cornil, and J. L. Brédas, Three-dimensional band structure and bandlike mobility in oligoacene single crystals: A theoretical investigation, J. Chem. Phys., 118, (2003), 3764.
- [63] J. Repp, G. Meyer, S. M. Stojković, A. Gourdon, and C. Joachim, Molecules on insulating films: Scanning-tunneling microscopy imaging of individual molecular orbitals, *Phys. Rev. Lett.*, **94**, (2005), 026803.
- [64] A. Facchetti, Semiconductors for organic transistors, *Mater. Today*, 10, (2007), 28.

- [65] C. Reese and Z. Bao, Organic single-crystal field-effect transistors, *Mater. Today*, 10, (2007), 20.
- [66] V. Podzorov, Organic single crystals: Addressing the fundamentals of organic electronics, MRS Bulletin, 38, (2013), 15.
- [67] L. Jiang, H. Dong, and W. Hu, Organic single crystal field-effect transistors: Advances and perspectives, J. Mater. Chem., 20, (2010), 4994.
- [68] T. Hasegawa and J. Takeya, Organic field-effect transistors using single crystals, Sci. Technol. Adv. Mater., 10, (2009), 024314.
- [69] L. Shi, Y. Guo, W. Hu, and Y. Liu, Design and effective synthesis methods for high-performance polymer semiconductors in organic field-effect transistors, *Mater. Chem. Front.*, 1, (2017), 2423.
- [70] B. S. Ong, Y. Wu, P. Liu, and S. Gardner, High-performance semiconducting polythiophenes for organic thin-film transistors, J. Am. Chem. Soc., 126, (2004), 3378.
- [71] H. Sirringhaus, M. Bird, T. Richards, and N. Zhao, Charge transport physics of conjugated polymer field-effect transistors, Adv. Mater., 22, (2010), 3893.
- [72] T. Yamamoto and K. Takimiya, Facile synthesis of highly π-extended heteroarenes, Dinaphtho[2,3-b:2',3'-f]chalcogenopheno[3,2-b]chalcogenophenes, and their application to field-effect transistors, J. Am. Chem. Soc., **129**, (2007), 2224.
- [73] K. Takimiya, H. Ebata, K. Sakamoto, T. Izawa, T. Otsubo, and Y. Kunugi, 2,7-Diphenyl[1]benzothieno[3,2-b]benzothiophene, a new organic semiconductor for air-stable organic field-effect transistors with mobilities up to 2.0 cm²V⁻¹s⁻¹, J. Am. Chem. Soc., **128**, (2006), 12604.
- [74] F. Garnier, G. Horowitz, D. Fichou, and A. Yassar, Molecular order in organicbased field-effect transistors, *Synth. Metals*, 81, (1996), 163.
- [75] A. Troisi, Charge transport in high mobility molecular semiconductors: Classical models and new theories, *Chem. Soc. Rev.*, 40, (2011), 2347.
- [76] G. Horowitz, R. Hajlaoui, and P. Delannoy, Temperature dependence of the fieldeffect mobility of sexithiophene: determination of the density of traps, J. Phys. III, 54, (1995), 355.

- [77] M. C. J. M. Vissenberg and M. Matters, Theory of the field-effect mobility in amorphous organic transistors, *Phys. Rev. B*, 57, (1998), 12964.
- [78] J. E. Anthony, J. S. Brooks, D. L. Eaton, and S. R. Parkin, Functionalized pentacene: Improved electronic properties from control of solid-state order, J. Am. Chem. Soc., 123, (2001), 9482.
- [79] C. D. Sheraw, T. N. Jackson, D. L. Eaton, and J. E. Anthony, Functionalized pentacene active layer organic thin-film transistors, Adv. Mater., 15, (2003), 2009.
- [80] H. Meng, F. Sun, M. B. Goldfinger, F. Gao, D. J. Londono, W. J. Marshal, G. S. Blackman, K. D. Dobbs, and D. E. Keys, 2,6-Bis[2-(4-pentylphenyl)vinyl]anthracene: A stable and high charge mobility organic semi-conductor with densely packed crystal structure, J. Am. Chem. Soc., 128, (2006), 9304.
- [81] H. Klauk, U. Zschieschang, R. T. Weitz, H. Meng, F. Sun, G. Nunes, D. E. Keys, C. R. Fincher, and Z. Xiang, Organic transistors based on di(phenylvinyl)anthracene: Performance and stability, *Adv. Mater.*, **19**, (2007), 3882.
- [82] S. Braun, W. R. Salaneck, and M. Fahlman, Energy-level alignment at organic/metal and organic/organic interfaces, Adv. Mater., 21, (2009), 1450.
- [83] A. Dodabalapur, L. Torsi, and H. E. Katz, Organic transistors: Two-dimensional transport and improved electrical characteristics, *Science*, 268, (1995), 270.
- [84] F. Dinelli, M. Murgia, P. Levy, M. Cavallini, F. Biscarini, and D. M. de Leeuw, Spatially correlated charge transport in organic thin film transistors, *Phys. Rev. Lett.*, **92**, (2004), 116802.
- [85] J. R. Brews, A charge-sheet model of the MOSFET, Solid State Electron., 21, (1978), 345.
- [86] M. Weis, Gradual channel approximation models for organic field-effect transistors: The space-charge field effect, J. Appl. Phys., 111, (2012), 054506.
- [87] D. James, J. Smith, M. Heeney, T. D. Anthopoulos, A. Salleo, and I. McCulloch, Organic semiconductor materials for transistors, H. Klauk, editor, In Organic Electronics II: More Materials and Applications (Wiley-VCH, 2012).

- [88] S. Brotherton, Introduction to thin film transistors: Physics and technology of TFTs (Springer, 2013).
- [89] S. Sze and K. Ng, Physics of Semiconductor Devices (Wiley, 2007).
- [90] B. M. Dhar, R. Özgün, T. Dawidczyk, A. Andreou, and H. E. Katz, Threshold voltage shifting for memory and tuning in printed transistor circuits, *Mater. Sci. Eng. R*, **72**, (2011), 49.
- [91] H. E. Katz and J. Huang, Thin-film organic electronic devices, Annu. Rev. Mater. Res., 39, (2009), 71.
- [92] W. L. Kalb and B. Batlogg, Calculating the trap density of states in organic fieldeffect transistors from experiment: A comparison of different methods, *Phys. Rev.* B, 81, (2010), 035327.
- [93] H. S. Wong, M. H. White, T. J. Krutsick, and R. V. Booth, Modeling of transconductance degradation and extraction of threshold voltage in thin oxide MOSFET's, *Solid State Electron.*, **30**, (1987), 953.
- [94] D. Boudinet, G. Le Blevennec, C. Serbutoviez, J. M. Verilhac, H. Yan, and G. Horowitz, Contact resistance and threshold voltage extraction in n -channel organic thin film transistors on plastic substrates, J. Appl. Phys., 105, (2009), 084510.
- [95] A. Ortiz-Conde, F. J. García Sánchez, J. J. Liou, A. Cerdeira, M. Estrada, and Y. Yue, A review of recent MOSFET threshold voltage extraction methods, *Microelectron. Reliab.*, 42, (2002), 583.
- [96] V. K. Singh and B. Mazhari, Measurement of threshold voltage in organic thin film transistors, Appl. Phys. Lett., 102, (2013), 253304.
- [97] M. L. Green, E. P. Gusev, R. Degraeve, and E. L. Garfunkel, Ultrathin (<4 nm) SiO₂ and Si-O-N gate dielectric layers for silicon microelectronics: Understanding the processing, structure, and physical and electrical limits, J. Appl. Phys., 90, (2001), 2057.
- [98] B. Wang, W. Huang, L. Chi, M. Al-Hashimi, T. J. Marks, and A. Facchetti, High-k gate dielectrics for emerging flexible and stretchable electronics, *Chem. Rev.*, 118, (2018), 5690.

- [99] D. K. Owens and R. C. Wendt, Estimation of the surface free energy of polymers, J. Appl. Polym. Sci., 13, (1969), 1741.
- [100] K. Takimiya, S. Shinamura, I. Osaka, and E. Miyazaki, Thienoacene-based organic semiconductors, Adv. Mater., 23, (2011), 4347.
- [101] E. D. Głowacki, M. Irimia-Vladu, M. Kaltenbrunner, J. Gsiorowski, M. S. White, U. Monkowius, G. Romanazzi, G. P. Suranna, P. Mastrorilli, T. Sekitani, S. Bauer, T. Someya, L. Torsi, and N. S. Sariciftci, Hydrogen-bonded semiconducting pigments for air-stable field-effect transistors, *Adv. Mater.*, 25, (2013), 1563.
- [102] J. Mei, Y. Diao, A. L. Appleton, L. Fang, and Z. Bao, Integrated materials design of organic semiconductors for field-effect transistors, J. Am. Chem. Soc., 135, (2013), 6724.
- [103] J. T. Quinn, J. Zhu, X. Li, J. Wang, and Y. Li, Recent progress in the development of n-type organic semiconductors for organic field effect transistors, J. Mater. Chem. C, 5, (2017), 8654.
- [104] W. Jiang, Y. Li, and Z. Wang, Heteroacenes as high performance organic semiconductors, *Chem. Soc. Rev.*, 42, (2013), 6113.
- [105] S. Liu, W. M. Wang, A. L. Briseno, S. C. Mannsfeld, and Z. Bao, Controlled deposition of crystalline organic semiconductors for field-effect-transistor applications, *Adv. Mater.*, **21**, (2009), 1217.
- [106] C. W. Sele, B. K. Charlotte Kjellander, B. Niesen, M. J. Thornton, J. B. P. Van Der Putten, K. Myny, H. J. Wondergem, A. Moser, R. Resel, A. J. Van Breemen, N. Van Aerle, P. Heremans, J. E. Anthony, and G. H. Gelinck, Controlled deposition of highly ordered soluble acene thin films: Effect of morphology and crystal orientation on transistor performance, Adv. Mater., 21, (2009), 4926.
- [107] G. Horowitz, M. Mottaghi, P. Lang, F. Rodriguez, A. Yassar, S. Lenfant, and D. Vuillaume, On the crucial role of the insulator-semiconductor interface in organic thin-film transistors, Proc. SPIE Org. Field-Effect Transistors V, 6336, (2006), 63360G.

- [108] S. Steudel, S. De Vusser, S. De Jonge, D. Janssen, S. Verlaak, J. Genoe, and P. Heremans, Influence of the dielectric roughness on the performance of pentacene transistors, *Appl. Phys. Lett.*, 85, (2004), 4400.
- [109] B. Wang, T. Zhu, L. Huang, T. L. D. Tam, Z. Cui, J. Ding, and L. Chi, Addressable growth of oriented organic semiconductor ultra-thin films on hydrophobic surface by direct dip-coating, *Org. Electronics*, 24, (2015), 170.
- [110] F. Gholamrezaie, A. M. Andringa, W. S. Roelofs, A. Neuhold, M. Kemerink, P. W. Blom, and D. M. De Leeuw, Charge trapping by self-assembled monolayers as the origin of the threshold voltage shift in organic field-effect transistors, *Small*, 8, (2012), 241.
- [111] S. Kobayashi, T. Nishikawa, T. Takenobu, S. Mori, T. Shimoda, T. Mitani, H. Shimotani, N. Yoshimoto, S. Ogawa, and Y. Iwasa, Control of carrier density by self-assembled monolayers in organic field-effect transistors, *Nature Materials*, 3, (2004), 317.
- [112] J. B. Koo, S. J. Yun, J. W. Lim, S. H. Kim, C. H. Ku, S. C. Lim, J. H. Lee, and T. Zyung, Low-voltage and high-gain pentacene inverters with plasma-enhanced atomic-layer-deposited gate dielectrics, *Appl. Phys. Lett.*, 89, (2006), 033511.
- [113] J. Lee, J. H. Kim, and S. Im, Pentacene thin-film transistors with Al₂O_{3+x} gate dielectric films deposited on indium-tin-oxide glass, *Appl. Phys. Lett.*, 83, (2003), 2689.
- [114] H. Jinno, T. Yokota, N. Matsuhisa, M. Kaltenbrunner, Y. Tachibana, and T. Someya, Low operating voltage organic transistors and circuits with anodic titanium oxide and phosphonic acid self-assembled monolayer dielectrics, Org. Electronics, 40, (2017), 58.
- [115] L. A. Majewski, R. Schroeder, and M. Grell, Low-voltage, high-performance organic field-effect transistors with an ultra-thin TiO₂ layer as gate insulator, Adv. Funct. Mater., 15, (2005), 1017.
- [116] Y. Gong, K. Zhao, H. He, W. Cai, N. Tang, H. Ning, S. Wu, J. Gao, G. Zhou, X. Lu, and J. M. Liu, Solution processable high quality ZrO₂ dielectric films for low operation voltage and flexible organic thin film transistor applications, *J. Phys. D: Appl. Phys.*, **51**, (2018), 115105.

- [117] W. He, W. Xu, Q. Peng, C. Liu, G. Zhou, S. Wu, M. Zeng, Z. Zhang, J. Gao, X. Gao, X. Lu, and J. M. Liu, Surface modification on solution processable ZrO₂ high-k dielectrics for low voltage operations of organic thin film transistors, J. Phys. Chem. C, **120**, (2016), 9949.
- [118] O. Acton, G. Ting, H. Ma, and A. K. Jen, Low-voltage high-performance C₆₀ thin film transistors via low-surface-energy phosphonic acid monolayer/hafnium oxide hybrid dielectric, *Appl. Phys. Lett.*, **93**, (2008), 083302.
- [119] O. Acton, G. Ting, H. Ma, J. W. Ka, H. L. Yip, N. M. Tucker, and A. K. Jen, Π-Σ-Phosphonic acid organic monolayer/sol-gel hafnium oxide hybrid dielectrics for low-voltage organic transistors, Adv. Mater., 20, (2008), 3697.
- [120] O. Acton, G. G. Ting, P. J. Shamberger, F. S. Ohuchi, H. Ma, and A. K. Jen, Dielectric surface-controlled low-voltage organic transistors via n-alkyl phosphonic acid self-assembled monolayers on high-k metal oxide, ACS Appl. Mater. Interfaces, 2, (2010), 511.
- [121] C. Bartic, H. Jansen, A. Campitelli, and S. Borghs, Ta₂O₅ as gate dielectric material for low-voltage organic thin-film transistors, Org. Electronics, 3, (2002), 65.
- [122] Z. R. Wang, J. Z. Xin, X. C. Ren, X. L. Wang, C. W. Leung, S. Q. Shi, A. Ruotolo, and P. K. L. Chan, Low power flexible organic thin film transistors with amorphous Ba_{0.7}Sr_{0.3}TiO₃ gate dielectric grown by pulsed laser deposition at low temperature, Org. Electronics, **13**, (2012), 1223.
- [123] C. D. Dimitrakopoulos, S. Purushothaman, J. Kymissis, A. Callegari, and J. M. Shaw, Low-voltage organic transistors on plastic comprising high-dielectric constant gate insulators, *Science*, 283, (1999), 822.
- [124] A. R. Benvenho, W. S. Machado, I. Cruz-Cruz, and I. A. Hümmelgen, Study of poly(3-hexylthiophene)/cross-linked poly(vinyl alcohol) as semiconductor/insulator for application in low voltage organic field effect transistors, J. Appl. Phys., 113, (2013), 214509.
- [125] S. Y. Yang, S. H. Kim, K. Shin, H. Jeon, and C. E. Park, Low-voltage pentacene field-effect transistors with ultrathin polymer gate dielectrics, *Appl. Phys. Lett.*, 88, (2006), 173507.

- [126] K. Haase, J. Zessin, K. Zoumboulis, M. Müller, M. Hambsch, and S. C. Mannsfeld, Solution shearing of a high-capacitance polymer dielectric for low-voltage organic transistors, Adv. Electron. Mater., 5, (2019), 1900067.
- [127] J. H. Park, D. K. Hwang, J. Lee, S. Im, and E. Kim, Studies on poly(methyl methacrylate) dielectric layer for field effect transistor: Influence of polymer tacticity, *Thin Solid Films*, **515**, (2007), 4041.
- [128] T. S. Huang, Y. K. Su, and P. C. Wang, Study of organic thin film transistor with polymethylmethacrylate as a dielectric layer, *Appl. Phys. Lett.*, **91**, (2007), 092116.
- [129] S. H. Kim, K. Hong, W. Xie, K. H. Lee, S. Zhang, T. P. Lodge, and C. D. Frisbie, Electrolyte-gated transistors for organic and printed electronics, *Adv. Mater.*, 25, (2013), 1822.
- [130] G. Palazzo, D. De Tullio, M. Magliulo, A. Mallardi, F. Intranuovo, M. Y. Mulla, P. Favia, I. Vikholm-Lundin, and L. Torsi, Detection beyond Debye's length with an electrolyte-gated organic field-effect transistor, *Adv. Mater.*, 27, (2015), 911.
- [131] S. Ono, K. Miwa, S. Seki, and J. Takeya, A comparative study of organic singlecrystal transistors gated with various ionic-liquid electrolytes, *Appl. Phys. Lett.*, 94, (2009), 063301.
- [132] Y. J. Jo, H. Kim, J. Ok, Y. J. Shin, J. H. Shin, T. H. Kim, Y. Jung, and T. il Kim, Biocompatible and biodegradable organic transistors using a solid-state electrolyte incorporated with choline-based ionic liquid and polysaccharide, Adv. Funct. Mater., 30, (2020), 1909707.
- [133] K. H. Lee, M. S. Kang, S. Zhang, Y. Gu, T. P. Lodge, and C. D. Frisbie, "Cut and stick" rubbery ion gels as high capacitance gate dielectrics, *Adv. Mater.*, 24, (2012), 4457.
- [134] S. H. Kim, K. Hong, W. Xie, K. H. Lee, S. Zhang, T. P. Lodge, and C. D. Frisbie, Electrolyte-gated transistors for organic and printed electronics, *Adv. Mater.*, 25, (2013), 1822.

- [135] D. Kong, R. Pfattner, A. Chortos, C. Lu, A. C. Hinckley, C. Wang, W. Y. Lee, J. W. Chung, and Z. Bao, Capacitance characterization of elastomeric dielectrics for applications in intrinsically stretchable thin film transistors, *Adv. Funct. Mater.*, **26**, (2016), 4680.
- [136] Y. Zhou, S. T. Han, Z. X. Xu, X. B. Yang, H. P. Ng, L. B. Huang, and V. A. Roy, Functional high-k nanocomposite dielectrics for flexible transistors and inverters with excellent mechanical properties, J. Mater. Chem., 22, (2012), 14246.
- [137] L. H. Chen, P. Lin, J. C. Ho, C. C. Lee, C. Kim, and M. C. Chen, Polyimide/Ta₂O₅ nanocomposite gate insulators for enhanced organic thin-film transistor performance, *Synth. Metals*, **161**, (2011), 1527.
- [138] G. M. Whitesides and B. Grzybowski, Self-assembly at all scales, Science, 295, (2002), 2418.
- [139] P. Fontaine, D. Goguenheim, D. Deresmes, D. Vuillaume, M. Garet, and F. Rondelez, Octadecyltrichlorosilane monolayers as ultrathin gate insulating films in metal-insulator-semiconductor devices, *Appl. Phys. Lett.*, **62**, (1993), 2256.
- [140] C. Boulas, J. V. Davidovits, F. Rondelez, and D. Vuillaume, Suppression of charge carrier tunneling through organic self-assembled monolayers, *Phys. Rev. Lett.*, 76, (1996), 4797.
- [141] J. Collet and D. Vuillaume, Nano-field effect transistor with an organic selfassembled monolayer as gate insulator, Appl. Phys. Lett., 73, (1998), 2681.
- [142] J. Collet, O. Tharaud, A. Chapoton, and D. Vuillaume, Low-voltage, 30 nm channel length, organic transistors with a self-assembled monolayer as gate insulating films, *Appl. Phys. Lett.*, **76**, (2000), 1941.
- [143] M. Halik, H. Klauk, U. Zschieschang, G. Schmid, S. Ponomarenko, S. Kirchmeyer, and W. Weber, Relationship between molecular structure and electrical performance of oligothiophene organic thin film transistors, Adv. Mater., 15, (2003), 917.
- [144] M. Halik, H. Klauk, U. Zschieschang, G. Schmid, C. Dehm, M. Schütz, S. Maisch,
 F. Effenberger, M. Brunnbauer, and F. Stellacci, Low-voltage organic transistors with an amorphous molecular gate dielectric., *Nature*, 431, (2004), 963.

- [145] Y. D. Park, D. H. Kim, Y. Jang, M. Hwang, J. A. Lim, and K. Cho, Low-voltage polymer thin-film transistors with a self-assembled monolayer as the gate dielectric, *Appl. Phys. Lett.*, 87, (2005), 243509.
- [146] H. Gleskova, S. Gupta, and P. Šutta, Structural changes in vapour-assembled noctylphosphonic acid monolayer with post-deposition annealing: Correlation with bias-induced transistor instability, Org. Electronics, 14, (2013), 3000.
- [147] S. Gupta and H. Gleskova, Dry growth of n-octylphosphonic acid monolayer for low-voltage organic thin-film transistors, Org. Electronics, 14, (2013), 354.
- [148] R. Luschtinetz, A. F. Oliveira, J. Frenzel, J. O. Joswig, G. Seifert, and H. A. Duarte, Adsorption of phosphonic and ethylphosphonic acid on aluminum oxide surfaces, *Surf. Sci.*, **602**, (2008), 1347.
- [149] T. Bauer, T. Schmaltz, T. Lenz, M. Halik, B. Meyer, and T. Clark, Phosphonateand carboxylate-based self-assembled monolayers for organic devices: A theoretical study of surface binding on aluminum oxide with experimental support, ACS Appl. Mater. Interfaces, 5, (2013), 6073.
- [150] R. Luschtinetz, A. F. Oliveira, H. A. Duarte, and G. Seifert, Self-assembled monolayers of alkylphosphonic acids on aluminum oxide surfaces - A theoretical study, *Zeitschrift fur Anorg. und Allg. Chemie*, 636, (2010), 1506.
- [151] I. N. Hulea, S. Fratini, H. Xie, C. L. Mulder, N. N. Iossad, G. Rastelli, S. Ciuchi, and A. F. Morpurgo, Tunable Fröhlich polarons in organic single-crystal transistors, *Nature Materials*, 5, (2006), 982.
- [152] C. Goldmann, D. J. Gundlach, and B. Batlogg, Evidence of water-related discrete trap state formation in pentacene single-crystal field-effect transistors, *Appl. Phys. Lett.*, 88, (2006), 063501.
- [153] T. N. Ng, J. A. Marohn, and M. L. Chabinyc, Comparing the kinetics of bias stress in organic field-effect transistors with different dielectric interfaces, J. Appl. Phys., 100, (2006), 084505.
- [154] K. Fukuda, T. Hamamoto, T. Yokota, T. Sekitani, U. Zschieschang, H. Klauk, and T. Someya, Effects of the alkyl chain length in phosphonic acid self-assembled

monolayer gate dielectrics on the performance and stability of low-voltage organic thin-film transistors, *Appl. Phys. Lett.*, **95**, (2009), 203301.

- [155] U. Kraft, U. Zschieschang, F. Ante, D. Kälblein, C. Kamella, K. Amsharov, M. Jansen, K. Kern, E. Weber, and H. Klauk, Fluoroalkylphosphonic acid selfassembled monolayer gate dielectrics for threshold-voltage control in low-voltage organic thin-film transistors, J. Mater. Chem., 20, (2010), 6416.
- [156] R. Acharya, B. Peng, P. K. L. Chan, G. Schmitz, and H. Klauk, Achieving ultralow turn-on voltages in organic thin-film transistors: Investigating fluoroalkylphosphonic acid self-assembled monolayer hybrid dielectrics, ACS Appl. Mater. Interfaces, 11, (2019), 27104.
- [157] U. Zschieschang, F. Ante, M. Schlörholz, M. Schmidt, K. Kern, and H. Klauk, Mixed self-assembled monolayer gate dielectrics for continuous threshold voltage control in organic transistors and circuits, *Adv. Mater.*, **22**, (2010), 4489.
- [158] I. Hirata, U. Zschieschang, T. Yokota, K. Kuribara, M. Kaltenbrunner, H. Klauk, T. Sekitani, and T. Someya, High-resolution spatial control of the threshold voltage of organic transistors by microcontact printing of alkyl and fluoroalkylphosphonic acid self-assembled monolayers, Org. Electronics, 26, (2015), 239.
- [159] A. L. Deman and J. Tardy, PMMA-Ta₂O₅ bilayer gate dielectric for low operating voltage organic FETs, Org. Electronics, 6, (2005), 78.
- [160] K. Shin, C. Yang, S. Y. Yang, H. Jeon, and C. E. Park, Effects of polymer gate dielectrics roughness on pentacene field-effect transistors, *Appl. Phys. Lett.*, 88, (2006), 072109.
- [161] D. W. Park, C. A. Lee, K. D. Jung, B. G. Park, H. Shin, and J. D. Lee, Low hysteresis pentacene thin-film transistors using SiO₂/cross-linked poly(vinyl alcohol) gate dielectric, *Appl. Phys. Lett.*, 89, (2006), 263507.
- [162] J. Zhou, K. Yang, J. Zhou, Y. Liu, J. Peng, and Y. Cao, Poly(3-hexylthiophene) thin-film transistors with dual insulator layers, Jpn. J. Appl. Phys., 46, (2007), 913.

- [163] F. Y. Yang, K. J. Chang, M. Y. Hsu, and C. C. Liu, Low-operating-voltage polymeric transistor with solution-processed low-k polymer/high-k metal-oxide bilayer insulators, Org. Electronics, 9, (2008), 925.
- [164] R. P. Ortiz, A. Facchetti, and T. J. Marks, High-k organic, inorganic, and hybrid dielectrics for low-voltage organic field-effect transistors, *Chem. Rev.*, **110**, (2010), 205.
- [165] H. S. Lee, D. H. Kim, J. H. Cho, Y. D. Park, J. S. Kim, and K. Cho, Enhancement of interconnectivity in the channels of pentacene thin-film transistors and its effect on field-effect mobility, *Adv. Funct. Mater.*, 16, (2006), 1859.
- [166] H. S. Lee, D. H. Kim, J. H. Cho, M. Hwang, Y. Jang, and K. Cho, Effect of the phase states of self-assembled monolayers on pentacene growth and thin-film transistor characteristics, J. Am. Chem. Soc., 130, (2008), 10556.
- [167] T. Miyadera, S. D. Wang, T. Minari, K. Tsukagoshi, and Y. Aoyagi, Charge trapping induced current instability in pentacene thin film transistors: Trapping barrier and effect of surface treatment, *Appl. Phys. Lett.*, **93**, (2008), 033304.
- [168] A. L. Briseno, S. C. Mannsfeld, M. M. Ling, S. Liu, R. J. Tseng, C. Reese, M. E. Roberts, Y. Yang, F. Wudl, and Z. Bao, Patterning organic single-crystal transistor arrays, *Nature*, 444, (2006), 913.
- [169] A. Virkar, S. Mannsfeld, J. H. Oh, M. F. Toney, Y. H. Tan, G. Y. Liu, J. C. Scott, R. Miller, and Z. Bao, The role of OTS density on pentacene and C₆₀ nucleation, thin film growth, and transistor performance, *Adv. Funct. Mater.*, **19**, (2009), 1962.
- [170] Y. Ito, A. A. Virkar, S. Mannsfeld, J. H. Oh, M. Toney, J. Locklin, and Z. Bao, Crystalline ultrasmooth self-assembled monolayers of alkylsilanes for organic fieldeffect transistors, J. Am. Chem. Soc., 131, (2009), 9396.
- [171] M. Tello, M. Chiesa, C. M. Duffy, and H. Sirringhaus, Charge trapping in intergrain regions of pentacene thin film transistors, Adv. Funct. Mater., 18, (2008), 3907.
- [172] A. Mityashin, O. M. Roscioni, L. Muccioli, C. Zannoni, V. Geskin, J. Cornil, D. Janssen, S. Steudel, J. Genoe, and P. Heremans, Multiscale modeling of the

electrostatic impact of self-assembled monolayers used as gate dielectric treatment in organic thin-film transistors, ACS Appl. Mater. Interfaces, 6, (2014), 15372.

- [173] O. Acton, I. Osaka, G. Ting, D. Hutchins, H. Ma, R. D. McCullough, and A. K. Jen, Phosphonic acid self-assembled monolayer and amorphous hafnium oxide hybrid dielectric for high performance polymer thin film transistors on plastic substrates, *Appl. Phys. Lett.*, **95**, (2009), 113305.
- [174] O. Acton, G. G. Ting, H. Ma, D. Hutchins, Y. Wang, B. Purushothaman, J. E. Anthony, and A. K. Jen, π-σ-Phosphonic acid organic monolayer-amorphous sol-gel hafnium oxide hybrid dielectric for low-voltage organic transistors on plastic, J. Mater. Chem., 19, (2009), 7929.
- [175] M. Aghamohammadi, R. Rödel, U. Zschieschang, C. Ocal, H. Boschker, R. T. Weitz, E. Barrena, and H. Klauk, Threshold-voltage shifts in organic transistors due to self-assembled monolayers at the dielectric: evidence for electronic coupling and dipolar effects, ACS Appl. Mater. Interfaces, 7, (2015), 22775.
- [176] M. Kaltenbrunner, T. Sekitani, J. Reeder, T. Yokota, K. Kuribara, T. Tokuhara, M. Drack, R. Schwödiauer, I. Graz, S. Bauer-Gogonea, S. Bauer, and T. Someya, An ultra-lightweight design for imperceptible plastic electronics., *Nature*, 499, (2013), 458.
- [177] D. Liu, Z. He, Y. Su, Y. Diao, S. C. Mannsfeld, Z. Bao, J. Xu, and Q. Miao, Self-assembled monolayers of cyclohexyl-terminated phosphonic acids as a general dielectric surface for high-performance organic thin-film transistors, Adv. Mater., 26, (2014), 7190.
- [178] X. Xu, Y. Yao, B. Shan, X. Gu, D. Liu, J. Liu, J. Xu, N. Zhao, W. Hu, and Q. Miao, Electron mobility exceeding 10 cm²V⁻¹s⁻¹ and band-like charge transport in solution-processed n-channel organic thin-film transistors, Adv. Mater., 28, (2016), 5276.
- [179] S. Gupta, S. Hannah, C. P. Watson, P. Šutta, R. H. Pedersen, N. Gadegaard, and H. Gleskova, Ozone oxidation methods for aluminum oxide formation: Application to low-voltage organic transistors, *Org. Electronics*, **21**, (2015), 132.

- [180] K. C. Chinnam, S. Gupta, and H. Gleskova, Aluminium oxide prepared by UV/ozone exposure for low-voltage organic thin-film transistors, J. Non. Cryst. Solids, 358, (2012), 2512.
- [181] U. Kraft, M. Sejfić, M. J. Kang, K. Takimiya, T. Zaki, F. Letzkus, J. N. Burghartz, E. Weber, and H. Klauk, Flexible low-voltage organic complementary circuits: Finding the optimum combination of semiconductors and monolayer gate dielectrics, Adv. Mater., 27, (2015), 207.
- [182] U. Zschieschang, M. Halik, and H. Klauk, Microcontact-printed self-assembled monolayers as ultrathin gate dielectrics in organic thin-film transistors and complementary circuits, *Langmuir*, 24, (2008), 1665.
- [183] M. Novak, T. Schmaltz, H. Faber, and M. Halik, Influence of selfassembled monolayer dielectrics on the morphology and performance of α, ω dihexylquaterthiophene in thin film transistors, *Appl. Phys. Lett.*, **98**, (2011), 093302.
- [184] M. Salinas, C. M. Jäger, A. Y. Amin, P. O. Dral, T. Meyer-Friedrichsen, A. Hirsch, T. Clark, and M. Halik, The relationship between threshold voltage and dipolar character of self-assembled monolayers in organic thin-film transistors, J. Am. Chem. Soc., 134, (2012), 12648.
- [185] H. Chen, W. Zhang, M. Li, G. He, and X. Guo, Interface engineering in organic field-effect transistors: Principles, applications, and perspectives, *Chem. Rev.*, **120**, (2020), 2879.
- [186] L. A. Majewski, R. Schroeder, M. Voigt, and M. Grell, High performance organic transistors on cheap, commercial substrates, J. Phys. D: Appl. Phys., 37, (2004), 3367.
- [187] T. Lei, M. Guan, J. Liu, H. C. Lin, R. Pfattner, L. Shaw, A. F. McGuire, T. C. Huang, L. Shao, K. T. Cheng, J. B. Tok, and Z. Bao, Biocompatible and totally disintegrable semiconducting polymer for ultrathin and ultralightweight transient electronics, *Proc. Natl. Acad. Sci.*, **114**, (2017), 5107.
- [188] J. Kwon, Y. Lee, Y. Jo, and S. Jung, Fabrication of ultrathin low-voltage-driven printed organic circuits with anodized gate islands, *Org. Electronics*, 62, (2018), 77.

- [189] S. Yogev, R. Matsubara, M. Nakamura, U. Zschieschang, H. Klauk, and Y. Rosenwaks, Fermi level pinning by gap states in organic semiconductors, *Phys. Rev. Lett.*, **110**, (2013), 036803.
- [190] J. W. Borchert, U. Zschieschang, F. Letzkus, M. Giorgio, R. T. Weitz, M. Caironi, J. N. Burghartz, S. Ludwigs, and H. Klauk, Flexible low-voltage high-frequency organic thin-film transistors, *Sci. Adv.*, 6, (2020), eaaz5156.
- [191] A. Jedaa, M. Burkhardt, U. Zschieschang, H. Klauk, D. Habich, G. Schmid, and M. Halik, The impact of self-assembled monolayer thickness in hybrid gate dielectrics for organic thin-film transistors, *Org. Electronics*, **10**, (2009), 1442.
- [192] Y. Jung, R. J. Kline, D. A. Fischer, E. K. Lin, M. Heeney, I. McCulloch, and D. M. DeLongchamp, The effect of interfacial roughness on the thin film morphology and charge transport of high-performance polythiophenes, *Adv. Funct. Mater.*, 18, (2008), 742.
- [193] G. Lin, Q. Wang, L. Peng, M. Wang, H. Lu, G. Zhang, G. Lv, and L. Qiu, Impact of the lateral length scales of dielectric roughness on pentacene organic field-effect transistors, J. Phys. D: Appl. Phys., 48, (2015), 105103.
- [194] K. Shin, S. Y. Yang, C. Yang, H. Jeon, and C. E. Park, Effects of polar functional groups and roughness topography of polymer gate dielectric layers on pentacene field-effect transistors, *Org. Electronics*, 8, (2007), 336.
- [195] D. Knipp, R. A. Street, and A. R. Völkel, Morphology and electronic transport of polycrystalline pentacene thin-film transistors, *Appl. Phys. Lett.*, 82, (2003), 3907.
- [196] K. Suemori, S. Uemura, M. Yoshida, S. Hoshino, N. Takada, T. Kodzasa, and T. Kamata, Influence of fine roughness of insulator surface on threshold voltage stability of organic field-effect transistors, *Appl. Phys. Lett.*, **93**, (2008), 033308.
- [197] M. Geiger, R. Acharya, E. Reutter, T. Ferschke, U. Zschieschang, J. Weis, J. Pflaum, H. Klauk, and R. T. Weitz, Effect of the degree of the gate-dielectric surface roughness on the performance of bottom-gate organic thin-film transistors, *Adv. Mater. Interfaces*, 7, (2020), 1902145.

- [198] S. E. Fritz, T. W. Kelley, and C. D. Frisbie, Effect of dielectric roughness on performance of pentacene TFTs and restoration of performance with a polymeric smoothing layer, J. Phys. Chem. B, 109, (2005), 10574.
- [199] D. Choi, T. K. An, Y. J. Kim, D. S. Chung, S. H. Kim, and C. E. Park, Effects of semiconductor/dielectric interfacial properties on the electrical performance of top-gate organic transistors, *Org. Electronics*, **15**, (2014), 1299.
- [200] M. Li, C. An, T. Marszalek, M. Baumgarten, K. Müllen, and W. Pisula, Impact of interfacial microstructure on charge carrier transport in solution-processed conjugated polymer field-effect transistors, Adv. Mater., 28, (2016), 2245.
- [201] S. S. Chang, A. B. Rodríguez, A. M. Higgins, C. Liu, M. Geoghegan, H. Sirringhaus, F. Cousin, R. M. Dalgleish, and Y. Deng, Control of roughness at interfaces and the impact on charge mobility in all-polymer field-effect transistors, *Soft Matter*, 4, (2008), 2220.
- [202] S. Lee, F. M. Li, and A. Nathan, Influence of surface energy and roughness on hole mobility in solution-processed hybrid organic thin film transistors, *IEEE J. Electron Devices Soc.*, 6, (2018), 653.
- [203] J. Evertsson, F. Bertram, F. Zhang, L. Rullik, L. R. Merte, M. Shipilin, M. Soldemo, S. Ahmadi, N. Vinogradov, F. Carlà, J. Weissenrieder, M. Göthelid, J. Pan, A. Mikkelsen, J. O. Nilsson, and E. Lundgren, The thickness of native oxides on aluminum alloys and single crystals, *Appl. Surf. Sci.*, **349**, (2015), 826.
- [204] K. D. Kim and C. K. Song, Low voltage pentacene thin film transistors employing a self-grown metal-oxide as a gate dielectric, *Appl. Phys. Lett.*, 88, (2006), 233508.
- [205] H. Kang, K.-k. Han, J. E. Park, and H. H. Lee, High mobility, low voltage polymer transistor, Org. Electronics, 8, (2007), 460.
- [206] N. Cabrera and N. F. Mott, Theory of the oxidation of metals, *Reports Prog. Phys.*, **12**, (1949), 163.
- [207] N. Cai, G. Zhou, K. Müller, and D. E. Starr, Tuning the limiting thickness of a thin oxide layer on Al(111) with oxygen gas pressure, *Phys. Rev. Lett.*, 107, (2011), 035502.

- [208] J. D. Baran, H. Grönbeck, and A. Hellman, Mechanism for limiting thickness of thin oxide films on aluminum, *Phys. Rev. Lett.*, **112**, (2014), 146103.
- [209] A. K. Harman, S. Ninomiya, and S. Adachi, Optical constants of sapphire (α-Al₂O₃) single crystals, J. Appl. Phys., 76, (1994), 8032.
- [210] M. D. Groner, F. H. Fabreguette, J. W. Elam, and S. M. George, Low-temperature Al₂O₃ atomic layer deposition, *Chem. Mater.*, **16**, (2004), 639.
- [211] J. Acharya, J. Wilt, B. Liu, and J. Wu, Probing the dielectric properties of ultrathin Al/Al₂O₃/Al trilayers fabricated using in situ sputtering and atomic layer deposition, ACS Appl. Mater. Interfaces, 10, (2018), 3112.
- [212] J. Li, J. Du, J. Xu, H. L. Chan, and F. Yan, The influence of gate dielectrics on a high-mobility n-type conjugated polymer in organic thin-film transistors, *Appl. Phys. Lett.*, **100**, (2012), 033301.
- [213] P. Klapetek, D. Nečas, and C. Anderson, Gwyddion user guide (2013).
- [214] T. Gredig, E. A. Silverstein, and M. P. Byrne, Height-height correlation function to determine grain size in iron phthalocyanine thin films, J. Phys. Conf. Ser., 417, (2013), 012069.
- [215] J. W. Borchert, B. Peng, F. Letzkus, J. N. Burghartz, P. K. Chan, K. Zojer, S. Ludwigs, and H. Klauk, Small contact resistance and high-frequency operation of flexible low-voltage inverted coplanar organic transistors, *Nat. Commun.*, 10, (2019), 1119.
- [216] C. W. Liang, T. C. Luo, M. S. Feng, H. C. Cheng, and D. Su, Characterization of anodic aluminum oxide film and its application to amorphous silicon thin film transistors, *Mater. Chem. Phys.*, 43, (1996), 166.
- [217] G. Z. Geng, G. X. Liu, F. K. Shan, A. Liu, Q. Zhang, W. J. Lee, B. C. Shin, and H. Z. Wu, Improved performance of InGaZnO thin-film transistors with Ta₂O₅/Al₂O₃ stack deposited using pulsed laser deposition, *Curr. Appl. Phys.*, 14, (2014), S2.
- [218] L. Lan and J. Peng, High-performance indium-gallium-zinc oxide thin-film transistors based on anodic aluminum oxide, *IEEE Trans. Electron Devices*, 58, (2011), 1452.

- [219] I. Kang, C. Avis, D. H. Kang, and J. Jang, Low-voltage poly-si TFTs with solution-processed aluminum oxide gate dielectric, *Electrochem. Solid-State Lett.*, 14, (2011), J51.
- [220] P. K. Nayak, M. N. Hedhili, D. Cha, and H. N. Alshareef, High performance In₂O₃ thin film transistors using chemically derived aluminum oxide dielectric, *Appl. Phys. Lett.*, **103**, (2013), 033518.
- [221] R. Branquinho, D. Salgueiro, L. Santos, P. Barquinha, L. Pereira, R. Martins, and E. Fortunato, Aqueous combustion synthesis of aluminum oxide thin films and application as gate dielectric in GZTO solution-based TFTs, ACS Appl. Mater. Interfaces, 6, (2014), 19592.
- [222] W. J. Lee, W. T. Park, S. Park, S. Sung, Y. Y. Noh, and M. H. Yoon, Largescale precise printing of ultrathin sol-gel oxide dielectrics for directly patterned solution-processed metal oxide transistor arrays, *Adv. Mater.*, 27, (2015), 5043.
- [223] V. Verlaan, L. R. van den Elzen, G. Dingemans, M. C. van de Sanden, and W. M. Kessels, Composition and bonding structure of plasma-assisted ALD Al₂O₃ films, *Phys. Status Solidi C*, 7, (2010), 976.
- [224] D. N. Seidman, Three-dimensional atom-probe tomography: Advances and applications, Annu. Rev. Mater. Res., 37, (2007), 127.
- [225] B. Gault, A. Chiaramonti, O. Cojocaru-Mirédin, P. Stender, R. Dubosq,
 C. Freysoldt, S. K. Makineni, T. Li, M. Moody, and J. M. Cairney, Atom probe tomography, *Nature Reviews Methods Primers*, 1, (2021), 1.
- [226] I. E. McCarroll, P. Bagot, A. Devaraj, D. E. Perea, and J. Cairney, New frontiers in atom probe tomography: a review of research enabled by cryo and/or vacuum transfer systems, *Materials Today Advances*, 7, (2020), 100090.
- [227] D. J. Larson, A. Cerezo, J. Juraszek, K. Hono, and G. Schmitz, Atom-probe tomographic studies of thin films and multilayers, *MRS Bulletin*, 34, (2009), 732.
- [228] V. Vovk, G. Schmitz, and R. Kirchheim, Three-dimensional atom probe investigation of co/al thin film reaction, *Microelectronic engineering*, **70**, (2003), 533.

- [229] D. Mangelinck, K. Hoummada, O. Cojocaru-Mirédin, E. Cadel, C. Perrin-Pellegrino, and D. Blavette, Atom probe tomography of ni silicides: First stages of reaction and redistribution of pt, *Microelectronic engineering*, 85, (2008), 1995.
- [230] P. Pareige, E. Cadel, X. Sauvage, B. Deconihout, D. Blavette, and D. Mangelinck, Atomic resolution analyses of nano-structured materials by atom probe tomography, *International Journal of Nanotechnology*, 5, (2008), 592.
- [231] D. Isheim, R. P. Kolli, M. E. Fine, and D. N. Seidman, An atom-probe tomographic study of the temporal evolution of the nanostructure of fe-cu based high-strength low-carbon steels, *Scripta materialia*, 55, (2006), 35.
- [232] L. F. White, J. R. Darling, D. E. Moser, D. A. Reinhard, J. Dunlop, D. J. Larson, D. Lawrence, and I. Martin, editors, Complex nanostructures in shocked, annealed, and metamorphosed baddeleyite defined by atom probe tomography (2018).
- [233] D. Joester, A. Hillier, Y. Zhang, and T. J. Prosa, Organic materials and organic/inorganic heterostructures in atom probe tomography, *Microscopy Today*, 20, (2012), 26.
- [234] A. P. Proudian, M. B. Jaskot, D. R. Diercks, B. P. Gorman, and J. D. Zimmerman, Atom probe tomography of molecular organic materials: sub-dalton nanometerscale quantification, *Chemistry of Materials*, **31**, (2019), 2241.
- [235] K. P. Pernstich, S. Haas, D. Oberhoff, C. Goldmann, D. J. Gundlach, B. Batlogg, A. N. Rashid, and G. Schitter, Threshold voltage shift in organic field effect transistors by dipole monolayers on the gate insulator, J. Appl. Phys., 96, (2004), 6431.
- [236] S. K. Possanner, K. Zojer, P. Packer, E. Zojer, and F. Schürrer, Threshold voltage shifts in organic thin-film transistors due to self-assembled monolayers at the dielectric surface, Adv. Funct. Mater., 19, (2009), 958.
- [237] S. Y. Yang, K. Shin, and C. E. Park, The effect of gate-dielectric surface energy on pentacene morphology and organic field-effect transistor characteristics, Adv. Funct. Mater., 15, (2005), 1806.
- [238] D. Janssen, R. De Palma, S. Verlaak, P. Heremans, and W. Dehaen, Static solvent contact angle measurements, surface free energy and wettability determination

of various self-assembled monolayers on silicon dioxide, *Thin Solid Films*, **515**, (2006), 1433.

- [239] D. Liu, X. Xu, Y. Su, Z. He, J. Xu, and Q. Miao, Self-assembled monolayers of phosphonic acids with enhanced surface energy for high-performance solutionprocessed n-channel organic thin-film transistors, *Angew. Chemie - Int. Ed.*, **52**, (2013), 6222.
- [240] H. R. Byun, E. A. You, and Y. G. Ha, Multifunctional hybrid multilayer gate dielectrics with tunable surface energy for ultralow-power organic and amorphous oxide thin-film transistors, ACS Appl. Mater. Interfaces, 9, (2017), 7347.
- [241] M. de Pauli, U. Zschieschang, I. D. Barcelos, H. Klauk, and A. Malachias, Tailoring the dielectric layer structure for enhanced carrier mobility in organic transistors: The use of hybrid inorganic/organic multilayer dielectrics, Adv. Electron. Mater., 2, (2016), 1500402.
- [242] T. W. Kelley, L. D. Boardman, T. D. Dunbar, D. V. Muyres, M. J. Pellerite, and T. P. Smith, High-performance OTFTs using surface-modified alumina dielectrics, *J. Phys. Chem. B*, **107**, (2003), 5877.
- [243] I. G. Hill, C. M. Weinert, L. Kreplak, and B. P. Van Zyl, Influence of self-assembled monolayer chain length on modified gate dielectric pentacene thin-film transistors, *Appl. Phys. A*, **95**, (2009), 81.
- [244] S. Hannah, J. Cardona, D. A. Lamprou, P. Šutta, P. Baran, A. Al Ruzaiqi, K. Johnston, and H. Gleskova, Interplay between vacuum-grown monolayers of alkylphosphonic acids and the performance of organic transistors based on Dinaphtho[2,3-b:2',3'-f]thieno[3,2-b]thiophene, ACS Appl. Mater. Interfaces, 8, (2016), 25405.
- [245] D. Gupta and Y. Hong, Understanding the effect of semiconductor thickness on device characteristics in organic thin film transistors by way of two-dimensional simulations, Org. Electronics, 11, (2010), 127.
- [246] R. Ruiz, A. Papadimitratos, A. C. Mayer, and G. G. Malliaras, Thickness dependence of mobility in pentacene thin-film transistors, *Adv. Mater.*, 17, (2005), 1795.

- [247] Y. Ran, L. ZhiDong, H. YuFeng, C. ShaoBo, T. Feng, H. YanBing, and L. XiaoJun, Effects of gate dielectric thickness and semiconductor thickness on device performance of organic field-effect transistors based on pentacene, *Sci. China Technol. Sci.*, 57, (2014), 1142.
- [248] Y. Xu, C. Liu, W. Scheideler, S. Li, W. Li, Y. F. Lin, F. Balestra, G. Ghibaudo, and K. Tsukagoshi, Understanding thickness-dependent charge transport in pentacene transistors by low-frequency noise, *IEEE Electron Device Lett.*, 34, (2013), 1298.
- [249] N. Matsuo and A. Heya, Dependence of electrical properties of pentacene thin-film transistor on active layer thickness, *IEICE Electron. Express*, 8, (2011), 360.
- [250] E. L. Granstrom and C. D. Frisbie, Field effect conductance measurements on thin crystals of sexithiophene, J. Phys. Chem. B, 103, (1999), 8842.
- [251] M. Kiguchi, M. Nakayama, K. Fujiwara, K. Ueno, T. Shimada, and K. Saiki, Accumulation and depletion layer thicknesses in organic field effect transistors, *Jpn. J. Appl. Phys.*, 42, (2003), L1408.
- [252] F. J. Zhang, C. A. Di, N. Berdunov, Y. Hu, Y. Hu, X. Gao, Q. Meng, H. Sirringhaus, and D. Zhu, Ultrathin film organic transistors: Precise control of semiconductor thickness via spin-coating, *Adv. Mater.*, 25, (2013), 1401.
- [253] F. De Angelis, S. Cipolloni, L. Mariucci, and G. Fortunato, Aging effects in pentacene thin-film transistors: Analysis of the density of states modification, Appl. Phys. Lett., 88, (2006), 193508.
- [254] F. De Angelis, M. Gaspari, A. Procopio, G. Cuda, and E. Di Fabrizio, Direct mass spectrometry investigation on Pentacene thin film oxidation upon exposure to air, *Chem. Phys. Lett.*, 468, (2009), 193.
- [255] O. D. Jurchescu, J. Baas, and T. T. Palstra, Effect of impurities on the mobility of single crystal pentacene, *Appl. Phys. Lett.*, 84, (2004), 3061.
- [256] U. Zschieschang, F. Ante, T. Yamamoto, K. Takimiya, H. Kuwabara, M. Ikeda, T. Sekitani, T. Someya, K. Kern, and H. Klauk, Flexible low-voltage organic transistors and circuits based on a high-mobility organic semiconductor with good air stability, *Adv. Mater.*, 22, (2010), 982.

- [257] Y. Fujisaki, Y. Nakajima, T. Takei, H. Fukagawa, T. Yamamoto, and H. Fujikake, Flexible active-matrix organic light-emitting diode display using air-stable organic semiconductor of Dinaphtho[2,3-b:2',3'-f]thieno[3,2-b]-thiophene, *IEEE Trans. Electron Devices*, **59**, (2012), 3442.
- [258] K. Fukuda, T. Sekitani, U. Zschieschang, H. Klauk, K. Kuribara, T. Yokota, T. Sugino, K. Asaka, M. Ikeda, H. Kuwabara, T. Yamamoto, K. Takimiya, T. Fukushima, T. Aida, M. Takamiya, T. Sakurai, and T. Someya, A 4 V operation, flexible braille display using organic transistors, carbon nanotube actuators, and organic static random-access memory, Adv. Funct. Mater., 21, (2011), 4019.
- [259] J. Milvich, T. Zaki, M. Aghamohammadi, R. Rödel, U. Kraft, H. Klauk, and J. N. Burghartz, Flexible low-voltage organic phototransistors based on airstable dinaphtho[2,3-b:2',3'-f]thieno[3,2-b]thiophene (DNTT), Org. Electronics, 20, (2015), 63.
- [260] S. Kim, H. Moon, H. Kwon, G. Lee, D. Yu, J. Choi, J. Park, S. J. Kim, and S. Yoo, Organic vapor-jet printing with reduced heat transfer for fabrication of flexible organic devices, *Adv. Mater. Technol.*, 4, (2019), 1800332.
- [261] K. Ishida, T. C. Huang, K. Honda, Y. Shinozuka, H. Fuketa, T. Yokota, U. Zschieschang, H. Klauk, G. Tortissier, T. Sekitani, H. Toshiyoshi, M. Takamiya, T. Someya, and T. Sakurai, Insole pedometer with piezoelectric energy harvester and 2 V organic circuits, *IEEE J. Solid-State Circuits*, 48, (2013), 255.
- [262] T. Sekitani, T. Yokota, K. Kuribara, M. Kaltenbrunner, T. Fukushima, Y. Inoue, M. Sekino, T. Isoyama, Y. Abe, H. Onodera, and T. Someya, Ultraflexible organic amplifier with biocompatible gel electrodes, *Nat. Commun.*, 7, (2016), 11425.
- [263] A. K. Mahfuzul Islam, M. Hamamatsu, T. Yokota, S. Lee, W. Yukita, M. Takamiya, T. Someya, and T. Sakurai, Programmable neuron array based on a 2-transistor multiplier using organic floating-gate for intelligent sensors, *IEEE J. Emerg. Sel. Top. Circuits Syst.*, 7, (2017), 81.
- [264] M. Elsobky, Y. Mahsereci, Z. Yu, H. Richter, J. N. Burghartz, J. Keck, H. Klauk, and U. Zschieschang, Ultra-thin smart electronic skin based on hybrid system-infoil concept combining three flexible electronics technologies, *Electron. Lett.*, 54, (2018), 338.

- [265] U. Zschieschang, F. Ante, D. Kälblein, T. Yamamoto, K. Takimiya, H. Kuwabara, M. Ikeda, T. Sekitani, T. Someya, J. B. Nimoth, and H. Klauk, Dinaphtho[2,3b:2',3'-f]thieno[3,2-b]thiophene (DNTT) thin-film transistors with improved performance and stability, Org. Electronics, 12, (2011), 1370.
- [266] Z. Ding, G. Abbas, H. E. Assender, J. J. Morrison, S. G. Yeates, E. R. Patchett, and D. M. Taylor, Effect of oxygen, moisture and illumination on the stability and reliability of Dinaphtho[2,3-b:2',3'-f]thieno[3,2-b]thiophene (DNTT) OTFTs during operation and storage, ACS Appl. Mater. Interfaces, 6, (2014), 15224.
- [267] T. Breuer, A. Karthäuser, H. Klemm, F. Genuzio, G. Peschel, A. Fuhrich, T. Schmidt, and G. Witte, Exceptional dewetting of organic semiconductor films: The case of Dinaphthothienothiophene (DNTT) at dielectric interfaces, ACS Appl. Mater. Interfaces, 9, (2017), 8384.
- [268] S. Herminghaus, K. Jacobs, K. Mecke, J. Bischof, A. Fery, M. Ibn-Elhaj, and S. Schlagowski, Spinodal dewetting in liquid crystal and liquid metal films, *Science*, 282, (1998), 916.
- [269] M. C. Jung, M. R. Leyden, G. O. Nikiforov, M. V. Lee, H. K. Lee, T. J. Shin, K. Takimiya, and Y. Qi, Flat-lying semiconductor-insulator interfacial layer in DNTT thin films, ACS Appl. Mater. Interfaces, 7, (2015), 1833.
- [270] R. Ruiz, D. Choudhary, B. Nickel, T. Toccoli, K. C. Chang, A. C. Mayer, P. Clancy, J. M. Blakely, R. L. Headrick, S. Iannotta, and G. G. Malliaras, Pentacene thin film growth, *Chem. Mater.*, 16, (2004), 4497.
- [271] A. C. Dürr, F. Schreiber, K. A. Ritley, V. Kruppa, J. Krug, H. Dosch, and B. Struth, Rapid roughening in thin film growth of an organic semiconductor (Diindenoperylene), *Phys. Rev. Lett.*, **90**, (2003), 016104.
- [272] D. Käfer, C. Wöll, and G. Witte, Thermally activated dewetting of organic thin films: The case of pentacene on SiO₂ and gold, *Appl. Phys. A*, **95**, (2009), 273.
- [273] A. Amassian, V. A. Pozdin, T. V. Desai, S. Hong, A. R. Woll, J. D. Ferguson, J. D. Brock, G. G. Malliaras, and J. R. Engstrom, Post-deposition reorganization of pentacene films deposited on low-energy surfaces, J. Mater. Chem., 19, (2009), 5580.

- [274] A. A. Virkar, S. Mannsfeld, Z. Bao, and N. Stingelin, Organic semiconductor growth and morphology considerations for organic thin-film transistors, Adv. Mater., 22, (2010), 3857.
- [275] H. Chen, M. Li, Z. Lu, X. Wang, J. Yang, Z. Wang, F. Zhang, C. Gu, W. Zhang, Y. Sun, J. Sun, W. Zhu, and X. Guo, Multistep nucleation and growth mechanisms of organic crystals from amorphous solid states, *Nat. Commun.*, **10**, (2019), 3872.
- [276] D. Yu, Y. Q. Yang, Z. Chen, Y. Tao, and Y. F. Liu, Recent progress on thin-film encapsulation technologies for organic electronic devices, *Opt. Commun.*, 362, (2016), 43.
- [277] J. M. Kim, J. Oh, K. M. Jung, K. C. Park, J. H. Jeon, and Y. S. Kim, Ultrathin flexible thin film transistors with CYTOP encapsulation by debonding process, *Semicond. Sci. Technol.*, 34, (2019), 075015.
- [278] D. Feili, M. Schuettler, T. Doerge, S. Kammer, and T. Stieglitz, Encapsulation of organic field effect transistors for flexible biomedical microimplants, *Sensors Actuators, A Phys.*, **120**, (2005), 101.
- [279] C. Pannemann, T. Diekmann, U. Hilleringmann, U. Schürmann, M. Scharnberg, V. Zaporojtchenko, R. Adelung, and F. Faupel, PTFE encapsulation for pentacene based organic thin film transistors, *Mater. Sci. Pol.*, 25, (2007), 95.
- [280] R. C. Haddon, A. S. Perel, R. C. Morris, T. T. Palstra, A. F. Hebard, and R. M. Fleming, C₆₀ thin film transistors, *Appl. Phys. Lett.*, 67, (1995), 121.
- [281] T. Breuer, T. Geiger, H. F. Bettinger, and G. Witte, Diels-Alder adduct formation at solid interfaces between fullerenes and acenes, J. Phys. Condens. Matter, 31, (2019), 034003.
- [282] T. Breuer and G. Witte, Diffusion-controlled growth of molecular heterostructures: Fabrication of two-, one-, and zero-dimensional C₆₀ nanostructures on pentacene substrates, ACS Appl. Mater. Interfaces, 5, (2013), 9740.
- [283] O. A. Melville, B. H. Lessard, and T. P. Bender, Phthalocyanine-based organic thin-film transistors: A review of recent advances, ACS Appl. Mater. Interfaces, 7, (2015), 13105.

- [284] M. Kothe, F. Widdascheck, and G. Witte, Titanylphthalocyanine films on Ag(111): An epitaxial metal/organic heterosystem with an exceptional smooth surface, J. Phys. Chem. C, 123, (2019), 6097.
- [285] Y. Fujinaka, R. Ye, K. Ohta, K. Nishidate, and M. Baba, Improved organic thin film transistor performance utilizing a DH-α6T submonolayer, *Mol. Cryst. Liq. Cryst.*, **580**, (2013), 110.

Acknowledgements

It is said that it takes a village to raise a child, and this research work took a a world of people without whom, it would have been impossible to complete.

I would like to express my gratitude to Prof. Guido Schmitz for accepting to be my thesis supervisor, his continuous support and imparting his vast knowledge and advice to my PhD projects. I am extremely grateful to Prof. Sabine Ludwigs and Prof. Peer Fischer for agreeing to be a part of my thesis committee, taking the time to read my thesis and providing their valuable and informative comments along with their support. I am deeply indebted to Dr. Hagen Klauk for giving me the opportunity to work in his group at the Max Plank Institute for Solid State Research, and providing constant support, encouragement and a nurturing atmosphere for professional and personal growth. Thank you for teaching me so many things, about transistors and organic electronics, but also how about scientific writing, the importance of being particular and precise, while always being patient and kind with me. I thank the entire group of Organic Electronics, with all past and present members for creating an intellectually stimulating and educative environment. Dr. Ute Zschieschang, for always lending a helping hand around the lab as well as keeping aside German short stories for me to read. Dr. James Borchert, Dr. Mélanie Brouillard, Vera Bader, Tobias Wollandt and Sabrina Steffens for their valuable inputs on presentations and during projects, as well as the wonderful company and interesting conversations during lunchtime and during travel to conferences. I would particularly like to express my deepest gratitude to Michael Geiger, for being the best project partner to collaborate with, lending an educative and enriching experience working closely together, and for being a friend with a helping hand in times of need.

My immense gratitude to Prof. Klaus von Klitzing for his endorsement and giving me the opportunity to attend the Lindau Nobel Laureate Meeting in 2016. I am grateful to Prof. von Klitzing and Prof. Mannhart for the annual invitation and a great experience at the Ringberg castle each year. I would also like to thank several people at the Max Planck Institute, Jürgen Weis for giving access to the Nanostructuring Lab and Thomas Reindl, Marion Hagel, Ulrike Waizmann and Bernhard Fenk for their expert technical assistance. Katrin Küster for giving access to the AFM and Thomas Whittles for giving valuable inputs on manuscripts. I would like to express special thanks to Birgit King, Frank Gottschalk, and Michael Eppard for being on top of all the administrative procedures and providing excellent assistance.

Throughout all the projects I had the wonderful opportunity to work with excellent scientists and collaborate on different projects with their expertise. Helena Solodenko and Rüya Duran from the University of Stuttgart, Dr. Boyu Peng and Prof. Paddy Chan from the University of Hong Kong, Prof. Gregor Witte, Dr. Tobias Breuer and Darius Günder from the University of Marburg, and Robin Lingstädt and Julia Deuschle from the STEM Group at the Max Planck Institute.

I consider myself incredibly lucky to have had the opportunity to live in a new country with the lovely people who made the whole experience as comfortable and pleasant for me as possible. To Ulrike and Günter, thank you for welcoming me to Stuttgart and into your family and being my constant support structure all these years. Swarali Paranjape, for being there for me whenever I have needed someone, and for bridging the gap between India and Germany for me. To the GBC Stuttgart Book Club, for the monthly book club meetings, wonderful conversations and enjoyable evenings. In particular, Sarah, Patricia, Nicole and Steph for being such wonderful role models and women I look up to every day. To Evelyn Regenfus, Inge Stephan, Jessica Moews and Sadia Shakil, for always checking up on me and sending me all their love and best wishes, especially during difficult times. Katrin Teutsch, for not only providing me with a lovely home but for taking care of everything I needed at all times. To Daniel Braun and Petr Hansa, thank you for not only teaching me the German language, but also for instilling the confidence in me to navigate a foreign land. To all my friends from back home and across the world, Omkar Sandaw, Tushar Bhitre, Rajashree Padmanabhi, Onkar Deshpande, Jasmine Latkar and Mary Lin, for your unwavering support and warm friendship all these years.

When the world went into lockdown and isolation, there was a network of PhD warriors who stuck together and supported each other through it all. I am immensely grateful to Dr. Donna Peach for setting up the PhDForum, a community of the most wonderful people who became family. Thank you to each and every member of the Forum, and special gratitude to Erin, Joanne, Susan, Jo (and Helen), Rachel, Bernadette, Katy, Dawn, Sónia, Julia, Teresa, Lyazat, Sandra, Colin, Riley, for always being there for me. Thank you for restoring my faith in humanity and setting a great example for what academia can be.

Throughout my entire PhD journey and from before it even started, I am blessed to have the support and encouragement of my entire extended family in India. I am incredibly grateful for the numerous care-packages, phone calls (and Zoom calls), and words and gestures of encouragement throughout these years. To Aai, Baba, Rasika, Mia and Bella: thank you for being by my side, catching me when I fall and pushing me when I need to fly. This was possible only because of your faith in me.